

University of Huddersfield Repository

Alyassin, Mohammad

ARABINOXYLAN PREBIOTICS CO-PRODUCTION WITHIN INTEGRATED BIOREFINERIES

Original Citation

Alyassin, Mohammad (2019) ARABINOXYLAN PREBIOTICS CO-PRODUCTION WITHIN INTEGRATED BIOREFINERIES. Doctoral thesis, University of Huddersfield.

This version is available at http://eprints.hud.ac.uk/id/eprint/34941/

The University Repository is a digital collection of the research output of the University, available on Open Access. Copyright and Moral Rights for the items on this site are retained by the individual author and/or other copyright owners. Users may access full items free of charge; copies of full text items generally can be reproduced, displayed or performed and given to third parties in any format or medium for personal research or study, educational or not-for-profit purposes without prior permission or charge, provided:

- The authors, title and full bibliographic details is credited in any copy;
- A hyperlink and/or URL is included for the original metadata page; and
- The content is not changed in any way.

For more information, including our policy and submission procedure, please contact the Repository Team at: E.mailbox@hud.ac.uk.

http://eprints.hud.ac.uk/

ARABINOXYLAN PREBIOTICS CO-PRODUCTION WITHIN INTEGRATED BIOREFINERIES

MOHAMMAD ALYASSIN

A thesis submitted to the University of Huddersfield in partial fulfilment of the requirements for the degree of Doctor of Philosophy

The University of Huddersfield

January 2019

Copyright statement

- I. The author of this thesis (including any appendices and/or schedules to this thesis) owns any copyright in it (the "Copyright") and he has given The University of Huddersfield the right to use such copyright for any administrative, promotional, educational and/or teaching purposes.
- II. Copies of this thesis, either in full or in extracts, may be made only in accordance with the regulations of the University Library. Details of these regulations may be obtained from the Librarian. This page must form part of any such copies made.
- III. The ownership of any patents, designs, trademarks and any and all other intellectual property rights except for the Copyright (the "Intellectual Property Rights") and any reproductions of copyright works, for example graphs and tables ("Reproductions"), which may be described in this thesis, may not be owned by the author and may be owned by third parties. Such Intellectual Property Rights and Reproductions cannot and must not be made available for use without the prior written permission of the owner(s) of the relevant Intellectual Property Rights and/or Reproductions.

Arabinoxylan prebiotics co-production within integrated biorefineries

Abstract

In cereal-based biorefineries, Distillers Dried Grains with Solubles (DDGS) is the main by-product, which is used in relatively low value animal feed formulations. A major component of DDGS is arabinoxylan (AX), which is a potential food ingredient and source of novel prebiotics. The production of AX uses large amounts of ethanol, which gives scope for feasible integration of AX extraction within a biorefinery. Meanwhile, enzymatic production of prebiotic arabinoxylan oligosaccharides (AXOS) and xylooligosaccharides (XOS) is of interest to animal feed formulators, as these prebiotics enhance feed conversion significantly.

The integration scenario would be implemented on the in-process streams, the Distillers Wet Grain (DWG) and the Solubles before they are combined. However, these are not readily available to study, as biorefineries are closed processes that do not allow in-process sampling of these streams. Therefore, this project used the GUNT CE-640 bioethanol unit to produce wet by-products representative of commercial DWG. Fermentation of ten batches of 6 kg of wheat yielded on average 1275 g (db) of DWG with 16% AX content and 800 g of dry Solubles with 11.4% AX content.

Enzymatic treatment of the DWG with commercial endoxylanase yielded less than 3%w/w of the prebiotic oligosaccharides. AX polysaccharides were extracted by alkaline oxidation, with and without further enzymatic purification, to give AX contents of 44% and 19%, respectively. Enzymatic treatment converted only 6% of the AX into AXOS/XOS, with significant production of undesired monosaccharides.

Limitations in the analytical methods used prompted the development of a new HPAEC-PAD method for the simultaneous measurement of mono- and oligosaccharides and uronic acids. The new method was used to quantify the profiles of XOS in commercial materials and following enzyme treatment of biomass materials and oligosaccharide standards up to DP6, the latter to reveal the xylanase mode of action. The commercial xylanase investigated showed a preference for larger XOS molecules and was incapable of acting on branched (AXOS) molecules.

Abstract

Two types of AX with different branching patterns were produced; highly branched AX was extracted chemically from DWG with further purification to give 51% w/w purity, and less branched AX was produced from the Solubles fraction by ultrafiltration to give 74% purity. Enzyme treatment converted 46% of the Solubles AX into oligosaccharides, showing this to be a promising feedstock for XOS production.

Mohammad Alyassin

PhD thesis, January 2019

Table of Contents

1	Arak	inoxylan, a potential co-product of integrated biorefineries	. 15
	1.1	Biorefineries in the recent UK and global context	. 15
	1.2	Arabinoxylan as a co-product of bioethanol production	. 21
	1.3	Scope of the thesis	. 24
2	6		20
2	Arac	nnoxylan structure and physicochemical properties	. 26
	2.1	Introduction	. 26
	2.2	Brief background of arabinoxylan	. 26
	2.3	Structure of Arabinoxylans	. 28
	2.4	Classification of arabinoxylans	. 31
	2.5	Occurrence of arabinoxylans	. 31
	2.6	Physicochemical properties of Arabinoxylans	. 31
	2.6.1	Water solubility	32
	2.6.2	Viscosity	32
	2.0.3	Ovidative Colation and Col Forming Capacity	33 22
	2.0.4	Form stabilising	55 2/
	2.0.5	Research objectives	34
	2.7	Summary	25
3	Pilot	-scale production of biorefinery by-products	. 36
	3.1	Introduction	. 36
	3.2	Bioethanol biorefineries – past, present and future	. 36
	3.2.1	Wet milling	37
	3.2.2	Dry milling	38
	3.3	Distillers dried grains with solubles	. 39
	3.3.1	DDGS and DWG	42
	3.4	Materials and equipment used for pilot-scale bioethanol and DWG production	. 43
	3.4.1	Chemicals and enzymes	43
	3.4.2 Э Е	Methods used for bioethanol and DWG production	44 ЛС
	3.3	Formentation method	. 40
	26	Characterisation of the by-products	40 10
	3.61	Ethanol nurity	رب . 49
	3.6.2	Constituent sugars analysis	49
	3.6.3	Protein content analysis	50
	3.6.4	Ash content determination	51
	3.7	Results	. 52
	3.7.1	Ethanol purity	52
	3.7.2	Constituent sugar analysis results	53
	3.7.3	Protein and ash content results	53
	3.8	Discussion	. 54
	3.9	Summary	. 56
Л	Arak	inovulan Oliaosaccharides Production	59
-	Δ 1	Introduction	50 . 52
	· T • I		

Introduction	. 58
The concept of functional food	. 58
AXOS and XOS prebiotic properties	. 60
Xylanases and AXOS in animal feed	. 63
	Introduction The concept of functional food AXOS and XOS prebiotic properties Xylanases and AXOS in animal feed

		Materials and weath a da fan DMC yn lawraas tweatwraut	~ -
	4.5	Materials and methods for DWG-xylanase treatment	65
	4.5.1	Enzymes and chemicals	65
	4.5.2	Enzyme treatment	66
	4.5.3	Protease pre-treatment	66
	4.5.4	Analytical Methods	67
	4.6	Enzyme treatment results	68
	4.6.1	Monosaccharide Analysis Results	69
	4.6.2	Oligosaccharide profile analysis results	72
	4.6.3	Protease treatment results	75
	4.7	Summary	76
5	Cher	nical extraction of AX from DWG	77
	5.1	Introduction	77
	5.2	Arabinoxylans extraction methods	77
	521	Hydrolysing and solubilising of AX	78
	522	Mechanically assisted extraction	90 81
	5 2 3	Purification of AX	83
	5.2.5	Dilat scale extraction of AV	00
	5.5		00
	5.4	Materials and methods	88
	5.4.1	Chemicals and enzymes	88
	5.4.2	Equipment	88
	5.4.3	Arabinoxylan extraction method	91
	5.4.4	Characterisation of the extracts	94
	5.4.5	Enzyme treatment	95
	5.5	Results and discussion	96
	5.5.1	Extracts characterisation	96
	5.5.2	Enzymatic treatment results 1	100
	5.6	Summary1	.03
6	Olia	osaccharides analysis method development	04
-	61	Introduction 1	01
	6.2	Alizacashavidas analysis mathada	04
	0.2	Oligosaccharides analysis methods	.04
	6.3	Materials and methods1	.14
	6.3.1	Materials 1	114
	6.4	Chromatographic conditions1	.14
	6.4.1	Preparation of standard solutions 1	115
	6.4.2	Sample preparation	116
	6.5	Results and discussion1	.17
	6.5.1	Chromatography and specificity 1	117
	6.5.2	Linearity1	119
	6.5.3	Limits of detection (LOD) and quantitation (LOQ)	120
	6.5.4	Accuracy and precision1	120
	6.5.5	System suitability tests	122
	6.5.6	Sample analysis results	122
	6.6	Summary1	.28
	••••		
7	Endo	xylanases degradation of XOS standards1	30
	7.1	Introduction 1	30
	7.2	Yulanasas — a raview of their features and applications	21
	7.2	Aylanases – a review of their reduces and applications	.51
	1.3	Classification of xylanases1	.32
	7.4	Industrial Applications of Xylanases1	.34
	7.5	Materials and methods1	.38
	7.6	Results and discussion1	.39
	7.7	Summary1	.48
		•	

iorefin	eries	150
8.1	Introduction	
8.2	Integrated biorefineries status	
8.3	Materials and methods	
8.3	.1 Methods	156
8.3	.2 Characterisation of the extracts	158
8.3	.3 Enzymatic treatment of WEAX and WUAX	158
8.4	Results and discussion	
8.5	Summary	165
6	nclusions and Recommendations	168
CO		
9.1	Progress made in the current work	
9.1 9.2	Progress made in the current work Recommendations for future work and industrial uptake	
9.1 9.2 <i>Rej</i>	Progress made in the current work Recommendations for future work and industrial uptake	
9.1 9.2 <i>Rej</i>	Progress made in the current work Recommendations for future work and industrial uptake ferences	
9.1 9.2 <i>Rej</i>	Progress made in the current work Recommendations for future work and industrial uptake ferences pendix	
9.1 9.2 <i>Rej</i>	Progress made in the current work Recommendations for future work and industrial uptake ferences pendix DWG production batches: Arabinoxylan prodauction batches:	
9.1 9.2 Rej Ap	Progress made in the current work Recommendations for future work and industrial uptake ferences pendix DWG production batches: Arabinoxylan prodauction batches:	
9.1 9.2 Rej Ap	Progress made in the current work Recommendations for future work and industrial uptake ferences pendix DWG production batches: Arabinoxylan prodauction batches: Oral presentations	

List of Figures

Figure 1-1. Average annual growth rates of renewable energy capacity and biofue	ls
production, End-2010 to End-2015 (Anon, 2016 a)	17
Figure 1-2. Actual bioethanol production in UK.	18
Figure 1-3. Global new investment in clean energy by sector (Louw, 2018)	19
Figure 1-4. New investment in clean energy in United Kingdom, by sector	19
Figure 2-1. Number of publications about arabinoxylan	27
Figure 2-2. Structure of AX, showing mono- and di-substituted xylose units and a	
ferulic acid link (Mendis <i>et al.,</i> 2016)	29
Figure 2-3. Dehydrodiferulate esters (Ferulic acid dimer)	30
Figure 2-4. Oxidative coupling of ferulic acid (Döring <i>et al.</i> , 2015)	33
Figure 3-1. Corn grain wet milling process.	38
Figure 3-2. Dry milling process for bioethanol production	39
Figure 3-3. Dry-grind ethanol production processes and by-products	41
Figure 3-4. GUNT CE-640 Bioethanol plant, general view	45
Figure 3-5. Process diagram of GUNT CE-640 Bioethanol plant	46
Figure 3-6. Process of DWG production using the GUNT CE-640 bioethanol plant.	48
Figure 3-7. CO_2 release from the water seal during fermentation	49
Figure 3-8. Purity of the ethanol produced in the GUNT CE-640 bioethanol plant	52
Figure 3-9. Mass balance of protein, AX and ash content before and after	
fermentation process	55
Figure 4-1. Arabinose, xylose and glucose standards separation chromatogram	69
Figure 4-2. Total released monosaccharides from DWG after xylanase treatment	
(measured by HPAEC-PAD after hydrolysis).	70
Figure 4-3. Amount of AXOS released from 2 g of DWG by enzymes at different	
dosages and pH	71
Figure 4-4. XOS standards separation by IPOS.	72
Figure 4-5. Oligosaccharides in the samples (%w/w of DWG) by enzymes at differe	ent
dosages and pH	73
Figure 4-6. Protease treatment effect on the released AXOS	75
Figure 5-1. Beckman centrifuge GS-6S	88

Figure 5-2.	Ultrafiltration systems: A Vivaflow, B LabStak
Figure 5-3.	Design of the hybrid ultrafiltration system
Figure 5-4.	Assembled ultrafiltration system91
Figure 5-5.	AXa extraction process
Figure 5-6.	AXb extract purification method94
Figure 5-7.	AXa extract after milling96
Figure 5-8.	Cumulative molar mass of AXb before and after Econase treatment 102
Figure 6-1.	Separation of Xylooligosaccharides by HPLC-RI (Voragen et al., 1986) 107
Figure 6-2.	Chromatogram of XOS standards DP 2-4 (1000 ppm) analysed by RP-HPLC
(Morgan <i>et</i>	al., 2017)
Figure 6-3.	Analysis of XOS with HPAEC-PAD (Falck et al., 2014)
Figure 6-4.	Effect of mobile phase strength, column temperature and flow rate on the
separation	of XOS (Cürten <i>et al.,</i> 2018)111
Figure 6-5.	XOS analysis in animal feed sample (Unpublished work, AB Enzymes labs).
Figure 6-6.	Separation of 1. arabinose; 2. galactose; 3. glucose; 4. xylose; 5. xylonic
acid; 6. glu	conic acid; 7. galacturonic acid; 8. glucuronic acid with step gradient
method (W	'ang <i>et al.,</i> 2012)
Figure 6-7.	Developed HPAEC-PAD method condition for the new method115
Figure 6-8.	Chromatogram of mono and oligosaccharides mixture using CarboPac
PA200	
Figure 6-9.	Chromatogram of mono and oligosaccharides mixture using CarboPac
PA20	
Figure 6-10	0. Chromatographs of Aberystwyth University spent grain samples 123
Figure 6-11	. AB Vista (108-656 XOS syrup) sample chromatogram
Figure 6-12	. XOS syrup enzyme treatment samples125
Figure 6-13	. Mono- and oligosaccharides released from wheat bran by xylanase
treatment.	
Figure 6-14	. Evolution of XOS from wheat bran over time following xylanase
treatment.	
Figure 7-1.	AX hydrolysing enzymes site of attack (Dornez et al., 2009)134
Figure 7-2.	Breakdown of xylohexaose upon Econase treatment

Figure 7-3. Breakdown of xylopentaose upon Econase treatment
Figure 7-4 Econase degradation of a mixture of xylobexaose and xylopentaose 142
Figure 7 4. Econose degradation of a mixture of Aylonexaose and Aylopentaose 142
Figure 7-5. Breakdown of xylotetraose upon Econase treatment
Figure 7-6. Breakdown of xylotriose upon Econase treatment144
Figure 7-7. Breakdown of xylobiose upon Econase treatment144
Figure 7-8. Breakdown of xylobiose upon severe Econase treatment145
Figure 7-9. Degradation of XOS (DP6-DP2) upon Econase treatment
Figure 7-10. Structure of 3^3 - α -L-arabinofuranosyl-xylotetraose and 2^3 - α -L-
arabinofuranosyl-xylotetraose147
Figure 7-11. Structure of 2^3 - α -L-Arabinofuranosyl-xylotriose
Figure 7-12. Treatment of 3^3 - α -L-arabinofuranosyl-xylotetraose and 2^3 - α -L-
arabinofuranosyl-xylotetraose with Econase147
Figure 8-1. Overview of biomass conversion processes and potential products152
Figure 8-2. Carlson filter press154
Figure 8-3. AMI [®] spiral ultrafiltration membrane and housing
Figure 8-4. UF system diagram155
Figure 8-5. Assembled large-scale ultrafiltration kit155
Figure 8-6. The process of WEAX and WUAX production156
Figure 8-7. XOS production from WEAX and WUAX via Econase treatment for 1 hour
and 2 hours162
Figure 8-8. WEAX treatment with Econase at 500×, 50×, 5× and 0.5× the commercial
dose163
Figure 8-9. Econase effect before and after thermal treatment at 70°C for 30 minutes.
164
104

List of Tables

Table 2-1 Potential uses of AX
Table 3-1. Composition of DDGS from different plants and sources (expressed on a %,
dry matter basis) (Jarret et al., 2011; Pedersen et al., 2014; Spiehs et al., 2002)42
Table 3-2. Constituent sugars analysis of the original wheat and the resulting DWG and
Solubles following ethanol production (analyses performed by Englyst Carbohydrate
Ltd.)
Table 3-3. Protein and ash analysis of the original wheat and the resulting DWG and
Solubles following ethanol production (db)54
Table 4-1. Dosage and pH of the enzyme incubation test. 66
Table 4-2. Protease treatment prior to Econase. 67
Table 4-3. Total released monosaccharides from DWG after xylanase treatment
(Measured by HPAEC-PAD after hydrolysis)70
Table 4-4. Amount of AXOS released from 2 g of DWG by enzymes at different dosages
and pH
Table 4-5. Oligosaccharides in the samples(%w/w of DWG) released by enzymes at
different dosages and pH73
Table 4-6. Protease Treatment results %w/w75
Table 5-1. AX recovery following hydrolysis (IPOS). 99
Table 5-2. XOS released from AXb after Econase treatment (IPOS)
Table 5-3. XOS from AXb after arabinofuranosidase and Econase treatment. 101
Table 5-3. XOS from AXb after arabinofuranosidase and Econase treatment
Table 5-3. XOS from AXb after arabinofuranosidase and Econase treatment
Table 5-3. XOS from AXb after arabinofuranosidase and Econase treatment
Table 5-3. XOS from AXb after arabinofuranosidase and Econase treatment.101Table 5-4. Molecular weight and polydispersity index for AXb before and after Econase101treatment.101Table 6-1. Analytical methods for the separation of oligosaccharides.105Table 6-2. Method linearity, LOD and LOQ for each analyte.119
Table 5-3. XOS from AXb after arabinofuranosidase and Econase treatment.101Table 5-4. Molecular weight and polydispersity index for AXb before and after Econase101treatment.101Table 6-1. Analytical methods for the separation of oligosaccharides.105Table 6-2. Method linearity, LOD and LOQ for each analyte.119Table 6-3. Method accuracy (reported as percentage recovery) and precision
Table 5-3. XOS from AXb after arabinofuranosidase and Econase treatment.101Table 5-4. Molecular weight and polydispersity index for AXb before and after Econase101treatment.101Table 6-1. Analytical methods for the separation of oligosaccharides.105Table 6-2. Method linearity, LOD and LOQ for each analyte.119Table 6-3. Method accuracy (reported as percentage recovery) and precision
Table 5-3. XOS from AXb after arabinofuranosidase and Econase treatment.101Table 5-4. Molecular weight and polydispersity index for AXb before and after Econase101treatment.101Table 6-1. Analytical methods for the separation of oligosaccharides.105Table 6-2. Method linearity, LOD and LOQ for each analyte.119Table 6-3. Method accuracy (reported as percentage recovery) and precision
Table 5-3. XOS from AXb after arabinofuranosidase and Econase treatment.101Table 5-4. Molecular weight and polydispersity index for AXb before and after Econase101Table 6-1. Analytical methods for the separation of oligosaccharides.105Table 6-2. Method linearity, LOD and LOQ for each analyte.119Table 6-3. Method accuracy (reported as percentage recovery) and precision .121Table 6-4. Method resolution, asymmetry factor, capacity factor and number of122Table 6-5. Concentration of XOS in Aberystwyth University spent grain samples.124
Table 5-3. XOS from AXb after arabinofuranosidase and Econase treatment.101Table 5-4. Molecular weight and polydispersity index for AXb before and after Econase101Table 6-1. Analytical methods for the separation of oligosaccharides.105Table 6-2. Method linearity, LOD and LOQ for each analyte.119Table 6-3. Method accuracy (reported as percentage recovery) and precision

Acknowledgements

I would like to express my profound gratitude and appreciation to my supervisor Professor Grant Campbell for his valuable guidance, persistent presence, continuous support, his patience, time and effort. It was a great pleasure to work under his direct supervision, and his motivation and support have massively improved my knowledge. I could not have imagined having a better advisor and mentor for my PhD study.

I also would like to thank my co-supervisor Dr. Vasileios Kontogiorgos for his constant support and constructive suggestions.

This work would not have been possible without the contribution of AB Agri and AB Vista; their materials, enzymes and financial support were the main drivers of this research, and I wish to extend my warmest thanks to Dr Mike Bedford and Dr Helen (Nell) Masey O'Neill for their guidance and support throughout this project.

I am grateful to all of those with whom I have had the pleasure to work with during this project, particularly Dr Nick Powles of IPOS who enriched this project with his experience and knowledge and whose analyses kept the project moving forward at a critical time, along with those of Englyst Carbohydrates. I also would like to thank the academic collaborators in Nottingham Trent University, especially Dr Emily Burton and Dr Dawn Scholey, and Judit Zdiarstek from AB Enzymes for help and advice.

A warm thanks goes to all the technicians at the University of Huddersfield for their assistance. I especially appreciate the support given by Dr Richard Hughes and Hayley Markham, including training in the use of specialist equipment.

And I am grateful to my fellow PhD students and researchers in the School of Applied Sciences for shared struggles and triumphs. We have learned a lot together, and I wish you all well in your own studies and careers.

List of abbreviations

AAFCO Association of American Feed Control Officials			
A/X	Arabinose to xylose ratio		
A2A3XXX	A3XXX 23,33-di-α-L-arabinofuranosyl-xylotriose		
A2XXX	23-α-L-arabinofuranosyl-xylotriose		
AACC	American Association for Clinical Chemistry		
ABF	Associated British Foods		
anova	analysis of variance		
AOAC	Association of Official Analytical Chemists		
Ara	Arabinose		
As	Asymmetry		
AX	Arabinoxylan		
АХа	First batch of AX extraction		
Axb	Second batch of AX extraction		
AXOS	Arabinoxylan Oligosaccharides		
CAZy	Carbohydrate-Active enZYmes		
CV	Coefficient of variation		
Da	Dalton		
db	Dry base		
DDGS	Distillers Dried Grains with Solubles		
DP	Degree of polymerisation		
DS	Distillers Solubles		
DWG	Distillers Wet Grain		
EU	Europian union		
FAO	Food and Agriculture Organization		
FFC	Functional Food Centre		
fuc	Fucose		
FuFoSE	The European Commission's Concerted Action on Functional Food Science in Europe		
Gal	Galactose		
GalA	Galactorunic Acid		
GC	Gas chromatography		
Glu	Glucose		
GluA	Glucorunic Acid		
HPAEC	High Performance Anion Exchange Chromatography		
HPLC	High performance liquid chromatography		
HPTLC	High performance thin-layer chromatography		
IPOS	Innovative Physical Organic Solutions		
IR	Infrared		
IUBMB	International Union of Biochemistry and Molecular Biology		
k'	Capacity factor		
LOD	Limits of Detection		

LOQ	Limits of Quantitation		
Mann	Mannose		
MS	Mass spectroscopy		
Mw	Molecular weight		
Mw	Molecular weight		
NSP	Non starch polysaccharides		
NSPase	Non starch polysaccharides hydrolysing enzymes		
PAD	Pulsed amperometric detector		
PV	Solar photovoltaics		
REN21	Renewable Energy Policy Network for the 21st Century		
RFA	Renewable Fuel Association		
Rha	Rhamnose		
RI	Refractive index		
Rs	Resolution factor		
RTFO	Renewable Transport Fuel Obligation		
RTFO	Renewable Transport Fuel Obligation		
S/N	Signal-to-noise		
SCF	Short chain fatty acids		
SCFA	Short chain fatty acids		
SEC-	SEC- Size exclusion chromatography with multi-angle laser light scatteri		
MALLS			
TLC	Thin-layer chromatography		
Uvs	Ultraviolet-Visible spectroscopy		
WEAX	Water-extractable arabinoxylan		
WUAX	Water-unextractable arabinoxylan		
X1	Xylose		
X2	Xylobiose		
ХЗ	Xylotriose		
X4	Xylotetraose		
X5	Xylopentaose		
X6	Xylohexaose		
XA2XXX	23-α-L-arabinofuranosyl-xylotetraose		
XA3XXX	33-α-L-arabinofuranosyl-xylotetraose		
XOS	Xylooligosaccharides		
Xyl	Xylose		

1 Arabinoxylan, a potential co-product of integrated biorefineries

1.1 Biorefineries in the recent UK and global context

The world is facing unprecedented challenges of providing sufficient food and fuel to cover the continuously expanding needs of the increasing population, expected to rise to 9.8 billion by 2050 (UN DESA, 2017). Today's energy production and consumption patterns indicate that the finite coal, oil and gas resources might serve humankind needs for only a few more decades (Agarwal *et al.*, 2017). Around 80% of the energy and 90% of the world's chemicals are produced from fossil resource (Maity, 2015). Such dependence on finite resources is an alarming indicator of catastrophic consequences. Escalating energy consumption is also accompanied by greater emissions of greenhouse gases, mainly CO₂, being emitted into the lower atmosphere and changing the climate at a rapid pace (Von Blottnitz and Curran, 2007; Hamit-Haggar, 2012; Saini *et al.*, 2018). The twin problems of fossil fuel depletion and climate change, in the face of an increasing, and increasingly resource demanding, global population, are the defining features of the early 21st century and exercise the resources and resolve of scientists, engineers, businesses, politicians and society globally.

In the context of increased urgency for a transition to sustainable energy resources with lower carbon footprints, and new paths that reduce the enslavement to the limited fossil fuels, renewable energy sources are continually being considered, including wind, tidal waves, solar energy and biofuels (IRENA, 2017). Among these, biofuels are the only supplier of energy in a liquid transport fuel form, in contrast to the other renewable energy sources (Bom *et al.*, 2007; Huber *et al.*, 2006; Nigam and Singh, 2011). Furthermore, unlike wind, wave and solar energy, biofuels are not intermittent and do not require energy storage systems or the employment of other fuel technology as a backup (Grahn *et al.*, 2009).

Driven by this awareness, the European Commission issued the first EU Biofuels directive in 2003 which targeted increasing the biofuel mixing rate in diesel and petrol

(Berti and Levidow, 2014). In 2005, the UK government announced the Renewable Transport Fuel Obligation (RTFO) which encouraged biofuel production from 2008 onwards. The RTFO motivated the biofuel industry to grow in the UK. For example, Vivergo Fuels in Kingston upon Hull, East Yorkshire, the UK's biggest producer of bioethanol, was started in 2007 as a joint venture between AB Sugar, BP and Du Pont to create a biorefinery for the future. In 2011 the production of bioethanol commenced, and by 2014 full production capacity of the plant was reached. However, in a setback for the emerging biorefinery industry in the UK, in September 2018 Vivergo ceased operating, citing "the Government's lack of pace over the past decade to introduce E10 [petrol blended with 10% bioethanol]." (https://vivergofuels.com/news/vivergo-fuelsbioethanol-plant-proposes-ceasing-production/). This unfortunate conclusion to the Vivergo initiative illustrates the economic fragility of biorefining in the face of unsupportive market and policy contexts, and the difficulty of bringing to fruition their potential contributions to the issues of energy security and climate change.

Prior to this latest development, in 2016 the Renewable Energy Policy Network for the 21st Century (REN21) released the Renewables Global Status report (Anon, 2016 a), which reported that in that year the world's renewable energy production had never been higher, suggesting efforts towards the transition to a more eco-friendly energy scenario were on course. However, the British Petroleum Statistical Review, also released in 2016, reported that the world's fossil fuel consumption had also never been greater (BPstats, 2016). The Renewables Global Status report showed that renewable energy production reached record levels in the year 2015, particularly from solar photovoltaics (PV), which had the most significant increase in renewable energy sector, with only a small growth rate of 3.9% in bioethanol production (Figure 1-1).



Figure 1-1. Average annual growth rates of renewable energy capacity and biofuels production, End-2010 to End-2015 (Anon, 2016 a).

The increase in renewable energy without decreasing fossil fuel consumption in part reflects the inability to directly replace fossil fuels. Despite the fact that electricity, provided by solar PV and wind, could be an attractive fuel for a small number of vehicles, unfortunately, it is not the case for the heavy duty and conventional transport applications which require a liquid type of fuel that biofuels, on the other hand, are able to deliver.

Henry Ford, as early as 1925, predicted the importance of biofuels (Agarwal et al., 2017):

"The fuel of the future is going to come from fruit like that sumac out by the road, or from apples, weeds, sawdust—almost anything. There is fuel in every bit of vegetable matter that can be fermented."

The United Kingdom has a total biofuel production capacity, bioethanol and biodiesel, of over 1,500 million litres per year (Alberici and Toop, 2013). Figure 1-2 shows the increase in cumulative production capacity of commercial scale bioethanol plants in the UK (operational and planned), from which it is seen that the bioethanol production capacity almost doubled in the year 2013 then decreased slightly in the subsequent years.



Figure 1-2. Actual bioethanol production in UK. Adapted from (Alberici and Toop, 2013; Phillips, 2017).

Despite the growth in production capability, UK bioethanol output itself has remained remarkably under capacity since 2010, particularly in 2011 when utilisation was only 17%, and the maximum utilisation percentage attained in this time period being just 65% (Alberici and Toop, 2013). However, the apparently increased production in 2016 allowed only a maximum blending of bioethanol in transport fuel of 3.3%, significantly below the aimed 10% target for 2020 (Phillips, 2017).

Moreover, Figure 1-2 shows that production capacity reached its maximum in 2013 at 900 million litres, then remained constant at 850 million litres in the years thereafter. The unchanging production capacity rate indicates the challenges facing the expansion of the bioethanol industry. Figure 1-3 presents the investments in clean energy globally. Clearly, investments in the solar and wind energy have dominated in recent years, whilst investments in biofuels have never been lower, reaching only \$7bn in 2017 compared to \$268bn investments in wind and solar energy.

Arabinoxylan, a potential co-product of integrated biorefineries



Figure 1-3. Global new investment in clean energy by sector (Louw, 2018).(\$bn/year)

In the UK, the investment scene in regard to clean energy sources is not dissimilar. Figure 1-4 shows the same global trend happening in the UK regarding growing investments in electrical power from wind and solar PV, with much less interest in biofuels.



Figure 1-4. New investment in clean energy in United Kingdom, by sector (Louw, 2018). (\$bn/year)

The decreasing investments in the biofuel industry are a result of the changing nature of the problems and obstacles hindering the growth of this industry. For decades, the biofuel business has been limited by multiple drivers, such as the fluctuations and rise in food commodity prices from which the "Food *vs* Fuel dilemma" ensued (Alberici and

Toop, 2013). In relation to demand, there are also several barriers regarding policies and markets; most importantly, the lack of clear, stable, long-term and profitable policy frameworks strongly stands in the way of the development of the biofuels industry (Anon, 2017). Moreover, the continued decrease in oil prices since 2014 has significantly raised the bar for biofuel production and led to the abandonment or interruption of some biofuel projects (Berti and Levidow, 2014); the focus shifted toward developing higher value applications rather than biofuels, as it has not been possible to compete with fossil fuels economically. The inability of biofuels to stand in the market as a financially viable business in the face of competition from fossil fuels and difficulties in overcoming the policy obstacles is the primary factor hindering the development of the biofuel industry, as illustrated by the closure of Vivergo.

Looking at a crude oil refinery, it is apparent that the number of commodities produced in such refineries contributes to their economic performance (Asche, Gjølberg, and Völker, 2003). Similarly in a biorefinery, a range of marketable products should be produced, alongside biofuel, to facilitate an economically-viable business that can compete against oil refineries. Also, oil refineries benefit from advanced and sophisticated process integration technologies which significantly enhance the overall efficiency and economics of oil refinery businesses (Lynd *et al.*, 2005; Lynd *et al.*, 2009; Martinez-Hernandez *et al.*, 2018). Production of a portfolio of products, such that "not a drop of oil is wasted", not only confers as much added value as possible to the whole feedstock, but also gives scope for highly integrated hence efficient and economic processes.

Although a portfolio of co-produced products is vital for biorefinery competitiveness, for generating a range of revenue streams and for providing integration opportunities as exploited within oil refineries. Not everyone working in biorefineries is as alert to the integration aspect as they are to that of revenue streams (Campbell *et al.,* 2018). Thus, for example, the definition developed by the International Energy Agency Bioenergy Task 42 Biorefinery (Anon, 2009) reads:

20

"Biorefinery is the sustainable processing of biomass into a spectrum of marketable products (food, feed, materials, chemicals) and energy (fuels, power, heat)."

This definition emphasises the "spectrum of marketable products" but makes no reference to integrated processes that allow economic co-production of these products. In this respect, a better definition is that given by the USA's National Renewable Energy Laboratory (NREL):

"A biorefinery is a facility that integrates biomass conversion processes and equipment to produce fuels, power and chemicals from biomass. The biorefinery concept is analogous to today's petroleum refineries, which produce multiple fuels and products from petroleum."

As Campbell et al. (2018) note,

"If a biological processing facility produces multiple products, this implies a complexity of operation that gives scope and opportunity for effective integration – but it does not imply that such integration is automatically and necessarily implemented. Unless this complex facility – with its multiple products – is deliberately integrated, it remains just a biological processing facility. It is not a biorefinery."

The NREL definition embodies the conceptual model for tomorrow's biofuel production where high-value commodities are produced, one of them being biofuel. This could be accomplished via biomass conversion and fractionation; Integrated production of biobased chemicals, heat, power and food within a biorefinery would present the ultimate economical option.

1.2 Arabinoxylan as a co-product of bioethanol production

In today's cereals biorefineries, the major co-product of bioethanol production is the Distillers Dried Grain with Solubles (DDGS), a relatively low-value product that goes to animal feed. DDGS come from the non-starch fraction of the cereals which mainly is the bran along with leftover protein and yeast biomass (Chatzifragkou *et al.,* 2015). Cereal

bran is the problematic fraction of cereals for biorefineries, as it comprises a large fraction of the raw material and is underexploited economically. Thus, extracting higher-value components of the bran has been a focus of attention towards producing added-value products.

Recently, studies have focused on the production of arabinoxylan (AX), a component of lignocellulosic biomasses, and a potential co-product of biorefineries that could provide a novel class of food ingredients and non-food products. DDGS composition makes it a potential source for extracting AX and arabinoxylan oligosaccharides (AXOS). Also, one process for producing arabinoxylan involves precipitating with ethanol, giving opportunities for integration with bioethanol production within biorefineries (Du *et al.*, 2009; Hollmann and Lindhauer, 2005; Misailidis *et al.*, 2009). Thus AX appears to be a particularly promising co-product for integration into a biorefinery to give both additional high-value revenue streams as well as scope for integration and the associated economic savings that define a true biorefinery.

Previous work has established that co-production of AX integrated with bioethanol could be economically feasible (Misailidis et al., 2009) and has established extraction protocols to produce AX from maize, wheat, and sugarcane bagasse (Bell 2015; Campbell et al., 2019; Peng et al., 2010; Rose et al., 2010). Further work has introduced bioethanol pinch analysis as a formal process integration tool to minimise the use of bioethanol in the AX extraction process and hence to reduce the cost (Martinez-Hernandez et al., 2013; 2018). However, the properties of extracts as functional food ingredients, as affected by source and extraction conditions, have not been extensively studied. Greater understanding of arabinoxylan functional properties is needed, as these properties vary according to feedstock and extraction conditions. Early studies on arabinoxylan extracts sometimes gave conflicting results due to differences in composition and purity of extracts (Biliaderis et al., 1995; Courtin and Delcour, 2002). For the purpose of undertaking extensive functional property investigations in food systems, AX extracts need to be available in large (kg) quantities. There is currently no commercial source of AX at this scale. Hence, to progress this line of research, there is a need to establish facilities for routine production of pure AX in kg quantities.

22

In addition, arabinoxylan oligosaccharides (AXOS) and xylo-oligosaccharides (XOS), the small degradation products of larger AX molecules, have become a further focus of interest, as they have prebiotic properties that could benefit both human and animal health. The production of a range of AX and AXOS/XOS products, precipitated using ethanol at different concentrations, also increases the scope for bioethanol integration to reduce the costs (Martinez-Hernandez *et al.*, 2018). Thus a scenario is envisaged in which a range of AX and AXOS/XOS products are produced, each adding revenue streams to enhance the economics of the biorefinery, while reducing processing costs through the efficiencies of integrated co-production.

A major hindrance to research on DDGS has been the inability to source the in-process material that eventually becomes the dry animal feed product, DDGS. Several studies have examined DDGS as a feedstock for further processing (Chatzifragkou *et al.*, 2015; Kosik *et al.*, 2017). However, this is not the material that would actually be used within a commercial process; materials for extraction would be accessed earlier in the process, prior to drying. Drying might alters the properties of the wet materials, these materials are called the Distillers Wet Grain (DWG) and the Solubles. However, existing bioethanol plants are closed systems that do not allow the in-process material to be sampled; this is why studies so far have utilised the more readily available DDGS.

A further hindrance has been the difficulty of quantifying the small oligosaccharides, due to a lack of standards and the inherent challenges of separating and detecting sugars.

Therefore, the current work undertook a wide-ranging investigation of pilot-scale production of AX materials and enzymatic studies of AXOS and XOS production from these materials. The work required development of a new method for separating and quantifying mixtures of mono- and oligo-saccharides, and applied the method to investigate the kinetics of enzymatic production of XOS using xylanases. The work thus addressed practical issues of AX and AXOS/XOS production as well as scientific issues of characterisation of the produced materials and understanding of xylanase kinetics.

1.3 Scope of the thesis

In conventional cereal-based biorefineries, the main by-product, DDGS, is a low-value co-product that goes to animal feed. The production of a portfolio of co-products is vital for the economic competitiveness of biorefineries. Recent studies have focused on the potential for production of arabinoxylan (AX), arabinoxylan oligosaccharides (AXOS) and xylo-oligosaccharides (XOS) within integrated biorefineries. The current work addresses issues of producing these materials at kg-scale and of characterising the products and the enzymatic degradation processes by which they are produced.

Arabinoxylan is the major hemicellulosic part of the cereal bran and a major component of DDGS. One of the processes for producing arabinoxylan uses ethanol for precipitation; co-production of AX with bioethanol production therefore gives scope for integrated processing and economic production of a portfolio of AX and AXOS/XOS products.

The evolving arabinoxylan story in the context of UK cereal biorefineries is presented in Chapter 2 along with the description of arabinoxylan structure, classification, physicochemical properties and the relation of these properties to the structure of the polysaccharide.

Chapter 3 describes industrial bioethanol production processes and illustrates the characteristics of the intermediate by-products of the fermentation and the effect of the drying on these intermediate products to give the final DDGS. The in-house bioethanol production procedure used in this project is also described, along with the analysis and characterisation of the products.

Chapter 4 introduces the concept of functional food ingredients and the biological importance of arabinoxylan prebiotics for human and animal nutrition. It also describes the role of xylan-hydrolysing enzymes in the in-situ production of XOS in animal feed, and presents trials performed in this project to produce XOS products from the bioethanol by-products.

Chapter 5 details the methods used for AX extraction from cereal biomass on small and large scales, and reports the large-scale extraction of AX soluble fibres from bioethanol by-products.

Having shown that characterisation of the products is a challenge and a limitation to research in this area, Chapter 6 reviews analytical methods used for the identification and quantification of arabinoxylan oligosaccharides, then details the development and validation of a powerful new analytical method for the simultaneous characterisation of mono- and oligosaccharides and uronic acids.

Chapter 7 describes arabinoxylan hydrolysing enzymes (xylanases), their classification, industrial applications and potential role in the production of AX prebiotics. Econase is a commercial endoxylanase widely used in animal feed formulations, which is believed to exert some of its benefits through production of prebiotic oligosaccharides, and for which there is a need to understand the kinetics of oligosaccharide production. The chapter presents experiments to evaluate the effects of this xylanase on XOS production and degradation.

Chapter 8 discusses the importance of the integration of AX production within biorefineries and reports the integration carried out in this project to produce two types of AX soluble fibres and the potential conversion of the AX into XOS using xylanases.

Chapter 9 concludes the thesis by summarising the main findings from the current work and presenting recommendations for progressing research in this area, in order to lead to a sufficient basis of knowledge to allow commercial production and exploitation of AX-based products within integrated biorefineries.

2 Arabinoxylan structure and physicochemical properties

2.1 Introduction

Hemicellulose is the second most abundant polysaccharide in nature after cellulose and, together with cellulose and lignin, forms the structural component of plant cell walls (Bacic *et al.*, 1988; Izydorczyk and Biliaderis, 1995). Hemicelluloses are polysaccharides of mainly 5-carbon sugars, the composition of which varies according to the plant species. In cereals, arabinoxylans (AX) are the major hemicelluloses, comprising, for example, 20-25% of wheat bran (Bacic and Stone, 1981). AX molecules consist of a linear chain of xylose with arabinose side chains and some phenolic acids attached to the arabinose branches. AX are novel polymers that possess unique functional properties that could be useful as a food ingredients or for non-food uses, offering the possibility of producing a portfolio of AX-based co-products within cereals biorefineries. This chapter reviews AX structure, classification and physicochemical properties, in order to provide the context for the objectives of the current work.

2.2 Brief background of arabinoxylan

At the beginning of the twentieth century, Hoffmann and Gortner (1927) reported a gummy non-starch polysaccharide from wheat flour which was comprised mainly of xylose and arabinose, both 5-carbon sugars; hence the new gum was named 'pentosan'. Later, the novel pentosan gum was described as consisting of a simple type of molecule made of the association of arabinose and xylose in cereals (Preece and Hobkirk, 1953). Subsequent research reported the presence of AX in the outer layers of cereal grains such as wheat, corn, rice, barley, oats, rye and sorghum (Vinkx and Delcour, 1996). AX have received increasing attention in the field of cereal science and bio-materials in the last few decades. A bibliographic search of ScienceDirect conducted in August 2018 for "Arabinoxylan" in the title, abstract or keywords shows that publications on this topic have increased nearly ten-fold in the last twenty years (Figure 2-1).



Figure 2-1. Number of publications about arabinoxylan. Source: ScienceDirect (August 2018).

The available literature describes mostly the characteristic properties of AX, the extractability on a bench-scale and the potential applications. Only a limited number of papers report larger-scale extraction, as extraction at commercial scale, or even pilot scale, is hampered by two factors: (i) the extraction requires large quantities of ethanol, which is costly and has handling implications in relation to safety and security. (ii) The absence of a clear market to commercialise AX-based products. These facts have been the reason for AX to remain only a theoretical material that might have a promising future. Thus, for example, Courtin and Delcour (2002) observed:

"In spite of early reports of their beneficial impact in breadmaking, AX have remained only a theoretical candidate for addition to the bread-recipe. This is related to the fact that studies of their functionality have sometimes yielded contradictory results and to their unavailability on a commercial scale."

In order to stimulate markets for AX, a substantial number of studies have focused on the benefits of AX in the fields of human health and prebiotic properties, breadmaking and the food industry, excipients in the pharmaceutical industry and food packaging. Table 2-1 summarises the small scale experiments for potential uses of arabinoxylan. However, the lack of commercial-scale production of AX is the main hindrance, as sufficient amounts are not available to demonstrate the technological uses of AX and promote their uptake in industry.

Field of study	Effect of AX	Reference
Breadmaking	-Increased the loaf volume	(Biliaderis <i>et al.,</i> 1995; Li <i>et</i>
	-Higher moisture content	<i>al.,</i> 2013)
Health	-Metabolic improvement during	(Akpinar <i>et al.,</i> 2010; Garcia
	diabetes	et al., 2007; Geraylou et al.,
	-Lower postprandial responses in serum	2013; Hughes et al., 2007;
	glucose, insulin and triglycerides	Lopez <i>et al.,</i> 1999; Lu <i>et al.,</i>
	-Lower blood cholesterol levels	2004; Staneva et al., 2014;
	-Obesity preventing	Tong <i>et al.,</i> 2014; Van
	-Prebiotic	Craeyveld <i>et al.</i> , 2008; D.
		Zhang <i>et al.,</i> 2015)
Food packaging	-Barrier for oil and fat	(Mikkonen <i>et al.,</i> 2009;
		Stepan <i>et al.,</i> 2014;
		Stevanic <i>et al.,</i> 2011)
Pharmaceuticals	-Stable emulsions	(Bashir <i>et al.,</i> 2014; Erum <i>et</i>
	-Foam stabilising	<i>al.,</i> 2015; Iqbal <i>et al.,</i> 2011;
	-Sustained release matrix	Izydorczyk <i>et al.,</i> 1991; Raja
	-Suspensions stabilising	et al., 2014; Yadav et al.,
	-Disintegrating agent	2007)

Table 2-1 Potential uses of AX.

2.3 Structure of arabinoxylans

Arabinoxylan is the main hemicellulose polysaccharide in cereals; it exists mainly in the outer layer and endosperm cell walls and is associated with cellulose and lignin (Izydorczyk and Biliaderis, 1995). AX comprise a linear backbone of β -D-xylopyranosyl linked via (1-4) glycosidic linkages. Izydorczyk (2008) reported that monomers of α -L-arabinofuranoside are linked to some of the xylopyranosyl chain at O-2, O-3 and/or both O-2, 3 positions. According to this conformation four main structural units of AX can be observed as illustrated in Figure 2-2: mono-substituted xylopyranosyl at O-2 or O-3, di-substituted xylopyranosyl at both O-2,3 and unsubstituted xylopyranosyl (Izydorczyk and Dexter, 2008; Mendis *et al.*, 2016).



Figure 2-2. Structure of AX, showing mono- and di-substituted xylose units and a ferulic acid link (Mendis et al., 2016).

The sequence and distribution of these four conformational structural units depends on the source of arabinoxylan (Gruppen *et al.*, 1993). Izydorczyk (1995) reported that the majority of arabinose side chains consist of one monomeric substitute, and just a minor proportion could contain two or more arabinose units linked via 1-2, 1-3, or 1-5 linkages (Izydorczyk and Biliaderis, 1995). The amount of arabinose in AX indicates the "embranchment" of the chains and is usually estimated by the Arabinose/Xylose (A/X) ratio, which varies between different plants and also between the different parts of a plant.

The terms "branching" and "embranchment" are widely used in the AX literature to refer to the arabinose substitutions (*e.g.* (Hoffmann *et al.,* 1992; Rattan *et al.,* 1994; Cyran and Saulnier, 2012; Coelho *et al.,* 2016)). These are not actually branches, suggesting a side-chain – the phrase "arabinose side chain" is also used, inaccurately suggesting a chain of arabinose units, so these terms are perhaps not the most helpfully precise words, but to be consistent with other literature, these terms are used in this thesis.

For example, the A/X ratio of the wheat endosperm AX is reported around 0.7 (Rattan *et al.*, 1994), whereas the ratio is greater in wheat bran and reaches a value of 1.07 (Shiiba and Nagao, 1993), and a ratio of 0.21 in wheat straw (Xu *et al.*, 2006). AX was reported to have A/X ratio of 2.6-4.1 in rye grain (Hansen *et al.*, 2003) and a ratio of 1.7 in rye bran (Figueroa-Espinoza *et al.*, 2004).

Phenolic acids such as hydroxycinnamic, ferulic and *p*-coumaric acid might appear esterified to O-5 of arabinofuranoside units in AX (Smith and Hartley, 1983). Ferulic acid, the most abundant phenolic acid in cereals, can form dimers of dehydrodiferulate esters, as shown in Figure 2-3, forming a covalent cross-linkage between AX chains and between AX and other cell wall components (lignin, proteins, etc.) (Santiago *et al.*, 2006; Santiago and Malvar, 2010).



Figure 2-3. Dehydrodiferulate esters (Ferulic acid dimer) (Santiago and Malvar, 2010).

Due to the hydrogen bonds between adjacent xylose units in the unsubstituted xylan polymers, a conformation of a three-folded and left-handed helix is observed where each group of three xylose sugars forms a turn in the helix and the final conformation may look like an extended twisted ribbon (Courtin and Delcour, 2002; Fincher and Stone, 1986; Izydorczyk and Biliaderis, 1995). Arabinose substitutions can distort this conformation, as the side units can sterically hinder this formation and a semi-flexible random coil conformation is proposed (Courtin and Delcour, 2002; Dervilly-Pinel *et al.,* 2001). Moreover, the ferulic acid content in the AX chains may form linking bridges between adjacent AX chains, resulting in higher molecular weight molecules, and may change the conformation and hence the properties.

2.4 Classification of arabinoxylans

Variations in the structural configuration of AX have been the main driver for dividing the polysaccharide into two main categories: water-extractable arabinoxylan (WEAX) and water-unextractable arabinoxylan (WUAX) (Vinkx and Delcour, 1996). WEAX are loosely attached to the cell wall and can easily migrate into aqueous solutions; they are the shorter chains with less cross-linking and less linkage to the other cell wall components (Izydorczyk and Biliaderis, 1995). WUAX are the harder to extract AX, they have much larger molecular weights and greater branching (*i.e.* arabinose substitution), and are more strongly attached to the cell wall matrix by covalent (*i.e.* ester and ether bonds, diferulic acid bridges) and non-covalent (*i.e.* hydrogen bonds) linkages to adjacent AX and other cell wall constituents (Fengler and Marquardt, 1988; Izydorczyk and Biliaderis, 1995).

2.5 Occurrence of arabinoxylans

AX are one of the major hemicelluloses in the plant kingdom, occurring in a wide variety of plants. They play an essential role in strengthening the structural support system by cross-linking the cellulose microfibrils and thereby regulating the cell expansion and strengthening the wall (Gibeaut *et al.*, 1991; Scheller and Ulvskov, 2010). Hemicelluloses are synthesised in the Golgi apparatus (Carpita and Gibeaut, 1993). Briefly, the synthesis of AX goes through four main steps: xylan chain initiation, xylan chain elongation, arabinose side chain addition and extracellular deposition (Jung *et al.*, 1993). Glycosyltransferases are the enzymes responsible for the biosynthesis of heteropolysaccharides such as AX; β -1,4-xylosyltransferase was found to be the enzyme catalysing the biosynthesis of the xylan backbone whereas AX-arabinosyltransferase adds arabinose side chains (Gibeaut and Carpita, 1991; Porchia *et al.*, 2002).

2.6 Physicochemical properties of Arabinoxylans

The reported physicochemical attributes of AX are primarily credited to its structural characteristics, *i.e.*, the length of xylan backbone, the degree of embranchment (the A/X

ratio) and the oxidative cross-linking by ferulic acid (Courtin and Delcour, 2002; Izydorczyk and Biliaderis, 1995). The main physicochemical properties of interest are water solubility, viscosity, water holding capacity and oxidative gelation.

2.6.1 Water solubility

In general, the solubility of AX in water is subject to the structure of the molecule, *i.e.* chain length, the degree of substitution (DS, equivalent to the A/X ratio), and the substitution pattern (Courtin and Delcour, 2002; Saulnier *et al.*, 2007). The degree of substitution seems to be the key factor affecting AX solubility; higher DS gives greater solubility in water, as arabinose "side chains" (more accurately, arabinose substitutions) prevent the intermolecular aggregation of unsubstituted xylose chains, while removing the side chains enzymatically usually results in lower solubility (Andrewartha *et al.*, 1979; Courtin and Delcour, 2002). Molecular weight (Mw) also plays an essential role in the solubility, with lower Mw AX exhibiting greater solubility (Mares and Stone, 1973; Courtin and Delcour, 2002; Saulnier *et al.*, 2007). Bian *et al.* (2010) applied a gradual precipitation on AX extracts using a range of ethanol concentrations of 10, 20, 30, 45, 60 and 80%. The high Mw AX with fewer branches precipitated first, whereas the lower Mw with more branching needed higher concentrations of ethanol to precipitate.

2.6.2 Viscosity

AX solutions are viscous and tend to exhibit pseudo-plastic behaviour due to the relatively high molecular weight and the stiff semi-flexible random coil conformation. Several factors affect the viscosity of AX solutions including Mw, A/X ratio, pH, AX concentration and ferulic acid content (Girhammar and Nair, 1992; Courtin and Delcour, 2002). Girhammar and Nair (1992) reported that higher Mw AX gave more viscous solutions, and that more viscous solutions were obtained in neutral pH; they noted it is possible that acidic and alkaline pH values break down the ester cross-linkages or even hydrolyse the polysaccharide chains in extreme pH conditions. Ferulic acid content significantly affects the solution viscosity as it may create cross-links between AX chains,

32

forming higher Mw molecules (Courtin and Delcour, 2002; Dervilly-Pinel *et al.*, 2001; Saulnier *et al.*, 2007).

2.6.3 Water Holding Capacity

Water holding capacity is one of the most attractive attributes of AX, particularly for bakers, for example. Courtin and Delcour (1998) added AX extracts to bread dough and quantified the dough behaviour by farinograph; they concluded that WUAX are capable of holding 7-10 times their weight in water, whereas WEAX presented lower water-binding capacities at 4-6 times their weight.

2.6.4 Oxidative Gelation and Gel-Forming Capacity

Izydorczyk *et al.* (1990) demonstrated that oxidising agents (*e.g.* hydrogen peroxide or peroxidase) could induce oxidative gelation in AX solutions, by catalysing the formation of covalent cross-linkages between AX molecules by the dimerisation of ferulic acid residues. Figure 2-4 describes the oxidative coupling of ferulic acid.



Figure 2-4. Oxidative coupling of ferulic acid (Döring et al., 2015).

These cross-linkages augment the viscosity and lead eventually to the formation of a gel (Dervilly-Pinel *et al.*, 2001a; Vinkx *et al.*, 1991). The high Mw, high ferulic acid and the less substituted xylan chain (low A/X) are the primary precursors of substantial crosslinking (Izydorczyk and Biliaderis, 1995). However, the rigidity of the gel is subject to the AX intrinsic viscosity, rather than the content of ferulic acid residues (Dervilly-Pinel *et al.*, 2001a). Other non-covalent linkages, such as hydrogen bonding and van der Waals, also contribute to the gel strength (Saulnier *et al.*, 2007).

2.6.5 Foam stabilising

Arabinoxylans possess emulsifying properties and the ability to stabilise protein films against thermal disruption as they increase the viscosity of the interlamellar liquid, alter the draining properties of the foam films and mediate interactions between proteins in the absorbed layer (Izydorczyk *et al.*, 1991; Sarker *et al.*, 1998; Xiang and Runge, 2016).

2.7 Research objectives

In the context of exploring strategies for enhancing the economic income from biorefineries, the integrated biorefinery approach is an effective basis for exploiting interaction opportunities arising from the co-production of several products. Biorefineries are currently derived principally from ethanol production, and ethanol can serve as a precipitating agent for arabinoxylan, which makes arabinoxylan a natural co-product of bioethanol that offer the opportunity to bring a new class of functional food ingredients to the market.

The primary motivation of the current work was identified earlier as the pilot-scale production of arabinoxylan and arabinoxylan oligosaccharides within integrated biorefineries. In order to investigate the opportunity, several objectives were identified (some of these coming to prominence as the project progressed):

(i) to perform the bioethanol fermentation process at pilot-scale to produce and characterise the wet intermediate by-products, *i.e.* DWG and Solubles.

(ii) to investigate the direct enzymatic production of AX prebiotics from the wet byproduct, DWG.

(iii) to extract AX chemically by means of the alkaline oxidative method on a pilot-scale from the DWG, and to characterise the extracts.

(iv) to develop a robust analytical method for the simultaneous determination of monosaccharides, xylo-oligosaccharides, arabinoxylan oligosaccharides and uronic acids.

(v) to study the enzymatic production and simultaneous degradation of XOS.

(vi) to produce and characterise water-extractable and water-unextractable AX within the biorefinery, and to evaluate the potential for production of XOS from these extracts.

2.8 Summary

Arabinoxylan classification and functional properties are determined by the average molecular properties such as the molecular weight, the degree of substitution and cross-linkage. Smaller AX molecules are more soluble in water and contain fewer ferulic acid cross-linkages and are usually classified as the water-extractable AX, whereas the larger molecules with high branching and more cross-linking are less soluble in water and are usually referred to as water-unextractable AX. Broadly speaking, the highly substituted WUAX shows greater water holding capacity and has the ability to form solutions of higher viscosity. Reducing the arabinose substitutions reduces both solubility and solution viscosity.

As noted above, while the scientific understanding of AX is extensive and detailed, its technological understanding has been hampered by lack of availability of larger than bench-scale quantities. Thus, one of the objectives of the current work was to produce significant quantities of AX using the School of Applied Sciences' pilot scale bioethanol plant. The production of the equivalent of industrial-scale Distillers Wet Grains (DWG) and Distillers Solubles (DS), from which to extract AX, is described in the next chapter.
3 Pilot-scale production of biorefinery by-products

3.1 Introduction

Bioethanol has become in recent decades the most important biofuel for transportation and the basis for the emergence of biorefineries. As noted in Chapter 1, the characteristics of biorefineries include a portfolio of co-products, and deployment of process integration to achieve efficiencies that will help biorefineries to become competitive with oil refineries. AX is a particularly promising candidate as a co-product with bioethanol, as ethanol is used to precipitate the AX, giving scope for integration and making production of AX economically feasible. The rise of biorefineries, giving the opportunity to create commercial sources of AX-based products, also requires the cocreation of markets for these products in the food, nutraceutical and pharmaceutical markets.

In order to successfully integrate AX production within biorefineries, the relevant streams of the bioethanol process, Distillers Wet Grain (DWG) and Distillers Solubles (DS), need to be characterised for their AX content and properties, and for the ease of recovering AX from these streams. However, the design of current bioethanol plants in the UK does not allow ready access to these in-process streams to take samples. Therefore, this part of the project aimed to operate the GUNT CE-640 bioethanol plant to produce material similar to the DWG and Solubles streams in industrial plants. This chapter reviews industrial bioethanol production and describes the operation of the GUNT pilot-scale process to produce and characterise DWG and DS for further investigations.

3.2 Bioethanol biorefineries – past, present and future

Alcoholic fermentation is one of the first bio-processes that early humankind exploited. Archaeologists report that fermented beverages existed in the Neolithic period (7000 BC) in China, Mesopotamia and ancient Egypt (McGovern *et al.,* 1996; Samuel, 1996; Steinkraus, 1997), with fermentation initiated by naturally occurring yeasts (Sicard and Legras, 2011). In China, the beverages were primary made from rice, whereas Mesopotamia and Egypt fermented fruits, honey and malt, while the utilisation of grapes was reported in Mesopotamia about 5400 BC (McGovern *et al.,* 1996). It is not documented exactly when and where fermented foods were initiated, but it is evident that fermentation has given humanity foods and drinks with pleasant flavours, aromas and textures and, in the case of drinks, with intoxicating properties.

Nowadays, fermentation continues to provide humankind not only with food and drinks but with fuel as well. The process of bioethanol production is not much different from the fermented beverage production process, except on a larger scale and without the aromatic and flavour concerns of beers and wines, for example.

In 2017, global bioethanol production reached about 100 billion litres, half of which was in the United States (Chatzifragkou *et al.*, 2015). About 95% of the world's bioethanol production is in the USA, China and Europe, utilising starch as the starting material (OECD/FAO, 2015). The production is chiefly from corn and soybeans in USA, sugar cane in Brazil, corn in China, and sugar beet, wheat and barley in Europe (Stein, 2007), with the production process varying depending on the feed material. For bioethanol production from cereals, there are two main processes: the wet milling process and the dry milling process.

3.2.1 Wet milling

Figure 3-1 illustrates the corn wet milling process. In this process, the corn is fractionated into relatively pure components (starch, gluten, oil) and germ meal and bran fractions that go to animal feed, before the starch is then hydrolysed to glucose and fermented to ethanol. Briefly, a first screening is applied to remove the foreign materials (crop residues, broken kernels, etc.), then the grains are soaked in large steeping tanks with a diluted solution of sulphur dioxide for 48 hours. The germ is separated from the soaked kernel and goes to oil production, and the leftovers are used

for feed and called the corn germ meal. The bran is separated by passing the slurry through sieves, leaving the starch and the gluten in the solution. The gluten is separated and dried to form the corn gluten meal (Blanchard, 1992; Elmekawy *et al.*, 2013; Stock, Lewis, Klopfenstein, and Milton, 2000). The process produces a range of relatively pure and high value products; however, this requires good quality feedstocks (first or second grade corn) and a sophisticated and expensive process requiring high capital investment (Belyea *et al.*, 2010).



Figure 3-1. Corn grain wet milling process.

3.2.2 Dry milling

In this process, the grain or a mixture of grains is milled without applying any fractionation prior to the fermentation; the entire milled grain then undergoes the fermentation process, and the starch portion is converted into ethanol and CO₂, Figure 3-2 shows a simplified process. The fermented mash goes through various separation

processes to separate the ethanol from the bulk of the fermentation media; the nonfermented residues are dried to give the main by-product, Distillers Dried Grain with Solubles (DDGS), which goes to animal feed and contributes a significant proportion of the biorefinery's revenues. In some processes the CO₂ might be captured and sold as an additional revenue stream.



Figure 3-2. Dry milling process for bioethanol production.

The main advantage of this process, compared with the wet milling process, is the ability to process different types of grains with limited restriction on the quality of the grain (Stock *et al.*, 1999). About 82% of bioethanol production worldwide uses the dry milling method (Wood *et al.*, 2012). The prevalence of this process is resulting in increasing amounts of DDGS coming into the animal feed market, which gives issues over the nutritional value and performance of DDGS for different types of livestock (cattle, pigs, poultry and fish).

3.3 Distillers dried grains with solubles

DDGS has historically been a by-product of the alcoholic beverage industry and a rich source of energy, protein, minerals and vitamins for animal feed (Cromwell *et al.*, 1993; Klopfenstein *et al.*, 2008). Recently, the surge in dry milling bioethanol production for fuel has resulted in much greater quantities of this by-product entering the animal feed

market. The availability of DDGS is expected to increase even more in the future. Exploiting DDGS in the animal feed market contributes significantly to the profitability of the biorefinery (Klopfenstein *et al.*, 2008). However, DDGS is still a relatively low-value product, and better economical exploitation of this material would significantly assist the development of the bioethanol industry.

Extraction of the AX offers a higher value product, but also offers a further benefit; AX is dietary fibre, which is detrimental to conversion of animal feeds into weight gain (humans should eat more fibre as, in general, we need to avoid weight gain, but animal nutrition is the opposite, where rapid weight gain is desired by livestock producers). Therefore reducing the fibre content of DDGS by extracting the AX would enhance the value of the remaining reduced-fibre DDGS. Cattle can cope with high fibre in their diet, but pigs and poultry require less fibre in the diet – removing AX from DDGS would potentially allow greater use of a DDGS in animal feed formulations for pigs and poultry.

DDGS is defined by the Association of American Feed Control Officials (Świątkiewicz and Koreleski, 2008) as:

"Distillers Dried Grains with Solubles is the product obtained after the removal of ethyl alcohol by distillation from yeast fermentation of a grain or a grain mixture by condensing and drying at least ¾ of the solids of the resultant whole stillage by methods employed in the grain distilling industry."

The definition reflects the production process for DDGS, which is illustrated in Figure 3-3. Briefly, the milling, liquefaction and saccharification of the whole grain are achieved by thermal and enzymatic hydrolysis to facilitate the breakdown of starch into fermentable glucose. Then, the yeast ferments the available sugars into ethanol and carbon dioxide. Following fermentation, the ethanol is separated by distillation and dehydrated using molecular sieves. The non-fermentable and non-volatile components are separated from the slurry by filtration and centrifuging to yield a liquid fraction called thin stillage (with a solids fraction of around 5%) and a solid fraction called Distillers Wet Grains (DWG). The thin stillage is evaporated into Distillers Solubles

(typically with a solids content of about 12% (Morey *et al.,* 2009)), which is added to the DWG and dried at high temperatures to produce the final DDGS (Jensen *et al.,* 2012; Kingsly *et al.,* 2010).



Figure 3-3. Dry-grind ethanol production processes and by-products (*simplified, adapted from Kingsly et al., 2010*).

The chemical composition of DDGS is similar to the original grain's non-starch components (it also contains yeast material from the fermentation). Variations in the raw material give variations of the composition of the final DDGS, and variations are reported among bioethanol plants and between different batches from the same production plant (Liu, 2011). The effect of the variation in the raw material (wheat, barley, maize or mixtures) has the clearest impact, but variations in grain growing, harvesting and handling also have notable impacts (Cromwell *et al.*, 1993; Kim *et al.*, 2007; Liu, 2008). Processing conditions such as the drying method and temperature also affect the final DDGS (Chatzifragkou *et al.*, 2015). Table 3-1 shows the different chemical compositions of wheat DDGS (Jarret *et al.*, 2011), maize DDGS (Spiehs *et al.*, 2002) and a blend of wheat, triticale, barley and rye DDGS (Pedersen *et al.*, 2014). Clearly, the difference in the nutrient composition of the parent grain is reflected in the DDGS; the oil in maize DDGS is greater than in wheat DDGS, whereas the latter contains a slightly greater amount of crude protein.

	Maize DDGS	Wheat DDGS	blend DDGS		
Dry matter	87.2–90.2	89.3–94.4	90.5–92.7		
Oil	10.2–11.4	3.6–5.6	8.1–12.8		
Protein	28.7–31.6	32.6–38.9	23.4–27.9		
Crude fibre	8.3–9.7	6.2-10.9	9.6–10.6		
Ash	5.2-6.7	4.3-6.7	3.4–7.3		

Table 3-1. Composition of DDGS from different plants and sources (expressed on a %, drymatter basis) (Jarret et al., 2011; Pedersen et al., 2014; Spiehs et al., 2002).

Generally, after the removal of the starch by fermentation, the remaining components of the grain are concentrated by a factor of around three (Rasco *et al.*, 1987). The rich protein and fibre content in the DDGS opens several exploitable possibilities other than animal feed, including the degradation of the polysaccharides for more ethanol production, production of value-added products, a source for food ingredients or use in bio-composites (Thacker and Widyaratne, 2007). However, the successful integration of any valorisation process of DDGS within biorefineries would need to be performed on the actual wet material, *i.e.* DWG and or DS, as the final dry material is not amenable to further processing (without redispersing it in water), and drying might change the characteristics of the material. The dry DDGS is easy to obtain, and a number of studies have investigated extraction from DDGS (Chatzifragkou *et al.*, 2015; Kosik *et al.*, 2017), but studies on the wet DWG would be more realistic both in terms of the characteristics of the material and the practicalities of extracting from it in the industrial context.

3.3.1 DDGS and DWG

Historically, there were two by-products of the bioethanol production process, the Distillers Dried Grain (DDG), and the Distillers Solubles (DS) (Thacker and Widyaratne, 2007). Current bioethanol plants tend to mix the spent grain and solubles during drying to produce the combined product, DDGS. In some cases, producers have arrangements with local livestock businesses that allow a moist product to be produced, as it is transported and consumed by animals before it can deteriorate, but in general a dry product is needed to give longer stability.

Figure 3-3 shows the flow of the ethanol production process in a dry-grind wheat biorefinery, where the mash is separated after distillation into the DWG (~60-70% moisture) and the thin stillage (~95% moisture) which contains all the solubles (Morey et al., 2009). The DWG and thin stillage undergo severe thermal treatment during the drying process, which alters the components of the DDGS. As noted section 3.3, extraction from DDGS, as well as being inappropriate in the industrial context, would yield products of different (and probably inferior) functionality compared with extraction from DWG. The severe thermal treatment has a major effect on the nutritional value of DDGS proteins (De Vries et al., 2013). Schroeder (2012) reported that the proportion of undegradable protein increased from 47% in the DWG to 54% in the DDG. The effect of the thermal treatment is likely to affect other components such as the cell wall polysaccharides. For instance, AX in the DDGS was reported to have a significantly lower A/X ratio as compared to the original grain AX as well as a higher proportion of water-extractable AX (Pedersen et al., 2014). These structural changes and changes in AX solubility are possibly a result of alteration in the molecular structure and/or the molecular cross-linking with other components in the cell wall (Kosik et al., 2017).

Hence, although some research in this area has investigated DDGS as a source of higher value materials, the current work started by producing a pilot-scale equivalent of DWG, to give a more representative material for subsequent studies of AX extraction and enzyme treatment.

3.4 Materials and equipment used for pilot-scale bioethanol and DWG production

Bioethanol and DWG were produced from milled wheat using the GUNT CE-640 bioethanol plant.

3.4.1 Chemicals and enzymes

Wheat was sourced from Target Feeds Ltd (Shropshire, UK), who provided the wheat double-milled through a 2 mm mesh. The enzymes used for starch hydrolysis, α -

amylase (SCHLIESSMANN-VF-) and gluco-amylase (SCHLIESSMANN-VZ-), the antifoaming agent (SCHLIESSMANN-EX-PROTIN) and dry baker's yeast from the strain *Saccharomyces cerevisiae* (KORNBRAND PREMIUM), were purchased from ReKru GmbH, Germany. Sodium hydroxide pellets and sulphuric acid (>95%) laboratory reagent grade were obtained from Fisher Scientific Ltd.

3.4.2 The CE-640 bioethanol plant

The CE-640 unit supplied by GUNT Technology Ltd. (Hamburg, Germany) is designed to imitate the industrial processes of ethanol production and facilitates the in-process sampling of the processed material through all the stages of the process. (https://www.gunt.de/en/products/process-engineering/biological-process-engineering/anaerobic-processes/biotechnical-production-of-

<u>ethanol/083.64000/ce640/glct-1:pa-148:ca-251:pr-69</u>). It comprises three main units: the mash tank, the fermentation tank and the distillation unit. Figure 3-4 shows an overall view of the unit components. The plant is controlled and monitored via the integrated programmable logic controller displayed on the right of Figure 3-4. The plant delivered to the University of Huddersfield features several modifications from the standard unit supplied by GUNT, to make it specifically suited to the AX extraction work: pumps capable of handling larger particulate solids, pH and full temperature control to both the mash and fermentation tanks, and corresponding control software modifications.

The mash tank is used for the liquefaction and saccharification of the starting materials, with a capacity of 40 litres. It is a double-jacketed stainless-steel vessel equipped with speed-regulated stirrer comprising a geared motor and a pitched blade on the shaft. The material in the mash tank is heated by direct steam injection through a jet, or via the double-jacket heating. The pH is monitored by an integrated pH probe and controlled by acid/base supply with diaphragm metering pumps.



Figure 3-4. GUNT CE-640 Bioethanol plant, general view.

The fermentation unit has a capacity of 60 litres, with all the control specifications of the mashing tank. In addition, the tank is sealed airtight, and the stirrer shaft runs into the container through a fermentation lock, which employs water as a sealing liquid. The generated CO_2 during the fermentation pearls up as gas bubbles through the sealing liquid of the fermentation lock.

The distillation unit is a modified boiler heater with a water bath. It contains a bubble cap tray column, dephlegmator (a device for giving partial condensation of a multicomponent vapour stream) and condenser.

The process diagram, Figure 3-5, presents all the components and measurements on the unit, as well as the pipe connections, supply lines and the manually operated valves.



Figure 3-5. Process diagram of GUNT CE-640 Bioethanol plant.

3.5 Methods for bioethanol and DWG production

Producing the DWG product requires a fermentation process that mimics the relevant industrial conditions. The process employed for producing ethanol and DWG in the GUNT plant is described below. The fermentation was followed by the characterisation of the starting and end materials; the analytical methods used are also described.

3.5.1 Fermentation method

In order to produce a fresh fermentation by-product, DWG, that is representative of the industrial by-product, industrial production conditions should be followed. However, it was difficult to ascertain exact industrial fermentation conditions as this information is commercially sensitive and also varies between production plants. Therefore, aiming to produce a representative DWG that has the characteristics of the material produced in the large-scale biorefineries, the production conditions were set according to the recommendations of the provider of the pilot scale fermentation unit, which were similar to the conditions reported in the literature.

Figure 3-6 details the flow of the DWG production process carried out in this project. Wheat was double-milled using a 2 mm mesh by Target Feeds Ltd.; however, this left a substantial proportion of large particles and intact grains which would have been incompletely fermented. The milled wheat was therefore sieved on a 1.4 mm mesh, and the large material milled using a Retsch ZM1000 mill (Retsch GmbH, Germany) without a screen, re-sieved (again on a 1.4 mm mesh) and the material passing through then thoroughly remixed with the original smaller material, while the oversized material was re-milled.

A batch of 6 kg of milled wheat was dispensed into the mash tank with 15 L water; direct steam injection heated the mash to 85°C. The total water addition was 25 L, corresponding to a solid:liquid ratio of 1:4. Starch hydrolysis was initiated by adding 1.2 mL of α -amylase (20 mL per 100 kg) with vigorous stirring at 200 rpm for one hour. The temperature of the mash was reduced to 55°C, and pH adjusted to 4.5-5 with sulphuric acid. Further hydrolysis was accomplished by adding 3.6 mL of gluco-amylase (60 mL per 100 kg) with continuous stirring for 2 hours. The mixture's temperature was reduced to 28°C, and the mash was pumped to the fermentation vessel. Yeast (6 g, equivalent to 100 g per100 kg) was added to the mash to start the fermentation, along with 0.9 mL anti-foaming agent (15 mL per 100 kg). The mash was fermented over 72 hours with continuous control of temperature in the range 25-28°C. The end of the fermentation is recognised by the ceasing of CO₂ release from the water seal, as illustrated in Figure 3-7.

After 72 hours, the mixture was pumped to the distillation column and distilled for 4 hours to separate the ethanol, then allowed to cool to room temperature overnight. The remaining slurry was filtered through a cloth to obtain the DWG, and the filtrate — the Solubles — was dried in an oven to recover the dry solubles. Both DWG and Solubles were stored in a freezer for further experiments.

47



Figure 3-6. Process of DWG production using the GUNT CE-640 bioethanol plant.



Figure 3-7. CO₂ release from the water seal during fermentation.

3.6 Characterisation of the by-products

A series of tests and analyses were carried out to quantify the yields and characterise the materials produced and the changes in the distribution of sugars after fermentation.

3.6.1 Ethanol purity

Since the produced ethanol is not the primary interest of this study, the tests carried out on the ethanol were limited to the estimation of the amount and the concentration using an alcohol hydrometer. The School of Applied Sciences was required to obtain a distillery licence for the GUNT bioethanol plant from HM Revenue and Customs; in accordance with the terms of the licence, all produced ethanol was recorded and stored in our "warehouse" (a locked cupboard).

3.6.2 Constituent sugars analysis

Samples of the raw material (wheat) and the produced materials (DWG and Solubles) were freeze-dried and sent to Englyst Carbohydrate Ltd. (Southampton, UK) for constituent sugar analysis, from which arabinoxylan contents could be calculated. (Later we were able to do these analyses ourselves, but in the early stages of the project we did not have access to the necessary facilities, so made use of this external lab.)

Measurement of the non-starch polysaccharides (NSP) was carried out after enzymatic removal of starch and acid hydrolysis of the NSP to their constituent sugars. For hydrolysis, samples were subjected to treatment with 12M sulphuric acid for 30 min at 35°C followed by 2M sulphuric acid treatment for 1 hour at 100°C. Thereafter, sugars were separated and measured by High Performance Anion Exchange Chromatography (HPAEC) system coupled with a Dionex[®] CarboPac[®] PA-100 column (250 x 4 mm) with pulsed amperometric detection (PAD) (Englyst *et al.*, 1994).

3.6.3 Protein content analysis

Kjeldahl analysis was used for the determination of the total protein content in the wheat, Solubles and DWG. Sulphuric acid, hydrochloric acid, methyl red (pH indicator), sodium hydroxide pellets, boric acid and Kjeltabs[®] (digestion catalyser) were obtained from Fisher Scientific Ltd., UK. Turbotherm Rapid Digestion Systems, Gerhardt[®] (Model TT125) was used for sample digestion, Gerhardt Kjeldahl VAPODEST[®], (Vapodest-10) was used for the steam distillation of released ammonia gas, and Schott Titronic[®] basic Titration Burette (auto-titrator) was used for the titration of ammonium salt.

The Kjeldahl total nitrogen determination method consisted of three stages: digestion, distillation and titration. During the digestion stage, 0.50 g of the tested material was placed into digestion tubes along with 2 Kjeltab[®] tablets (anhydrous sodium sulphate and a catalyst), 20 mL concentrated sulphuric acid and a few anti-bumping agents (for prevention of bubbling). The digestion process converts the nitrogen into ammonia salt ((NH₄)₂SO₄), and other organic matter to CO₂ and H₂O.

A control sample was also prepared and digested accordingly. Samples were digested for 1 hour and 20 minutes. At the end of the digestion stage a greenish-blue colour was developed, indicating the completion of the digestion process (increasing the digestion time was needed in order to obtain the green colour). Once digestion process was completed, digestion tubes were left to equilibrate to room temperature.

50

The Gerhardt Kjeldahl Vapodest[®] distillation unit was used to convert the ammonium sulphate into ammonia gas by the addition of sodium hydroxide. The formed ammonia was distilled to a receiving flask containing 25 mL of 3% boric acid and two drops of methyl red (pH indicator). Boric acid was used to convert the ammonia gas into ammonium borate salt.

Determination of the nitrogen content was achieved through titration of the formed ammonium borate against 0.1 M hydrochloric acid; average titration volumes from three replicates were obtained, and the blank titre value was used for calculating the nitrogen percentage presented in the samples:

$\%N = \frac{[Sample \ titration \ volume \ (mL) - Blank \ titration \ volume \ (mL)] \times Molarity \ of \ HCl \times 14 \times 100}{Weight \ of \ sample \ (mg)}$

After the determination of the total nitrogen content, the resultant nitrogen was converted to a protein percentage context by the protein conversion factor multiplication (F = 5.7) for wheat-related products (AOAC, 979.09).

3.6.4 Ash content determination

The term "ash" refers to the inorganic matter in the studied material. The origin of the ash content could be from the plant itself, such as salts absorbed from the soil during growth, or from the supply chain, for example, soil particles carried on the plant, or from the processing of the material, *e.g.* addition of acids and bases may increase the ash content. Ash content is usually determined by combusting the studied subject in a laboratory furnace under controlled conditions.

In this work, the ash content was determined according to the standard biomass analytical method published by the National Renewable Energy Laboratory (NREL), which recommends the expression of the ash as the percentage of residues remaining after dry oxidation at 550° to 600°C (Sluiter *et al.*, 2008).

Samples of the wheat, Solubles and DWG were put into porcelain crucibles and placed in a temperature-controlled furnace at $575^{\circ}C \pm 25^{\circ}C$ for four hours. Crucibles were removed from the furnace directly into a desiccator and left to cool to room temperature and weighed. Crucibles were put in the furnace again and heated for one hour, transferred to the desiccator and weighed again. This was repeated until a constant weight was recorded.

3.7 Results

About 60 kg of wheat (10 batches, details are listed in appendix 1) were processed according to the procedure detailed above. On average, each batch of 6 kg yielded around 2.2 L of ethanol, 1.275 kg on dry base (db) or 5.100 kg of 75% moisture content DWG, and 0.800 kg (db) of Solubles. Thus the total yield of DDG plus S was around 2 kg or one-third of the starting wheat, in line with industrial yields.

3.7.1 Ethanol purity

The concentration of the distilled ethanol was measured by an alcohol hydrometer. Figure 3-8 shows the resulting concentration, which was typically around 89% v/v.



Figure 3-8. Purity of the ethanol produced in the GUNT CE-640 bioethanol plant.

3.7.2 Constituent sugar analysis results

Table 3-2 presents the results from Englyst for the distribution of the constituent sugars and starch content in the wheat, Solubles and DWG. The amount of AX in the wheat was around 5.3%, which becomes concentrated following processing to about 16.7% in the DWG and 11.4% in the Solubles. Total AX was calculated as $[0.88 \times (ara + xyl)]$. (The factor of 0.88 accounts for the addition of a water molecule (molecular weight 18) during the hydrolysis of each monomer of molecular weight 150: (150-18)/150 = 0.88.) The starch content was about 68% in the raw material (4.08 kg starch in 6 kg of wheat), the unfermented starch by the end of the process is about 7% of DWG (around 89 g starch in 1.275 kg of DWG). Therefore, about 98.5% of the total starch was successfully fermented and converted into ethanol and carbon dioxide, while the remaining material was concentrated by a factor of around three.

Table 3-2. Constituent sugars analysis of the original wheat and the resulting DWG and Solubles following ethanol production (analyses performed by Englyst Carbohydrate Ltd.).

Sample			Constit	Total NSP	Starch					
	rha	fuc ara xyl			man	gal	glu	gal-A	g/100g	g/100 g
Wheat	0.0	0.0	2.1	3.9	0.1	0.3	2.4	0.1	$\textbf{8.9}\pm\textbf{0.0}$	68.4 ± 0.7
DWG	0.1	0.0	5.4	11.3	1.2	0.9	10.2	0.4	$\textbf{29.6} \pm \textbf{0.8}$	$\textbf{7.0} \pm \textbf{0.3}$
Solubles	0.1	0.0	3.5	7.9	1.0	1.1	1.4	0.1	15.0 ± 0.4	0.0

3.7.3 Protein and ash content results

Table 3-3 presents the titres obtained for each sample, the calculated average titre, the % total nitrogen, and the corresponding percentage protein content. The protein contents of the two end-products are similar at around 37-39%, similar to literature values (Jarret *et al.*, 2011; Liu, 2011), and both around threefold greater than the 12.77% initial protein content of the wheat. Table 3-3 also reports the ash contents of the wheat, DWG and Solubles. It is notable that the ash content of the Solubles was much higher than of the DWG. Given their relative proportions (about 50% more DWG than Solubles was produced, on a dry basis), overall the ash content if the two were combined would be around 4.72%, again nearly three times the initial ash content of

the wheat. The ash contents of the DWG and Solubles are lower than reported in literature (Jarret *et al.,* 2011; Liu, 2011), possibly because the purity of the chemicals used.

Table 3-3. Protein and ash analysis of the original wheat and the resulting DWG and Solublesfollowing ethanol production (db).

Sample	Titre 1	Titre 2	Titre 3	Average Titre	Average Blank	N %	Protein %	Ash %
Wheat	9.4	9.5	9.5	9.50	1.5	2.24	12.77 ± 0.09	1.8 ± 0.08
DWG	25.9	26.1	26.1	26.05	1.5	6.87	$\textbf{39.18} \pm \textbf{0.18}$	2.2 ± 0.00
Solubles	25.0	25.0	24.9	24.95	1.5	6.57	$\textbf{37.42} \pm \textbf{0.09}$	8.5 ± 0.40

3.8 Discussion

Ten batches, each of 6 kg of wheat, were subjected to liquefaction, saccharification and fermentation in the GUNT CE-640 bioethanol plant, each yielding about 2.2 L of ethanol at 89% v/v purity. The remaining mash was put through a solid/liquid separation process, which resulted in two end products, the Solubles (800 g db) and the DWG (1275 g db), corresponding to a combined yield of around 33%, in line with the typical yield of DDGS following ethanol production from wheat (Jensen and Björnsson, 2012).

Figure 3-9 shows the mass balance of the protein, AX and ash in the starting wheat and the two end products. In 6000 g of wheat at 12.77% protein, there was initially 766 g protein. The resulting DWG and Solubles had between them 798 g, a difference of 4% which reflects accumulated errors in the analyses. The protein was concentrated by a factor of three during the process, consistent with the literature and industrial performance.



Figure 3-9. Mass balance of protein, AX and ash content before and after fermentation process.

In like manner, the 108 g of ash initially increased to 124 g ash in the two end products, an increase of 15% as a result of the enzyme solutions and the chemicals (acid and base) added during the process to adjust the pH. It is notable that the 15% increase is less than the typical increase reported in the literature (20-30%) (Jarret *et al.*, 2011). This difference is probably due to the quality of the chemicals and enzymes used during the process, the most important being the mineral content of the water. The Solubles had a much greater ash content than the wheat or DWG, indicating that the additional mineral material added during the process mostly ended up in the Solubles fraction, as expected if it is mostly coming from water and aqueous solutions.

The original wheat arabinoxylan (356 g) was divided between the DWG and the Solubles; the majority ended up in the DWG, but in total around 12% of the original AX appeared to be lost during the process. This loss could be due either to degradation during the hydrolysis of starch or during distillation. The concentration of AX in the material increased by a factor of about 2.5, from ~6% in the wheat to ~15% in the DWG and Solubles, corresponding with literature descriptions of commercial DDGS (Kosik *et al.,* 2017).

Considering the arabinose to xylose ratio (A/X), which is one of the most critical characteristics of AX besides the molecular weight, the constituent sugars analysis revealed an A/X ratio of 0.54 in the original wheat, while the ratio was lower in the DWG

(0.48) and the Solubles (0.44). Similar decreases are reported in the literature, these differences may relate to differences in their molecular structure and in their cross-linking with other components in the cell wall (Kosik *et al.*, 2017).

It can be concluded that the DWG obtained by the fermentation procedure followed in the current work has a composition similar to the intermediate by-products produced in industrial plants. Therefore, it is a suitable material for subsequent studies on enzyme treatment, AX and AXOS extraction and performance in animal feeding trials.

3.9 Summary

Distillers Dried Grains with Solubles is the second most important product from cereal biorefineries, after ethanol. DDGS is marketed for use in animal feed formulations. The protein component of DDGS is valuable for animal feed, but the fibre component, which is dominated by arabinoxylans, is deleterious in animal feed and would be better exploited for other end uses, including for increasing fibre consumption in the human food chain, while also exploiting other functional properties of AX. Meanwhile, understanding the efficacy of xylanases in animal feed formulations, in terms of their production of prebiotically beneficial arabinoxylan oligosaccharides, is a further important area of investigation.

Commercial production of AX-based products within an integrated biorefinery would access the wet materials following fermentation before it is dried, termed in the current work Distillers Wet Grains (DWG). Accessing DWG from commercial facilities in order to undertake AX extraction and other studies is problematic, as commercial bioethanol plants are closed systems in which this in-process material is inaccessible. Hence, the current work aimed to use a pilot-scale bioethanol production facility to produce equivalent and representative DWG material for further studies.

The GUNT CE-640 bioethanol plant was used to process 10×6 kg batches of wheat, yielding on average 2.2 L of ethanol at 89% v/v, Solubles (0.8 kg db) and DWG (1.275 kg db). The DWG produced in this work appears to have similar yields and compositions

to the material produced in large-scale biorefineries. This material was therefore used for subsequent studies on AX extraction and the production of arabinoxylan oligosaccharides, as described in the following chapters.

4 Arabinoxylan oligosaccharides production

4.1 Introduction

The identification of prebiotics, which stimulate the proliferation of healthy bacteria in human and animal digestive systems, is one of the most interesting developments in the food ingredient sector and of increasing interest in the animal feed sector as well. Arabino- and xylo-oligosaccharides exhibit promising prebiotic activity that promotes their use either by direct addition or through *in-situ* production by xylanases. This chapter reviews the importance of XOS prebiotics and presents experimental work to study the ability of xylanases to produce XOS from the freshly produced DWG, utilising different enzymes and pre-treatments.

4.2 The concept of functional food

In the 1980s, the Japanese food market witnessed the first usage of the term "functional food" to describe food commodities that contained special fortifying ingredients that offer healthy physiological advantages to the consumer (Aachary and Prapulla, 2011). Functional foods were a legislative food category called "FOFHU" and were required to fulfil three nutritional specifications: 1) to show effective results in clinical trials; 2) to be safe for clinical and non-clinical examinations; 3) to have quantifiable active components (Martirosyan and Singh, 2015). In Europe, the functional food concept was defined by the European Commission's Concerted Action on Functional Food Science in Europe (FuFoSE) as (Diplock and Fern, 1999):

"A food product can only be considered functional if together with the basic nutritional impact it has beneficial effects on one or more functions of the human organism thus either improving the general and physical conditions or/and decreasing the risk of the evolution of diseases. The amount of intake and form of the functional food should be as it is normally expected for dietary purposes. Therefore, it could not be in the form of pill or capsule just as normal food form." However, European legislation considers functional food as a concept rather than a food category (Stanton *et al.,* 2005). Currently, the European food categories are: conventional foods, modified foods, foods for special dietary use and medical foods (Martirosyan and Singh, 2015). But food developers are allowed to make two claims: nutrition claims and health claims. Nutrition claims indicate the basic nutritional value and the energy provided, whereas health claims cite the food's ability to prevent, manage or heal illness.

The Functional Food Centre (<u>http://functionalfoodscenter.net/</u>) in the United States defines functional foods as:

"Natural or processed foods that contains known or unknown biologically-active compounds, which in defined amounts provide a clinically proven and documented health benefit for the prevention, management, or treatment of chronic disease."

The definition underlines the necessity of specific "bioactive compounds" in functional foods that have health effects, such as probiotics and prebiotics.

The word "probiotic" is the exact opposite of "antibiotic" in the Greek language, and it means "for life". The term probiotic was first used to describe substances produced by one microorganism that stimulate the growth of another (Lilly and Stillwell, 1965). Parker (1974) defined the probiotics as:

"Organisms and substances which contribute to intestinal microbial balance."

This definition connected probiotics to intestinal health, however, it does not clarify the nature of probiotics as it included both "organisms and substances". Fuller (1989) refined the definition to refer to the microorganism:

"A live microbial feed supplement which beneficially affects the host animal by improving its intestinal microbial balance." The recent definition declares the nature of probiotics to be viable live organisms. Lactic acid bacteria and bifidobacteria, as natural components of the intestinal microbiota, are the most studied and used probiotics in dairy products (Siró *et al.,* 2008).

Prebiotics, by contrast, are nutrients that possess the ability to modify the intestinal microbiota. Early reports described the so-called "Bifidus factor" as a component of human milk which is beneficial for the bifidobacteria in infants (Hutkins *et al.,* 2016). Thereafter, the Bifidus factor was found to be a complex of oligosaccharides and glycans. The term and definition of prebiotics were first introduced by Gibson and Roberfroid (1995) as a functional food ingredient:

"A non-digestible food ingredient that beneficially affects the host by selectively stimulating the growth and/or activity of one or a limited number of bacteria in the colon, and thus improves host health."

From the latter broad definition, it is concluded that prebiotics include the oligo- and polysaccharides that resist acid and enzymatic hydrolysis in the human intestinal tract, can be fermented by microflora and stimulate the growth of gut bacteria. The most proposed and available prebiotics in the market are oligosaccharides such as lactulose, fructo-oligosaccharides, galacto-oligosaccharides, soybean oligosaccharides, lacto-sucrose, isomalto-oligosaccharides, gluco-oligosaccharides, xylan oligosaccharides (XOS) or palatinose (Macfarlane *et al.*, 2006; Manning and Gibson, 2004; Okazaki *et al.*, 1990).

4.3 AXOS and XOS prebiotic properties

The prefix "oligo-" comes from the Greek $\delta\lambda i\gamma o\iota$ which means "a few or containing a small number of units" (Oxford Dictionary). Oligosaccharides are small chains of a limited number of sugars connected to form one molecule; the IUB-IUPAC nomenclature defines oligosaccharides as saccharides containing three to ten monomers (Voragen, 1998), while some other authorities consider saccharides containing between three and nineteen units as oligosaccharides. However, from a nutritional point of view, most disaccharides have similar properties to tri- and tetrasaccharides and are also considered oligosaccharides (Crittenden and Playne, 1996).

Xylan oligosaccharides are a novel class of prebiotics that occur naturally in a variety of fruits, vegetable bamboo, honey and milk (Poesen *et al.*, 2016; Rastall and Gibson, 2015; Singh *et al.*, 2015) and that could be produced from lignocellulosic biomass. The structure of XOS is similar to the parent AX: a linear β -(1-4)-D-xylopyranose with or without substitutions of arabinose, uronic acid derivatives, ferulic and coumaric acid and other phenolic acids (Gullón *et al.*, 2014). The variation in the XOS structure depends mainly on the parent plant type, the part of the plant and the method of production (Suzuki *et al.*, 2000).

In general, the prebiotic effectiveness of oligosaccharides is subject to their fermentability by the gut bacteria, which is related to their structure and degree of polymerisation (DP). Van Craeyveld et al. (2008) reported that AX oligosaccharides with average DP of 3 (and a maximum no greater than DP5) showed greater ability to elevate the concentration of bifidobacteria when compared with the larger DP. Furthermore, DP2 and DP3 XOS exhibited faster utilisation kinetics than DP4 and DP5 XOS (Gullón et al., 2008a). It is believed that the xylan polysaccharides are not susceptible to the gut bacteria and possess no prebiotic properties at all (Falck et al., 2013; Neyrinck et al., 2012). Predominantly, XOS of a degree of polymerisation of DP2-DP4 are desired for food applications (Vázquez et al., 2000). The influence of the XOS substitutions on the fermentation kinetics showed that both the linear XOS and the branched AXOS are more susceptible to the intestinal microbiota than the more complex XOS containing glucuronic acid, while GluA-XOS fermentation yielded higher concentrations of shortchain fatty acids (Kabel et al., 2002). Similarly, XOS extracted from wheat bran showed better prebiotic activity than the XOS extracted from Bengal Gram husk; this could be attributed to the higher arabinose substitutions in the wheat bran (A)XOS (Madhukumar and Muralikrishna, 2012). Therefore, it can be concluded that the prebiotic activity of XOS and AXOS is subject to the degree of polymerisation, substituent types and the arabinose-to-xylose ratio (degree of substitution).

Arabinoxylan oligosaccharides are a white powder, pH stable over the range 2.5 - 8.0 and thermally stable up to 100° C (Courtin *et al.*, 2009). Wang *et al.* (2009) reported that XOS could tolerate pasteurisation and high temperature sterilisation in a low pH environment. The pH and thermal stability favour XOS over inulin for a variety of food applications, such as in juice, as inulin is more susceptible to decomposition at low pH and high temperatures (Courtin *et al.*, 2009).

As prebiotics, XOS specifically invigorate the growth or the activity of a group of bacteria (Immerzeel et al., 2014; Mendis and Simsek, 2014) and thereby improve health, reduce gut infections and suppress colon cancer (Macfarlane et al., 2006). They are preferentially used by *Bifidobacterium* spp (Okazaki et al., 1990) and utilised as a carbon source by Weissella strains (Patel et al., 2013) and Bifidobacterium spp (Fujikawa et al., In vitro tests showed that XOS are the preferred substrate for the 1991). Bifidobacterium spp over the other non-digestible oligo-saccharides of hexoses; their activity was comparable to raffinose oligosaccharides and higher than fructooligosaccharides (Jaskari et al., 1998; Vázquez et al., 2000). XOS seemed to improve the overall gut health and regulate the bowel movement in cases of severe constipation (Chung et al., 2007; Tateyama et al., 2005) with no reports of any side effects such as increased flatulence, intestinal bloating or intestinal irritation, which have been reported to accompany the consumption of fructo-oligosaccharides (Bruggencate et al., 2006).

Prebiotics can mediate the protective immune response in the intestinal tract, caused by foreign antigens from food or microbes, and help to avoid hypersensitivity reaction against antigens (Schley and Field, 2002). The mediation is mainly a result of the production of short chain fatty acids (SCFA) (acetate, propionate, butyrate and lactate). XOS demonstrated an anti-inflammatory, immunomodulatory and free radical scavenging activity (Chen *et al.*, 2012; Hromádková *et al.*, 2013). Also, the XOS fermentation products, SCFA, have an important role in adjusting the pH in the intestine to prevent the growth of unwanted pathogenic bacteria (Hughes *et al.*, 2007; Knudsen and Lærke, 2010; Wang *et al.*, 2010; Boll *et al.*, 2015; Rumpagaporn *et al.*, 2015). XOS have been reported to deliver several health-related advantages, such as improving the clinical conditions associated with diabetes and reducing mortality rates in rodents (Gobinath *et al.*, 2010). Also, XOS demonstrated an effective role in the regulation of obesity and associated metabolic disorders in mice (Neyrinck et al., 2012). XOS were reported effectual against the induced oxidative stress associated with a high-fat diet in rats (Wang *et al.*, 2011); similarly, they played a protective role against high-protein diet-induced genotoxicity (colonic DNA damage). However, such results were not obtained for inulin (Christophersen *et al.*, 2013). This protective activity was ascribed to the production of SCFA as they seem to activate the genes responsible for cellular energy metabolism and anti-inflammatory response (Fukuda *et al.*, 2011). Moreover, XOS confer advantageous effects in regulating glucose and lipid profile and thereby on human energetic metabolism (Staneva *et al.*, 2014).

In parallel, significant efforts have been paid to increase the amount of AXOS in animal feed. Several pieces of research suggest that AXOS influence the growth performance, mortality rates and general gut health of broilers (Austin *et al.*, 1999; Courtin *et al.*, 2008; Lei *et al.*, 2016). The production of AXOS in animal feed is achieved by the addition of xylanases to the feed (Lei *et al.*, 2016).

4.4 Xylanases and AXOS in animal feed

Xylanases are naturally occurring enzymes that bio-catalyse the degradation of xylan chains (Bonner, 2007). They have been exploited in a variety of industrial applications including bleaching in the paper industry, breadmaking, fruit juice and beer clarification, and animal feed (Sharma, 2013), with a global annual market share of 1 billion USD with approx. 40% of the value being for the non-starch polysaccharides hydrolysing enzymes (NSPases) in 2018 (https://www.marketsandmarkets.com/Market-Reports/feed-enzyme-market-1157.html).

Xylanases are one of the most consumed enzymes in the pig and poultry feed formulation industry, used traditionally by feed manufacturers to achieve performance consistency (Bedford and Morgan, 1996; Paloheimo *et al.*, 2010; Sitanaka *et al.*, 2018).

Despite the wide usage of xylanases in animal feed, the function of these enzymes is not completely understood. It is established that AX behave as an antinutrient in the birds' feed, which was explained by two mechanisms. The first mechanism suggests that AX are resistant to enzymatic degradation in the small intestine and thereby hinder endogenous enzyme (*e.g.* protease and amylase) gaining access to the intact endosperm cells due to the lack of functional NSPase enzyme suite in the animal (Bedford and Morgan, 1996; Pettersson and Åman, 1989). Selle *et al.* (2009) reported that utilisation of xylanases in the feed formulation improved overall feed intake and digestibility and thereby growth performance.

The second mechanism is based on the ability of AX to increase the viscosity of intestinal fluids, which is known to reduce the diffusion rate of nutrients, decrease feed passage rates and increase the population of gut flora (Feighner and Dashkevicz, 1988; Fengler and Marquardt, 1988; Salih *et al.*, 1991). Adding xylanases to the feed significantly reduces the viscosity in the intestines and improves the weight gain (Wu *et al.*, 2004; Zyłe *et al.*, 1999). The two mechanisms are connected and indicate the importance of degrading the large molecules of AX in animal feed to reach an optimum feed utilisation.

More recently, attention has turned to the beneficial role of the (A)XOS produced by xylanases (Broekaert *et al.*, 2011; Craeyveld *et al.*, 2008). This role was linked to the fermentation of AXOS in the animal's digestive system and the production of SCFA (Choct *et al.*, 1999). It was found that SCFA promotes the release of the neuropeptide peptide YY (PYY) which increases the efficiency of digestion and nutrient absorption after a meal by delaying gastric emptying (le Roux and Bloom, 2005; May *et al.*, 2015). Alijosius *et al.* (2018) reported that the xylanases affected the tenderness and colour of chicken breast meat. This effect could be attributed to the produced XOS prebiotic or the phenolic compound in the feed, although more studies are needed to reveal the impact of XOS and other prebiotics on animal feed. These studies re-emphasise the necessity to understand the susceptibility of feed substrates to release XOS when treated with enzymes. Also, it underlines the importance of producing considerable amounts of XOS and AXOS to be incorporated in animal feed trials in order to gain greater knowledge about the effects of including these prebiotics in feed formulations.

DDGS as a bioethanol by-product is marketed as a nutritional ingredient for animal feed. It is evident that the severe drying conditions reduce the nutritional value of DDGS for animal feed and change the characteristics of AX in the process (De Vries *et al.*, 2013; Kosik *et al.*, 2017; Pedersen *et al.*, 2014; Schroeder, 2012). Therefore, there is a great opportunity to increase the value of DDGS via xylanase treatment prior to drying. The wet DWG produced as described in the previous chapter offers a suitable substrate to examine the enzymatic production of (A)XOS or (A)XOS-rich materials within biorefineries.

This chapter presents experiments performed to study the ability of xylanases to produce (A)XOS from freshly produced DWG, utilising different enzymes and pretreatment methods, in order to evaluate the possibility of valorising this biorefinery by-product by increasing the nutritional value of the DDGS.

4.5 Materials and methods for DWG-xylanase treatment

Aiming to produce meaningful levels of oligosaccharides from the freshly produced DWG for animal feeding, three enzymes were tested at two dosages under two pH conditions. Pretreatment with protease enzyme was also investigated.

4.5.1 Enzymes and chemicals

Xylanases: Econase XT 25L, Xylanase 1 EL-2016/002288 and Xylanase 2 EL-2016/002287 were kindly provided by AB Agri Ltd. The first of these is a commercial endoxylanase marketed to the animal feed industry; the other two are development enzymes. The dose of Econase when added to animal feed is usually 100 mL per tonne, preliminary results did not show any XOS production at this dosage level, hence a 500× higher dose of 50 mL per kg was used throughout this project. Citric acid monohydrate, Sodium dihydrogen phosphate dodecahydrate, sodium hydroxide and trifluoroacetic acid were obtained from Fisher Scientific Ltd, UK. Samples were treated with enzymes within a shaking water bath Thermo-Haake[®], model (SWB20), and centrifuged using Eppendrof[®] centrifuge, model 5702.

4.5.2 Enzyme treatment

A sample of 2 g (db) of DWG was transferred into a 200 mL beaker, citrate-phosphate buffer added to give a solid-to-liquid ratio around 1:10, and the beakers moved into a shaking water bath of 42°C. Two buffer solutions were prepared at pH 5.5 (the enzyme optimum pH range) and a lower pH 2.5 in order to investigate the enzymes' performance in acidic conditions during digestion in the animal's gastric system. The studied enzyme was added according to the dosages listed in Table 4-1. The enzymes were added at 50 mL per kg dose and at 10x that dose; two blank samples were also prepared, to observe the pH effect. Samples were incubated for 1 hour, after which they were removed from the water bath and allowed to cool to room temperature. The pH was adjusted to 7±0.1 with 1M sodium hydroxide. The resultant slurry was centrifuged at 4000 rpm for 15 minutes. Supernatants were collected, filtered through a syringe filter (0.45 μ m pore) and freeze-dried at 0.04 mbar and -55°C for 72 hours.

9	Sample ID	Enzyme	Enzyme Dose	рН
sample 1	Econase ×1- 2.5	Econase XT 25L	0.1 mL	2.5
sample 2	Econase ×1- 5.5	Econase XT 25L	0.1 mL	5.5
sample 3	Econase ×10- 2.5	Econase XT 25L	1 mL	2.5
sample 4	Econase ×10- 5.5	Econase XT 25L	1 mL	5.5
sample 5	XYL1 ×1- 2.5	Xylanase 1 EL-2016/002288	0.1 mL	2.5
sample 6	XYL1 ×1- 5.5	Xylanase 1 EL-2016/002288	0.1 mL	5.5
sample 7	XYL1 ×10- 2.5	Xylanase 1 EL-2016/002288	1 mL	2.5
sample 8	XYL1 ×10- 5.5	Xylanase 1 EL-2016/002288	1 mL	5.5
sample 9	XYL2 ×1- 2.5	Xylanase 2 EL-2016/002287	0.03 mL*	2.5
sample 10	XYL2 ×1- 5.5	Xylanase 2 EL-2016/002287	0.03 mL	5.5
sample 11	XYL2 ×10- 2.5	Xylanase 2 EL-2016/002287	0.3 mL	2.5
sample 12	XYL2 ×10- 5.5	Xylanase 2 EL-2016/002287	0.3 mL	5.5
sample 13	Buffer 2.5	-	x0	2.5
sample 14	Buffer 5.5	-	x0	5.5

Table 4-1. Dosage and pH of the enzyme incubation test.

*mL of 1 mg/mL solution

4.5.3 Protease pre-treatment

It was hypothesised that wheat protein denaturation during the fermentation and distillation process may create a protective protein matrix, preventing the xylanases

from accessing the structure of the bran particles and hence reducing the activity of the enzyme and decreasing the production of AXOS. Therefore, the DWG was treated with protease to degrade the proteins, then followed by Econase treatment as described in Table 4-2. The supernatant was collected, freeze-dried and sent for analysis at IPOS labs for sugar analysis (see below).

Sample ID	1 (DWG + Econase)	2 (DWG + Protease + Econase)							
DWG (g) db	2	2							
Buffer (pH = 8) (mL)	20	20							
Protease (mL)	-	0.1							
Incubate for 1 hour at 50 °C									
	Boil the mix for 10 min to deactivate	e the protease							
	Adjust the pH to 2.5 by citric acid								
Econase (mL) 0.1 0.1									
Incubate for 1 hour at 42 °C									
	Cool, neutralise with NaOH 1M an	d centrifuge							

Table 4-2.	Protease	treatment	prior to	Econase.
	11010430	cicacificite		Leonase.

4.5.4 Analytical Methods

The analytical methods applied to the dried samples were mainly the analysis of constituent sugars after hydrolysis and screening the profile of oligosaccharides.

4.5.4.1 Constituent sugars analysis

The AX content of the released material collected in the supernatant was quantified by hydrolysing the material to its constituent arabinose and xylose, which were then measured using high performance anion exchange chromatography with a pulsed amperometric detector (HPAEC-PAD). The HPAEC Instrument used was a Dionex ICS–3000 Ion Chromatography System (Dionex Corporation, CA, USA), which comprises a Dionex Auto Sampler, Dionex ICS 3000 Eluent Organiser, Dionex ICS 3000 Detector Chromatography, Dionex ICS 3000 Dual Pump System and Dionex ICS 3000 Eluent Generator Reagent-Free Ion Chromatography. The column used was a CarboPac[®] PA20 analytical column, 3×150 mm 6.0 µm particle size. The chosen mobile phase was 10mM

NaOH with an adapted flow rate of 0.3 mL min⁻¹ and an injection volume of 20 μ L. Chromeleon[®] Xpress software was utilised for data processing (Englyst *et al.,* 1994).

A hydrolysis step was performed to induce chemical bond cleavage through water addition, which facilitates the conversion of AX and AXOS into their sugar moieties subcomponents. 4.0 mg of the dried supernatant was transferred carefully into a pressure tube, 2 mL of 2M trifluoroacetic acid was added, the tubes were capped and heated at 120°C for 2 hours, then left to equilibrate to the ambient room temperature. A constant stream of nitrogen gas at 60°C was maintained in order to evaporate the trifluoroacetic acid. Subsequently, the dried residues were dissolved in 4 mL of ultra-purified water for HPAEC-PAD analysis.

4.5.4.2 Oligosaccharide profile analysis

Monosaccharide analysis of hydrolysed samples is relatively straightforward and readily achieved by HPAEC. Oligosaccharide analysis, by contrast, is more challenging. At this stage of the project, we had not developed the new method that is described later in Chapter 6. We therefore collaborated with Innovative Physical Organic Solutions (IPOS), Huddersfield, who were developing a method for screening of oligosaccharide distribution utilising a newly developed size exclusion chromatography method (Powles, N., unpublished work). Samples were sent to IPOS and the analysis performed using an HPLC coupled to a refractive index detector (RID). The instrument was run isocratically using ultra-pure water, at a flow rate of 0.2 mL min⁻¹.

4.6 Enzyme treatment results

The monosaccharide and oligosaccharide analyses were performed in parallel on the treated samples, aiming to construct a full qualitative and quantitative picture of the hydrolysate distribution and thereby develop a detailed understanding of the xylanases' activities at different pH values and dosages.

4.6.1 Monosaccharide analysis results

Both samples and standards were analysed via HPAEC-PAD as described in section 4.5.4.1, with calibration curves constructed for the standards of the three sugars of interest, arabinose, xylose and glucose. Figure 4-1 illustrates the standards mixture chromatogram.



Figure 4-1. Arabinose, xylose and glucose standards separation chromatogram.

Retention times were recorded to provide a valid comparative background between the analysed samples. To construct the calibration curves, area under the curve (AUC) values were integrated via the chromatographic software Chromeleon[®] and plotted against concentrations for each standard.

After hydrolysing all the samples, the solutions were introduced to the HPEAC-PAD system. Comparisons were implemented between the resultant peaks and the standards for identification and quantification purposes. The amounts of the released monosaccharides in the samples were calculated and are shown in Table 4-3 and Figure 4-2. Clearly, the amounts of monosaccharides in the Econase-treated samples are higher than the samples treated with the other xylanases. In the case of Econase, the low pH proved to be a better environment, releasing more AX material into the supernatant, but this was not apparent for the other two enzymes. Increasing the dose by a factor of 10 appeared to reduce the release of AX.

		-		
Sample ID		Arabinose mg	Glucose mg	Xylose mg
Sample 1	Econase ×1- 2.5	4.79 ± 0.24	5.95 ± 0.30	8.09 ± 0.41
sample 2	Econase ×1- 5.5	3.73 ± 0.19	5.34 ± 0.27	6.18 ± 0.31
Sample 3	Econase ×10- 2.5	2.96 ± 0.15	5.46 ± 0.27	5.26 ± 0.26
Sample 4	Econase ×10- 5.5	2.33 ± 0.12	6.09 ± 0.31	4.39 ± 0.22
Sample 5	XYL1 ×1- 2.5	2.84 ± 0.14	2.84 ± 0.14	4.08 ± 0.21
Sample 6	XYL1 ×1- 5.5	2.76 ± 0.14	2.89 ± 0.15	2.98 ± 0.15
Sample 7	XYL1 ×10- 2.5	2.21 ± 0.11	2.66 ± 0.13	3.35 ± 0.17
Sample 8	XYL1 ×10- 5.5	1.98 ± 0.16	2.74 ± 0.22	3.00 ± 0.24
Sample 9	XYL2 ×1- 2.5	1.68 ± 0.14	1.78 ± 0.14	1.42 ± 0.12
Sample 10	XYL2 ×1- 5.5	2.19 ± 0.18	2.01 ± 0.16	2.53 ± 0.20
Sample 11	XYL2 ×10- 2.5	2.08 ± 0.17	1.90 ± 0.15	2.38 ± 0.19
Sample 12	XYL2 ×10- 5.5	2.21 ± 0.18	2.20 ± 0.18	2.82 ± 0.23
Sample 13	Buffer 2.5	1.56 ± 0.13	1.73 ± 0.14	1.35 ± 0.11
Sample 14	Buffer 5.5	1.57 ± 0.13	1.74 ± 0.14	1.47 ± 0.12

 Table 4-3. Total released monosaccharides from DWG after xylanase treatment (Measured by



HPAEC-PAD after hydrolysis)

Figure 4-2. Total released monosaccharides from DWG after xylanase treatment (measured by HPAEC-PAD after hydrolysis).

The amount of AXOS released from each sample was calculated as $[(Ara + XyI) \times 0.88]$. Table 4-4 and Figure 4-3 report the AXOS amounts in the samples following treatment with the three enzymes at two dosages and two pH levels.

Table 4-4. Amount of AXOS released from 2 g of DWG by enzymes at different dosages and pH.

						•								
Sample	1	2	3	4	5	6	7	8	9	10	11	12	13	14
AXOS (mg)	11.3	8.7	7.3	5.91	6.1	5.1	4.9	4.4	2.7	4.2	3.9	4.4	2.6	2.7



Figure 4-3. Amount of AXOS released from 2 g of DWG by enzymes at different dosages and pH.

The results suggest that lower pH can increase the effectiveness of Econase XT 25L; incubating the DWG at pH 2.5 instead of 5.5 gave 25% more AXOS. Moreover, Econase XT 25L had the strongest efficacy among the tested enzymes. For this reason, and because it is currently a commercial product, this enzyme was used for subsequent studies.

The sample treated with the lower dose of Econase XT 25L seemed to release more AXOS than the sample treated with 10 times the dosage. It is conceivable that the enzymes have limited ability to penetrate the cell wall matrix, therefore only a limited amount of AX might be released, after which the hydrolysing ability of the enzymes is directed towards the already released AX molecules, such that the released AXOS were further hydrolysed into smaller moieties. Considering that the followed analysis procedure was designed to accommodate the analysis of polysaccharides by hydrolysing
them into the monosaccharides, it is possible that the hydrolysing process degraded a considerable amount of the oligosaccharides produced by the high enzyme dosage. It is well known that monosaccharide analysis is sensitive to the hydrolysis conditions used and that the chosen conditions need to strike a balance between adequate hydrolysis and minimal degradation of the hydrolysis products, with appropriate conditions needing to be established for different analytes (Ranganathan *et al.*, 1985). The current findings probably indicate the unsuitability of the applied hydrolysis procedure for the measurement of the oligosaccharides in these samples. However, despite uncertainly over the accuracy of the analysis, it remains clear that the amount of AX/AXOS released by enzymes was very small, no more than 0.57% of the treated raw material.

4.6.2 Oligosaccharide profile analysis results

The applied chromatographic method developed by IPOS, the so-called the DPn method, offered the opportunity to examine the oligosaccharides profile by separating the oligosaccharides according to their degree of polymerisation. The technique is limited to polysaccharides with a degree of polymerisation range 1-6. Figure 4-4 shows the separation of XOS standards of DP2-6.



Figure 4-4. XOS standards separation by IPOS.

The analysis results of the enzymatic treatment are summarised in Table 4-5 and Figure 4-5.

Sample ID	DP6	DP5	DP4	DP3	DP2	DP1	
Sample 1	ple 1 Econase ×1- 2.5		0.032	0.003	0.132	0.203	0.893
sample 2	Econase ×1- 5.5	0.074	0.081	0.049	0.347	0.683	2.683
Sample 3	Econase ×10- 2.5	0.015	0.055	0.027	0.097	0.247	1.176
Sample 4	Econase ×10- 5.5	0.056	0.058	0.006	0	0.519	2.410
Sample 5	XYL1 ×1- 2.5	0.077	0.098	0	0	0.335	1.261
Sample 6	XYL1 ×1- 5.5	0.058	0.087	0	0	0.472	1.554
Sample 7	XYL1 ×10- 2.5	0.009	0.101	0	0	0.135	0.971
Sample 8	XYL1 ×10- 5.5	0.060	0.036	0.038	0	0.151	1.149
Sample 9	XYL2 ×1- 2.5	0.007	0.085	0.114	0	0.495	1.750
Sample 10	XYL2 ×1- 5.5	0.006	0.030	0.027	0	0.142	0.392
Sample 11	XYL2 ×10- 2.5	0	0	0	0	0.278	0.745
Sample 12	XYL2 ×10- 5.5	0.060	0.063	0	0	0.407	0.993
Sample 13	Buffer 2.5	0.096	0.060	0	0	0.461	2.043
Sample 14	Buffer 5.5	0.224	0.049	0	0	0.512	2.421

Table 4-5. Oligosaccharides in the samples(%w/w of DWG) released by enzymes at differentdosages and pH.



Figure 4-5. Oligosaccharides in the samples (%w/w of DWG) by enzymes at different dosages and pH.

Clearly, the amount of released AXOS was much less than 1% in all samples. In fact, the analysis shows that the monosaccharides are the overriding hydrolysate. Also, most of the liberated AXOS are DP2 and DP3 with traces, or absence, of DP4, DP5 and DP6. Contrary to the monosaccharides analysis results, higher pH (5.5) yielded higher amounts of DP2 in the Econase treated samples.

The results show a similar activity of all three enzymes in terms of the released oligosaccharides with degree of polymerisation 1 to 6. Moreover, the results suggest that the enzymes' performance in generating the AXOS is comparable to the buffer effect by itself as seen in Figure 4-5.

The high content of monosaccharides varied between the studied samples, the highest in the Econase-treated samples and the control samples at low pH values. The monosaccharide could be glucose resulting from the presence of unfermented residues of starch that had been hydrolysed during the treatment, or could be xylose resulting from the enzyme activity. The IPOS method does not distinguish the detected monosaccharides, glucose, arabinose and xylose, hindering interpretation of the results beyond the overall observation that the enzymes were not very active on this substrate.

In the case of Econase, contrary to the results from the monomers analysis, the results here suggest that the enzyme has a better performance in higher pH environment and it also shows no difference between high and low doses. This variation of results could be attributed to the fact that the DPn method is limited to measure and detect only the small oligomers (DP1-DP6) where the constituent sugar analysis was able to detect the larger polysaccharides.

Both analytical procedures indicated that only minor amounts of AXOS were released. Hence, it was investigated whether chemical extraction of AX followed by enzymatic hydrolysis could give higher yields of the desired AXOS; this investigation is reported in the next chapter.

4.6.3 Protease treatment results

The dried supernatant following protease pretreatment was also analysed for the oligosaccharides in IPOS labs; the content and the w/w% of the oligos were calculated as shown in Table 4-6 and Figure 4-6. Degradation of the protein increased the total amount of XOS released (DP2-DP6) from 3.54% to 4.0%, an increase that is probably insignificant and certainly not substantial.

	%w/w							
	DP6 DP5 DP4 DP3 DP2							
DDG + XT	0.10	0.69	0.35	0.40	2.00	5.10		
DDG + Protease + XT	0.20	1.00	0.50	0.50	1.80	5.10		

Table 4-6. Protease Treatment results %w/w.



Figure 4-6. Protease treatment effect on the released AXOS.

Both Table 4-6 and Figure 4-6 present no significant outcome of the protease treatment of the substrate especially with the mono and di-saccharides. The graph shows a slightly higher content of DP5, but this could be a systemic error as there was no difference between the two samples at the other DPs.

4.7 Summary

Distillers Wet Grain was subject to a series of investigations to evaluate its potential as an XOS production source. The work focused on the enzymatic hydrolysis of DWG using a variety of xylanases and conditions, as well as pretreatment with protease to reduce any protective effect of denatured protein.

Three xylanases were tested at two dosages; 50 mL per kg and 10× that dose, and at pH values of 2.5 and 5.5. Econase outperformed the other two enzymes and was more effective at the lower pH, but increasing the dosage did not release more AX material, and in any case the amounts released were small.

The analysis method developed in IPOS labs, the DPn method, gave a qualitative picture of the distribution of the oligosaccharides that was helpful at this stage of the study; however, later in the work a better method was developed, as described below in Chapter 6.

The poor overall performance of the studied xylanases argued against further studies of AXOS production from the DWG by the direct enzymatic method. Animal feed trials would require typically 4 kg of DWG with a content of 10% AXOS in order to show measurable effect on weight gain; the results presented here showed that this was not achievable by enzymatic treatments. It might be more efficient to extract AX from DWG by chemical means, hydrolyse this more concentrated AX using Econase, and then deliver this product to the animal feed trials. Therefore, the next chapter describes studies of AX extraction from DWG.

5 Chemical extraction of AX from DWG

5.1 Introduction

The fractionation of biomass residues in order to produce valuable materials has attracted a great deal of attention recently. Globally, the amount of agricultural residues produced annually is estimated at around 5 billion tons (Naidu, Hlangothi, and John, 2018). (These estimations are based on the amounts of harvested crops, due to lack of systematic tracking of under-exploited agricultural residues (Smil, 1999)). DDGS is an increasingly available biorefinery by-product that goes to animal feed. The composition of the DDGS and the parent DWG suggests great potential for the isolation of arabinoxylan. Therefore, the extraction of AX within biorefineries where the starting material, DWG, is abundantly available seems to be a plausible and promising scenario. Isolating AX has been reported in the literature via several methods and pathways. This chapter reviews previous work on extracting AX, discusses the effects of the extraction method on the final product, and demonstrates the extraction procedure followed in the current project to isolate AX soluble fibres to be used as a substrate for the XOS production.

5.2 Arabinoxylans extraction methods

The liberation of arabinoxylan molecules from the cereal cell wall matrix is hindered by the complicated and massive structure tethering cell wall components together (Courtin and Delcour, 2001). AX naturally form ester- and/or ether-linked lignin-polysaccharide complexes, with hydrogen bonds between polysaccharides chains (Ebringerová and Heinze, 2000). Severe treatments are therefore required to separate AX from other cell wall constituents (Courtin and Delcour, 2001). Typically, the extraction process of AX follows a chemical solubilising process or mechanical-assisted hydrolysing method followed by a final purification procedure.

5.2.1 Hydrolysing and solubilising of AX

5.2.1.1 Water extraction

AX are classified into water-extractable and water-unextractable categories. The waterextracted AX was reported to preserve the native structure of the molecule, but in low yields (Ebringerová and Heinze, 2000). The low yield of this method is a result of the robust attachment to lignin through ferulate links and the hydrogen bonds between AX and cellulose (Maes and Delcour, 2002; Vinkx and Delcour, 1996). Non-covalent bonds are weak, however, the abundance of these bonds prevents AX solubilisation under moderate extraction conditions (Slavin, 2006). Applying higher temperatures and pressures during the procedure can significantly increase AX solubilisation into the extraction medium; however, the harsher conditions solubilise other cell wall components as well and consequently reduce the purity of the extracts, which therefore require further purification treatments (Schooneveld-Bergmans *et al.*, 1999). Moreover, as noted above, one of the advantages of water extraction is that it preserves the molecules' original conformation; the application of high temperature and pressure was reported to modify the structure of AX and hydrolyse the polymer into smaller molecules with a lower degree of polymerisation (Josefsson *et al.*, 2002).

5.2.1.2 Alkaline extractions

The presence of the hydroxyl ions in alkaline media has the ability to induce modifications of chemical bonds and to disrupt hydrogen and covalent bonds. The alkaline breaking of hydrogen bonds between AX and cellulose and of ester links loosens the linkage between AX and the other cell wall components and eases the liberation of AX from the matrix (Cyran *et al.*, 2004; Fincher and Stone, 1986). The temperature of the alkali greatly affects the rate of the hydrolysing reaction and the extraction yields. Bergmans *et al.* (1996) reported an increase of the extraction yield of AX from wheat bran from 29% at 20°C up to 50% at 95°C using barium hydroxide. However, the elevated temperature seemed to decrease the extraction yields with other alkalis, due

to the hydrolysis of the linkages between xylose units in AX and the resulting degradation of the AX chains. By contrast, the concentration of the alkali used did not show a great effect on the extraction yields; elevation of the sodium hydroxide concentration from 0.25 to 0.5M was reported to have no effect on yields (Ayala-Soto *et al.*, 2016).

The nature of the alkali used for the extraction affects the extraction yield (Chanliaud *et al.*, 1995). Bergmans *et al.* (1996) demonstrated that substitution of barium hydroxide with calcium or sodium hydroxide slightly reduced the yield of the extraction. Among various alkalis examined, barium hydroxide conferred relatively pure AX extracts with relatively high yields (Bergmans *et al.*, 1996; Gruppen *et al.*, 1991). The mechanism causing this selective action is not well explained, it was assumed that the barium ions prevent the degradation of β -glucans and they selectively interact with pentoses (Fincher and Stone, 1986; Gruppen *et al.*, 1991)

Alkaline extraction is usually associated with lignin removal or degradation by oxidation using dilute concentrations of hydrogen peroxide (Maes and Delcour, 2001). Vinkx and Delcour (1996) reported that the oxidation activity might also help solubilising recalcitrant materials. Furthermore, under the oxidation conditions, AX chains may undergo non-enzymatic oxidative gelation; inducing the oxidative coupling of ferulic acid residues may result in the formation of ferulate dimers cross-linking the AX molecules (Vinkx and Delcour, 1996). Oxidative conditions significantly elevate the yields of alkaline extraction. It was reported that a hydrogen peroxide concentration of 2% at a pH of 11.5 for four hours at 60°C was sufficient to recover 77% of the total arabinose and 65% of the total xylose of the starting wheat bran (Maes and Delcour, 2001). Increasing the hydrogen peroxide concentration showed no effect on the yield.

5.2.1.3 Acid extractions

The conversion of carbohydrates in acidic conditions into unpleasant compounds – from a food chemist's point of view – such as furfural aldehyde products has hindered to some extent the expansion of this route of extraction (Almeida *et al.*, 2007; Pedersen *et*

79

al., 2010). Particularly under strong acidic conditions (*e.g.* high temperature and acid concentration) the furfural products might undergo a conversion reaction to produce unpleasant compounds such as formic and levulinic acids (Larsson *et al.*, 1999). However, a formulation of formic acid, acetic acid and H₂O in a ratio of 30:60:10 effectively yielded 76.5% of wheat straw hemicellulose (Xu *et al.*, 2006), which is comparable to the results of alkaline oxidative extraction (Maes and Delcour, 2001).

5.2.1.4 Enzyme extraction

Naturally, a variety of enzymes are capable of hydrolysing polysaccharides into lower molecular weights or shorter chains or even all the way to their constituent monosaccharides (Dekker, 1985). Other enzymes are capable of mediating the breakdown of the connections between polysaccharides and other components of plant cell wall, or hydrolysing the side chains of polysaccharides and weakening the connection of AX to other components.

Xylanases degrade the xylan backbone in a random manner resulting in a general decrease in the degree of polymerisation (Araki and Kitamikado, 1982; McCleary *et al.*, 1982). Xylanases are one group of endogenous enzymes in cereals (Bergmans *et al.*, 1996; Cleemput *et al.*, 1997; Kulp, 1968; Moore *et al.*, 1990; Preece and MacDougall, 1958; Slade *et al.*, 1989; Taiz and Honigman, 1976). Endo- β -(1,4)-xylanases (EC 3.2.1.8) are a major group of xylanases and the most studied xylanase (Escarnot *et al.*, 2012); belonging to glycan-hydrolysing families GH10 and GH11 (Biely *et al.*, 1997; Henrissat, 1991). Xylanases have been a successful tool for the extraction of AX as they can attack the polysaccharide main backbone and release (solubilise) portions of WUAX (Courtin and Delcour, 2001). GH11 xylanases appeared to be more effective than GH10 for the extraction of AX from wheat bran, with yields of 41% and 18% respectively (Maes *et al.*, 2004). This difference in activity was explained by two theories: GH11 xylanases have the ability to reach the water-insoluble AX more than GH10 (Maes *et al.*, 2004), or they might have greater penetration ability inside the cell wall matrix (Beaugrand *et al.*, 2004).

The association of other enzymes, such as feruloyl esterase, with xylanases for the extraction of AX has been reported as a successful strategy (Faulds *et al.*, 2003; Kroon *et al.*, 1999). Feruloyl esterases break down the ester linkage between AX and ferulic acid, which reduces the attachment of AX to other cell wall components such as lignin, and helps degrade the di-ferulate cross-linkages between AX chains (Faulds *et al.*, 2002; Faulds and Williamson, 1995).

In the same context, the cellulose hydrolysing enzyme, cellulase, has been reported to increase the yield of enzymatically extracted AX when used in combination with endoxylanases, the synergistic effect of this combination possibly a result of the simultaneous release of both cellulose and AX from the cell wall matrix (Escarnot *et al.*, 2011; Murashima *et al.*, 2003; Petit-Benvegnen *et al.*, 1998). Furthermore, α -L-arabinofuranosidases are responsible for "shaving" or removing the arabinose side branches attached to the main backbone, thereby exposing greater unsubstituted xylose regions (Beldman *et al.*, 1996). Due to this effect, the xylan chain will be more vulnerable to xylanase attack and solubilising activity (Figueroa-Espinoza and Rouau, 1998). Surprisingly, the combination of different xylanases together did not enhance the yields of released AX (Beaugrand *et al.*, 2004; Escarnot *et al.*, 2012; Morgan *et al.*, 2017).

In general, enzymatically extracted AX has a lower A/X ratio, which can be explained by the xylanase preference to attack and release the unsubstituted, or less substituted, xylan chain regions, whereas the highly branched regions remain attached to the matrix of the cell wall (Beaugrand *et al.*, 2004). Also, the overall liberation yields are much lower compared to the alkaline hydrogen peroxide method (Zhou *et al.*, 2010).

5.2.2 Mechanically assisted extraction

Fragmentation of the plant cell wall, to enhance AX extraction process, has been the aim of many attempts to apply physical and mechanical technologies. The applied methods aim to force open the rigid texture of the cell wall to expose and/or release its constituents.

5.2.2.1 Ultrasound treatment

The ultrasound technique is based on the application of acoustic energy and solvents to facilitate the release of cell wall components into the solution. The mechanism is based on the creation of fluctuating pressure which propagates throughout the substrate (Reis *et al.*, 2015); the pressure fluctuation generates microscopic and unstable bubbles, which collapse immediately, creating high shear forces against any adjacent material and, as a result, cavitate the material. The temperature in the vicinity of the cavitating bubbles also increases (Ebringerová and Hromádková, 2010). The cavitation phenomenon is capable of enhancing the process of AX extraction when combined with enzymatic or alkaline pathways (Ebringerova *et al.*, 2002; Minjares-Fuentes *et al.*, 2016; Wang *et al.*, 2014). The ultrasonic-assisted extracted AX was reported to show modified functional properties such as weak gel properties and lower intrinsic viscosity (Ebringerova and Hromadkova, 1997; Sun *et al.*, 2002). These functional modifications are primarily explained by the smaller Mw of the extracted AX; it appears that ultrasound not only releases AX from the matrix but also degrades the AX polymers into shorter polymer units (Ebringerova and Hromadkova, 1997).

5.2.2.2 Steam explosion

At the beginning of the twentieth century, Mason (1928) was granted a patent for an apparatus for the disintegration of woody fibrous materials. Simply, the substrate is exposed to highly pressurised steam, then after a certain period of time, the pressure is expeditiously decreased, leading to a substantial breakdown of the structure of the lignocellulosic material (Avellar and Glasser, 1998; Cara *et al.*, 2006; Josefsson *et al.*, 2002). Steam explosion is considered an environmental-friendly and effective pretreatment to degrade biomasses. Steam explosion effectively degrades the interpolymeric phenolic cross-linkages, which leads to the general loosening of the cell wall structure (Merali *et al.*, 2015). However, the powerful steam explosion impact largely depolymerises AX into the constituent oligosaccharides or even monosaccharides (Sun *et al.*, 2005). This degradation is the main drawback of steam

explosion for AX fibre extraction, as it significantly affects the functional properties of the extracted AX.

5.2.2.3 Extrusion

The thermo-mechanical treatment of extrusion is carried out by applying high pressure, temperature and mechanical shear (Honců *et al.*, 2016). The implication of extrusion in AX extraction was investigated. The chemical reactions that occur during extrusion change the properties of the constituent polymers of the material; the breaking down of the side chains and the general depolymerisation notably increase the solubility of the AX chains. Extrusion exhibits many advantages such as short treatment time, low water consumption, high adaptability, productivity and environmental and cost-effective process (Jacquemin *et al.*, 2012; Zeitoun *et al.*, 2010). However, reported yields of AX from extrusion-based processes are relatively low with low purity (Zeitoun *et al.*, 2010).

5.2.3 Purification of AX

The extraction of arabinoxylan from the cell wall structure is based mainly on the liberation and solubilisation of AX polysaccharides into the extraction medium. Unfortunately, this non-selective extraction process acts on other components of the cell wall in the same manner. Substantial amounts of protein, cellulose, lignin and other constituents are extracted as well. Hence, purification of the crude AX extracts is usually required. Generally, extract purification involves the de-activation of the enzymes applied during the extraction procedure and the employment of hydrolysis enzymes to break down the residual starch and protein in the crude extracts, after which the extracts are precipitated and dialysed to obtain the final AX-rich extracts (Saeed *et al.,* 2011; Zhang *et al.,* 2014).

5.2.3.1 Amylase and protease purification

Dervilly *et al.* (2000) successfully employed α -amylase and β -glucosidase for the degradation of residual starch in the AX extracts, leading to the partial removal of the starch from the extracts. Likewise, protease was employed to decrease the protein content in the extracts (Bell, 2015; Dervilly *et al.*, 2000; Izydorczyk and Dexter, 2008; Linares-Pastén *et al.*, 2017).

5.2.3.2 Concentration

Membrane dialysis is one of the most common purification procedures used in a laboratory (Preece and Hobkirk, 1953; Peng *et al.*, 2012). Ultrafiltration is an effective concentration method as it can be used at bench, pilot and commercial scales (Krawczyk *et al.*, 2008). In previous (unpublished) work in the University of Huddersfield, following guidance from Hollmann and Lindhauer (2005) and Bell (2015), ultrafiltration was employed and successfully reduced the volume of the extracts (prior to the ethanol precipitation) to one-fifth the original volume, thereby significantly reducing the amount of ethanol used for precipitating AX and hence the cost and safety hazards. Jacquemin *et al.* (2015a) note that "the problems [of] alcohol recycling (costs and security) hinder the industrial development of xylan production". Therefore, minimising the cost of ethanol usage for AX production is key to successfully bringing new AX-based products to market (Martinez-Hernandez *et al.*, 2018), and ultrafiltration is therefore a key technology for economically viable production of AX.

Meanwhile, diafiltration – dialysis by filtration – is one of the most attractive procedures for the purification of the extracts; the injection of pure deionised water to replace the permeate effectively reduces the non-organic content of the extracts and eliminates the content of the small organic molecules that can pass the ultrafiltration membrane pores. This increases the concentration of the larger molecules (Jacquemin *et al.*, 2012, 2015; Laine *et al.*, 2015).

5.2.3.3 Precipitation

After extraction and purification, AX is recovered from the liquid extract by precipitation. Ethanol had been successfully employed to precipitate AX from the liquefied extracts; as noted throughout this thesis, the ready availability of ethanol is the basis for the economic feasibility of AX production in an integrated biorefinery. The concentration of ethanol used for the precipitation significantly affects the properties of the final product. For example, it was reported that ethanol concentration affects the A/X ratio of the extracted AX; increasing the ethanol concentration usually yields a higher ratio of A/X (Gruppen et al., 1992; Viëtor et al., 1992). Bian et al. (2010) investigated in detail the effect of ethanol concentration on AX precipitation by performing a series of gradual concentrations of ethanol (10, 20, 30, 45, 60 and 80%) and observing the properties of the precipitated fractions. Variations in the branching distribution patterns were reported according to the ethanol concentration. Large molecules of AX with low branching were obtained first at low concentrations of ethanol. Whereas, increasing the ethanol strength caused the low Mw AX with higher branches to start to precipitate (Bian et al., 2010). Martinez-Hernandez et al. (2018) showed how a range of products precipitated at different ethanol concentrations increased the scope for bioethanol pinch analysis to minimise ethanol usage. Underlining how producing a portfolio of AX-based products, with different properties and end-use applications, is the way forward to creating economic processes and bringing affordable products to market.

Another precipitation technique, salting-out, has also been employed for AX recovery; it is believed that adding salt in high concentration to the solution might result in attracting the water molecules to the salt ions, hence decreasing the amount of water available to interact with the polymer. Izydorczyk *et al.* (1992) demonstrated the usage and influence of ammonium sulphate concentration on AX by adding gradually higher concentrations of the salt in the precipitation process (60, 70, 80 and 95%). The results were similar to the ethanol concentration effect, where increasing the concentration of the salt increased the A/X ratio of the resultant AX and precipitated smaller molecules

of AX (Izydorczyk and Biliaderis 1992). However, salting out does not lend itself as naturally as ethanol precipitation for integration into a bioethanol plant.

5.3 Pilot-scale extraction of AX

The lack of commercial sources of arabinoxylans has been the major barrier to establishing their potential role as food ingredients or for other applications (Courtin and Delcour, 2002; Hollmann and Lindhauer, 2005; Du *et al.*, 2009). The majority of investigations reported in the literature describing the structure, properties and potential uses of AX have been performed on bench-scale extracts. These small-scale experiments have allowed a profound understanding of AX properties and potentials. However, the utilisation of AX in food, feed or pharmaceutical investigations requires larger amounts (kg-scale quantities). Also, the characteristics of small-scale extracts might vary from those from scaled-up processes (Annison, Choct, and Cheetham, 1992). Therefore, several attempts have been made to extract AX at larger scales.

Annison *et al.* (1992) reported one of the earliest trials for the extraction of wheat bran pentosans on a large scale. The procedure employed an amylase and protease treatment to remove the starch and protein content of the bran, followed by an alkaline extraction using 0.2 M sodium hydroxide at 80°C for two hours. Thereafter the liquefied AX were separated from the solids by centrifuging. Finally, AX was precipitated by ethanol at 80%. The extraction resulted in 1200 g of AX-rich extract from 15 kg of wheat bran, a yield of 8%, with a purity of about 86%. The apparent Mw of the extract was around 500 kDa with a high polydispersity profile ranging between 5.3 – 758 kDa.

Hollmann and Lindhauer (2005) extracted AX from wheat bran in a study designed to demonstrate commercial feasibility in terms of the safety and efficacy of the materials and processes used. Five kg batches of milled (0.5 mm) wheat bran were boiled in 70% ethanol solution at 80°C for 4 hours. The washed bran was recovered using a 0.4 mm sieve and air dried. WEAX was removed by suspending 3.5 kg of the washed bran in 40 L water and heated to 40°C for 2 hours to remove coloured contaminants, fat and low Mw carbohydrates, and also to deactivate endogenous enzymes. For the extraction

of AX, 2 kg of the washed and WEAX-free bran was suspended in 40 L of 2% hydrogen peroxide solution, adjusting the pH to 11 with sodium hydroxide, and holding for 4 hours at 40°C. Thereafter, the extract was filtered, treated with protease then concentrated by ultrafiltration using a membrane with a molecular weight cut-off of 10 kDa. AX was precipitated by 65% ethanol; the collected extract was treated with β -glucosidase then exhaustively dialysed and finally freeze-dried. By the end of the procedure, 350 g (a 7% yield) of 80% purity AX was obtained, with Mw around 100-110 kDa.

Jacquemin *et al.* (2012) extracted AX from a mixture of wheat straw and wheat bran in a ratio of 6.2:1. A slurry of 86 kg material was mixed with 5.8 kg sodium hydroxide, and water (at liquid/solid ratio of 10) and injected into a co-penetrating and co-rotating twin-screw extruder. The liquids were separated by centrifugal filtration, the filtrate was concentrated by ultrafiltration using a membrane with a molecular weight cut-off of 30 kDa. The concentrated supernatant was passed over strong anion-exchange resin, then AX was precipitated using 65% ethanol. The reported yield was 33.7% with a total sugar content of 50%.

Arabinoxylan extraction requires a lengthy procedure of numerous steps, with various reports of procedures for the extraction and for the number and range of subsequent purification steps. The obvious common obstacle in all the proposed procedures is the extensive consumption of ethanol, which makes the extraction of AX expensive and hazardous. This has hindered the development of routine AX extraction to produce reasonable amounts for technological studies or for implementation at commercial scale.

Therefore, this part of this project aimed to extract AX on a larger (pilot) scale, to be used for the oligosaccharide production for animal feed trials within bioethanol biorefinery where ethanol can be used for the extraction, recovered by distillation and re-used again.

5.4 Materials and methods

5.4.1 Chemicals and enzymes

Sodium hydroxide pellets, hydrogen peroxide (>30%w/v), sulphuric acid 98% and trifluoroacetic acid 99.5% were purchased from Fisher Scientific UK Limited. The antifoaming agent used was Dimeticon SILFAR[®] SE 4 made by Wacker Chemie AG, Germany, and provided kindly by IMCD Group, UK. High viscosity wheat arabinoxylan was purchased from Megazyme (Bray, Ireland).

The enzymes used for the purification, α -amylase and gluco-amylase, were purchased from ReKru GmbH, Germany. Cellulase from *Aspergillus sp.* (C2605) and protease from *Bacillus licheniformis* (P4860) were purchased from Sigma Aldrich, UK.

5.4.2 Equipment

The extraction was carried out using the double jacketed mash tank of the GUNT CE-640 bioethanol plant shown earlier in Figure 3-4. Extracts were centrifuged using a Beckman centrifuge GS-6S with swing bucket rotor and capacity of 4 × 750 ml (Beckman Coulter Ltd., UK), shown in Figure 5-1.



Figure 5-1. Beckman centrifuge GS-6S.

For the concentration of the samples, a hybrid in-house assembled ultrafiltration system was employed as detailed below. Extracts were dried in a Christ Freeze Dryer Alpha 1-

4 ld plus (Martin Christ Gefriertrocknungsanlagen GmbH, Germany). Samples were prepared for the analysis using Zymark Turbovap[®] LV sample concentrating unit, MA, USA.

For the molecular weight evaluation, size exclusion chromatography (SEC-MALLS) was used, comprising a Shimadzu HPLC system (Shimadzu U.K. Ltd, Milton Keynes, UK) connected to three columns in series: Aquagel-OH 40, 50 and 60 (15 μ m particle size, 25 cm × 4 mm, Agilent, Oxford, UK). The detection system comprised three detectors: UV/VIS detector SPD-20 A, differential index detector (Optilab rEX) and multiangle laser light scattering (MALLS) detector (mini-DAWN, Wyatt Technology, Santa Barbara, CA, USA). The eluted mobile phase was 50 mM NaNO3 (containing 0.02% NaN3 as a preservative) at a flow rate of 1 mL min⁻¹. LC Real Time[®] software was used for the integration and analysis of the chromatographs.

For sugar analysis, a Dionex ICS–3000 high performance anion exchange chromatography with pulsed amperometric detection (HPAEC-PAD) (Dionex Corporation, CA, USA) was used.

For concentrating the samples, two ultrafiltration systems were available: a small-scale Sartorius[®] Vivaflow[®]200 system, and a larger Alfa Laval system, both shown in Figure 5-2. The Sartorius[®] Stedim Vivaflow[®]200 Ultrafiltration system (Sartorius Stedim Biotech, Göttingen, Germany) uses cross-flow membrane cassettes each with a membrane surface area of 200 cm² and 10 kDa molecular weight cut-off. The system is equipped with a Masterflex[®] Easy-Load[®] peristaltic pump. The second system is an Alfa Laval LabStak[®] M10 ultrafiltration system (Alfa Laval Ltd, UK). The LabStak[®] M10 has a cross-flow flat sheet membrane with a surface area of 336 cm² and 10 kDa molecular weight cut-off, and an Eco[®] GC4-KDTTTUG gear pump.

89



Figure 5-2. Ultrafiltration systems: (A) Vivaflow, (B) LabStak.

Both ultrafiltration systems were designed to process small samples on bench scale with maximum filtration rate around 200 mL h⁻¹. For pilot-scale extraction, the average sample volume is around 30 litres, therefore, concentrating the sample to one third would take at least two weeks of continuous operation. Looking at both systems, the Vivaflow system has high quality membrane cassettes coupled with a peristaltic pump of limited capability, whereas the Alfa Laval system is coupled with a powerful gear pump and a membrane cassette of a small surface area which limits the performance of the system.

Therefore, utilising the Eco[®] gear pump to feed four Vivaflow cassettes at the same time, as seen in Figure 5-3 and Figure 5-4, was implemented to scale up the ultrafiltration system and increase the overall performance. A pressure gauge and a safety pressure relief valve were attached to the new system in order to control the pressure and protect the membranes from high pressure load. All the connectors, manifold, valves and tubes needed to assemble the hybrid system were purchased from Amazon, UK.



Figure 5-3. Design of the hybrid ultrafiltration system.



Figure 5-4. Assembled ultrafiltration system.

In this part of the project, two extraction batches are presented; for the first batch, two Vivaflow systems were used simultaneously, while for the second batch the hybrid system with four Vivaflow cassettes was employed.

5.4.3 Arabinoxylan extraction method

Arabinoxylan was extracted from the fresh DWG by the alkaline oxidative method; two batches of AX extracts are presented in this work (AXa and AXb), varying in the ultrafiltration system used and the subsequent purification employed. Both extraction batches were performed according to an extraction procedure adapted from Hollmann and Lindhauer (2005) with modifications to accommodate the starting material and the available laboratory facilities.

The extraction procedure for both AXa and AXb was initiated by suspending 10.1 kg (2.55 kg db) DWG in 25 L of water and heating to 50°C in the double-jacketed mash tank of the CE-640 plant. Alpha-amylase, gluco-amylase and cellulase were added at dosages of 2 mL, 4 mL and 5 mL, respectively, and the mixture incubated at 50°C for 6 hours. The pH was adjusted to 6.5 during this incubation. Protease was added at a dose of 8 mL after adjusting the pH to 8; the incubation continued at 50°C for another 10 hours. Sodium hydroxide solution (40%) was added to raise the pH to 11, the pH was monitored during the extraction process, and alkaline solution was added when needed to maintain the alkali conditions. The temperature was increased to 60°C using direct steam injection, and 10 mL of the anti-foaming agent was added. Hydrogen peroxide was added slowly at a rate of 100 mL every 5 min, to avoid excessive foaming, while maintaining vigorous stirring. A total of two litres of 30% hydrogen peroxide was added over a period of about 3 hours; the extraction was continued for a total of 4 hours.

At the end of the extraction, the mixture was cooled to 25°C and neutralised with concentrated sulphuric acid. The whole slurry was centrifuged, supernatants were collected, the solids were washed with 5 L deionised water, the mixture was centrifuged, and the supernatant was poured into the previously collected supernatant.

For the first batch, AXa, the liquids were ultrafiltered using two Vivaflow systems over a 10 kDa membrane to reduce the total volume to one third, *i.e.* 27 L of filtrate was concentrated to 9 L. Thereafter, a sufficient amount of industrial methylated spirit (18 L) was added to give a final concentration of ethanol of 65% and left overnight at 4°C to precipitate the AX. The precipitated AX was recovered by centrifugation, freeze-dried and stored in a plastic bag. Figure 5-5 summarises the extraction procedure.



Figure 5-5. AXa extraction process.

For the second batch, AXb, the extract solution was concentrated using the hybrid ultrafiltration system, which was slightly faster than the original Vivaflow system. After concentrating the extract, AX was precipitated by ethanol, centrifuged and dried in an oven at 40°C overnight. Thereafter, the extract was dissolved in 10 L of deionised water and enzymatic treatments utilising amylase, protease and laccase were applied and

followed by ultrafiltration, ethanol precipitation, centrifuging and oven drying. Figure 5-6 shows the purification steps applied.



Figure 5-6. AXb extract purification method.

5.4.4 Characterisation of the extracts

Due to the strict time scale of the animal feed trials carried out by the academic partner in Nottingham Trent University, batches were characterised differently as follows:

For the first extract, AXa, the constituent sugar analysis was performed by Englyst Carbohydrates Ltd. The Kjeldahl analysis method was used for protein content (Section 3.6.3), and the ash content was determined as described in Section 3.6.4.

For the second extract, AXb, protein, ash and total carbohydrate content were analysed using the same methods used for AXa. The average molar masses (Mw, weight average molar mass and Mn, number average molar mass) were evaluated by size exclusion chromatography (SEC). The extract was solubilised in 50 mM NaNO3 solution (the mobile phase) at a concentration of 3 mg mL⁻¹ under continuous stirring overnight at room temperature. The sample was injected into the system after calibration with dextran of known molecular weights, and weight average molar mass and polydispersity index calculated.

The constituent sugars analysis was performed utilising the HPAEC-PAD system. The sample was solubilised in ultrapure water (at a concentration of 2 mg mL⁻¹) under continuous stirring overnight at room temperature. Precisely 1 mL of the solution was pipetted into a pressure tube (Ace glass), trifluoroacetic acid was added to make the acid concentration equal to 4 M, the tube was closed tight and moved to a heat block at 120°C for two hours. The tubes were left to equilibrate to room temperature, then the contents were quantitatively moved to 15 mL centrifuge tubes and dried under a continuous compressed air flow at 60°C using a Turbovap® evaporator. After complete drying, the sample was re-dissolved in MilliQ water, filtered over a 0.45 μ m syringe filter and injected into the chromatography system.

In order to calculate the recovery of the hydrolysis method, arabinoxylan high viscosity standard (Megazyme) was used to spike the sample at three levels of concentration (25, 50 and 75 ppm), hydrolysed according to the above method and analysed by the HPAEC system.

5.4.5 Enzyme treatment

Extract AXb was subject to enzymatic treatment with Econase to investigate the potential of XOS production from the chemically extracted AX from the Distillers Wet Grain. The extract was solubilised, Econase was added at four dosages 0X, 1X, 10X and 100X where X is 50 mL per kg, pH was adjusted to 5.5, and the incubation lasted for one hour in a shaking water bath at 55°C. At the end of the incubation period, samples were submerged in an ice bath to stop the enzyme activity, freeze-dried and submitted to IPOS for XOS analysis.

In order to investigate the branching (substitution) effect on the enzyme activity, a solubilised sample of the extract was subject to arabinofuranosidase activity for 24 hours. Then four samples were prepared following the same steps described above.

5.5 Results and discussion

5.5.1 Extracts characterisation

Extracts AXa and AXb were 900 g and 420 g, respectively, and almost identical in appearance (odourless with a non-crystalline starchy appearance after milling, and white in colour), shown in Figure 5-7.



Figure 5-7. AXa extract after milling.

Protein contents were 14.00 ± 0.15 and $10.00 \pm 0.20\%$ for AXa and AXb, respectively, giving a total amount of protein of 126 g in AXa and 42 g in AXb. Evidently the purification treatment successfully decreased the amount of the protein in AXb to one third of the total protein in the un-purified extract (AXa), although other components were also removed, such that the protein concentration only decreased from 14% to 10%. The prolonged protease treatment had been expected to eliminate the protein content in the extract by degrading the total protein to smaller molecules that can pass the ultrafiltration membrane's pores; evidently the degradation was incomplete even after 24 hours. Apparently, the protease was incapable of degrading the proteins; this could be a result of the wheat protein denaturation during the bioethanol process (*i.e.*

the distillation). Both De Vries *et al.* (2013) and Schroeder (2012) reported major changes in the nutritional value of DDGS proteins and the increased amount of undegradable proteins in the DDGS. These changes were attributed to the drying process of the DWG to the DDGS, but the current results suggest these changes in protein digestibility occur partially during the ethanol distillation prior to the DDGS drying.

The ash content in the extracts was $40.08 \pm 0.11\%$ for AXa and $13.75 \pm 0.38\%$ for AXb. The huge ash content is from sodium sulphate resulting from the NaOH used in the extraction and the sulphuric acid used for the neutralisation. The purification steps applied in the second extraction, ethanol precipitation and diafiltration, significantly reduced the amounts of salt in the extracts from 360 g in AXa to 57.75 g in AXb. The ethanol precipitation step had a minor effect of the salt content as sodium sulphate is ethanol insoluble and precipitates in the 65% ethanol solution along with the AX. Hydrochloric acid should be used for the neutralisation in future subsequent extraction processes as it is more soluble in alcoholic solutions (Farelo *et al.*, 2004).

The constituent sugars analysis of the AXa extract by Englyst Labs showed an arabinose content of 8.7 \pm 0.1% and xylose 11.1 \pm 0.1%, the total amount of AX is calculated as 19.5 \pm 0.2% of the extract. This represents about 175 g of the total 900 g extract. Considering that the amount of AX in the starting 2.55 kg DWG was 16.6% (db), corresponding to 423 g, the yield of the extraction can be estimated at 41% of the available AX in the DWG. Moreover, the AX content in the original wheat is 5.9% (713 g), this means that the extraction yielded 25% of the total AX in original wheat.

The second extract, AXb, was analysed using the HPAEC-PAD system. The analysis showed an arabinose content of 21.4 ± 0.2 % and xylose content of 24.3 ± 0.11 %. The total amount of AX was 44.4 ± 0.3 % of the extract, which represents about 186 g of the total 420 g extract. This is slightly higher than the amount of AX in the first extract and represent 43.6% of the available DWG AX and 26% of the original wheat AX.

Clearly, the purification process did not reduce the total amount of AX recovered, which indicates the stability of the extracted AX and suggests further treatments could be applied to increase the purity further without compromising the absolute yield.

Traditionally, the first step in the analysis of polysaccharides is the breakdown of polysaccharides into monosaccharides, for which acidic hydrolysis had been the method of choice. Countless reports in the literature have used acidic hydrolysis for polysaccharide analysis, using an unlimited variation of conditions: hydrolysis temperature, acid concentration, hydrolysis time and, most importantly, different substrates (biomass and extracts), mostly with minimal justification for the conditions used. In general, the acid hydrolysis process can be summarised in three major steps: the breakdown of the biomass structure to release the major components (the intermolecular bonds); hydrolysing the biomass component into its basic units (the intramolecular bonds); and the degradation of the resultant units (monosaccharides into furfural products). Acid hydrolysis of polysaccharides is therefore a balance between the first two steps, the producing steps, and the third degrading step. This balance has been for a long time the "elephant in the room" due to the impact of the variation in the hydrolysed material structure, composition and degree of polymerisation on the kinetics of the hydrolysis (Ranganathan et al., 1985; Puls, 1992; Willför et al., 2009; Banerjee, 2014). As a wider range of polysaccharides such as AX become of increasing scientific and commercial interest, the issue of hydrolysis for reliable sugar analysis is going to become increasingly important.

Moreover, preliminary results of analysing the same AX extract sample before and after the xylanase treatment following the same hydrolysis procedure showed a huge difference in the detected amount of AX, which also emphasises the impact of the degree of polymerisation of the polysaccharide on the degradation of monosaccharides during the hydrolysis. Englyst *et al.* (1994) and Sluiter *et al.* (2008) studied the recovery of the acidic hydrolysis using standards of monosaccharides. This ignores the matrix structure of the carbohydrate as the rate of degradation of the added monosaccharides will be different from the original constituent monosaccharides which first have to be released. According to Willför *et al.* (2009) "there is only little sense in calibrating with monosaccharides".

Fortunately, the recent availability of commercial AX pure standards with a variety of molecular weights has enabled the relatively accurate estimation of hydrolysis recovery by spiking the sample with a similar pure AX polymer. Using this approach, the calculated recovery in the current work was estimated at 90.8% of the hydrolysed arabinoxylan. Table 5-1 shows the calculated recovery of the hydrolysis method from triplicate samples. Clearly, there is a minor effect related to the amount of the AX in the sample. At all levels of added AX, the recovery was consistently 90.8%; evidently, there was a negligible effect of the amount of the AX in the sample, which reflects the effect of polysaccharide quantity on the kinetics of the hydrolysis.

Original sample	AV standard	Theoretical	Detected		
concentration	addod ppm	concentration	concentration	% Recovery	
ppm	ppm added ppm		ppm		
20.14	0.00	20.14	20.14	—	
20.14	25.00	45.14	42.84	90.81 ± 0.15	
20.14	50.00	70.14	65.54	90.80 ± 0.01	
20.14	75.00	95.14	88.22	90.78 ± 0.01	

Table 5-1. AX recovery following hydrolysis (IPOS).

The molecular weight of AXb, measured using SEC-MALLS, showed an average Mw of 234.7 \pm 0.009% kDa. The calculated polydispersity index (Mw/Mn) was 1.6 \pm 0.013% kDa, which indicates a wide distribution of the molecular sizes in the extract. The average Mw was nearly double that reported by Hollmann and Lindhauer (2005), who extracted from washed wheat bran after the gentle removal of WEAX at 40°C, whereas the extraction was from DWG in the current project. Clearly there are numerous possible reasons for the difference, but it is reasonable to suppose that the distillation process (continuous boiling for 4 hours) was able to release a greater amount of small molecules of AX into the Solubles and washing them out than the less severe washing procedure reported by Hollmann and Lindhauer (2005).

5.5.2 Enzymatic treatment results

In order to explore the suitability of the DWG AX for the production of arabinoxylan oligosaccharides, samples of AXb were incubated with Econase then sent to IPOS to run their DPn method and estimate the content of the produced XOS. Theoretically, the produced extract should contain more available AX for the enzyme activity than the content of the raw material. Therefore, Econase was applied in three levels of dosages: 1× (the base level of 50 mL per kg), 10× and 100×, plus a 0× blank sample.

Table 5-2 presents the XOS percentage released from the AX extracts after the Econase treatment. Surprisingly, the enzyme showed only very small effects on the extract; at the 1× dose there was some evidence of production of the xylose monomer and xylobiose (DP2) and xylotriose (DP3), but no evidence of the higher oligosaccharides. The higher dosages increased production of DP2 and DP1, the latter unexpected as monosaccharide production should not be seen from an endoxylanase.

%	DP1	DP2	DP3	DP4	DP5	DP6
0×	0.0	0.0	0.0	0.0	0.1	0.0
1×	2.1	0.4	0.1	0.0	0.0	0.0
10×	4.6	1.0	0.1	0.0	0.0	0.4
100×	24.8	5.9	0.2	0.0	0.4	0.0

Table 5-2. XOS released from AXb after Econase treatment (IPOS).

This limited xylanase activity could be a result of the branching pattern of the extracted AX; the A/X ratio of 0.88 indicates highly branched molecules which would be less susceptible to enzyme attack, the arabinose side chains hindering access to the xylan backbone. Therefore, the extract was subject to treatment with arabinofuranosidase, an enzyme that acts to remove arabinose side chains. This "shaving" activity should expose the xylan chains and grant more access for Econase. Hence, the extract was treated for 24 hours with arabinofuranosidase at 40°C and pH 5, then treated with Econase under similar conditions to the previous treatment. The samples were coded as A0×, A1×, A10× and A100×, as shown in Table 5-3.

%	DP1	DP2	DP3	DP4	DP5	DP6
A0×	0.6	0.0	0.0	0.0	0.0	0.0
A1×	2.1	0.4	0.0	0.0	0.0	0.0
A10×	4.6	1.0	0.1	0.0	0.0	0.4
A100×	23.0	5.4	0.2	0.0	0.4	0.1

Table 5-3. XOS released from AXb after arabinofuranosidase and Econase treatment.

Clearly, pretreatment with the "shaving" enzyme had no effect on the Econase performance. The arabinofuranosidase treatment shows only a minor difference in the monosaccharides content in the first sample (A0×), compared to Econase treatment in Table 5-2, while the sample only treated with the arabinofuranosidase had almost the same monosaccharide content in the samples 100× and A100×. Since the analytical method used lacked the ability to distinguish the type of monosaccharide, it is not possible to determine whether the monosaccharides resulting from the treatment are arabinose from the arabinofuranosidase activity or xylose from the Econase activity.

Since only a minor xylanase effect was observed using the DPn method on the concentration of oligosaccharides released, samples 0X and 10X were analysed by SEC-MALLS in order to investigate the effect on the large molecules. Table 5-4 presents the molecular weight and polydispersity index for the AXb sample before and after the Econase treatment. Clearly, Econase acted to break down the large molecules; the decrease in the average Mw to about half, along with the increase of the polydispersity index, indicates the breakdown of the large AX polysaccharides into a wide range of smaller polysaccharides. So, it is evident that the Econase acted on AX molecules in the AXb extract, reducing molecular weight but with only a minor production of oligo-level saccharides. It is of course possible that the xylanase is very active towards these oligosaccharides, such that as soon as they are produced they are degraded.

Table 5-4. Molecular weight and polydispersity index for AXb before and after Econasetreatment.

Samples	Mw	(kDa)	Mw/Mn		
0X	234.7 ± 0.009%		1.6	± 0.013%	
10X	106.0	± 0.014%	2.1	± 0.020%	

The change in the molecular weight distribution induced by Econase activity is clearly present in the cumulative molar mass elution profile of the extracts present in Figure 5-8. The graph shows a general reduction in the molecular weight of the sample components, as well as reduced homogeneity of the Mw distribution.



Figure 5-8. Cumulative molar mass of AXb before and after Econase treatment.

It is evident that Econase is performing on the extract, and it is plausible that more extended treatment could facilitate the degradation of AX to the desired oligosaccharides. However, the amounts of the monosaccharides produced along with the oligosaccharides would not provide the ideal oligosaccharide substrate for animal feed trials or it would require further purification steps. The inability to identify the monosaccharides produced limits the ability of the IPOS method to determine the enzyme activity pattern with any precision. These results also raise questions about the enzyme behaviour towards oligosaccharides; it is important to clarify whether the enzyme is producing and degrading the oligosaccharides to smaller oligos and all the way to monosaccharides. Such a question has not been addressed in the literature, as the general understanding of endoxylanases is that they do not produce monosaccharides.

5.6 Summary

The extraction of AX from the bioethanol in-process material, DWG, was performed using the CE-640 plant to produce the DWG and to undertake the extractions. Two batches of AX were extracted to study the potential for XOS production using the commercial xylanase Econase. The first batch was sent to Nottingham Trent University to be tested in animal feed trials aimed at establishing the benefits of adding the AX into the diets of chickens. The second batch was subjected to several enzyme treatments aiming to produce significant amounts of XOS material. The preliminary experiments showed that Econase acted on the AX in the extract, as evidenced by the reduction in average molecular weight, but its activity appeared to be hindered by the high arabinose content. Econase could not produce XOS from the extracts as it had a high branching pattern.

The DPn chromatographic method has only been recently developed by IPOS, and the performance of this technique is not well established yet. Also, the developed method lacks the ability to distinguish the released mono- and oligosaccharides upon treatment with the enzymes. The identification of these saccharides is crucial to understand the effect of the enzymes on the extracted AX. Therefore, while the IPOS analyses allowed the project to move forward usefully at the time, the next step of this project was dedicated to developing and validating a new analytical method for the accurate screening of the enzyme degradation products.

6 Oligosaccharides analysis method development

6.1 Introduction

Arabinoxylan oligosaccharides (AXOS) and xylan oligosaccharides (XOS) are an evolving subject of interest due to their impact on human and animal health via prebiotic activity (Ou and Sun, 2014; Suwa et al., 1999). The increasing interest in (A)XOS highlights the importance of establishing robust analytical procedures that can routinely measure and characterise these prebiotics in a variety of matrices; accurate measurement of the amount of prebiotics in food and feed is a prerequisite for their production and exploitation. A significant amount of research has been conducted using highly sophisticated instrumentation to characterise the conformation and structure of (A)XOS. Meanwhile, the recent availability of commercial high purity standards of these prebiotics has facilitated the development of several techniques for the measurement of (A)XOS. Chromatographic separation methods, including liquid and gas chromatography, have been applied for the separation and measurement of these oligomers. However, the currently available methods either require lengthy sample preparation steps, in the case of GC, or lack the resolution and the ability to measure both mono- and oligosaccharides simultaneously, in the case of HPLC. This chapter reviews the available analytical methods for the determination of (A)XOS in literature, then describes the development and validation of a new method for the simultaneous determination of 16 analytes including mono- and oligosaccharides and uronic acids.

6.2 Oligosaccharides analysis methods

Historically, the main challenges facing the development of methods for measuring carbohydrates were the lack of a high resolution separation method and the difficulty of detecting low concentrations of carbohydrates (Rocklin and Pohl, 1983). Classic analytical techniques used for oligosaccharide separation included paper, thin-layer, high performance thin layer, ion exchange and gel filtration chromatography (Gauch *et al.,* 1979; Voragen *et al.,* 1986). Now the fast growth of enzyme markets and the

increased demand for oligosaccharides is demanding more simplified, fast and accurate analyses that can be exploited in routine standard procedures. Table 6-1 summarises the methods reported in the literature for the analysis of oligosaccharides. Ion exchange methods with refractive index detection have been widely used, while High Performance Anion Exchange Chromatography with Pulsed Amperometric Detection (HPAEC-PAD) has become increasingly the method of choice.

				-	
Method	Analytes	Mobile Phase	Stationary phase	Detection	Reference
	DP 2-4	Water	Aminex HPX-87P	Refractive index	Voragen <i>et al.</i> 1986
	DP 2-5	Water	Aminex HPX42A	Refractive index	Brienzo, Carvalho, and Milagres 2010
	DP 2-5	Water	RSO- oligosaccharides column Phenomenex	Refractive index	Sabiha-Hanim, Noor, and Rosma 2011
	DP 2-6	Water	Aminex HPX-42A	Refractive index	Uçkun Kiran <i>et al.</i> 2013
lon exchange	DP 2-3	Acetonitrile and water	ZORBAX carbohydrate	Refractive index	Jayapal <i>et al.</i> 2013
	DP 2-4	Water	Aminex HPX-87C	Refractive index	De-Figueiredo <i>et al.</i> 2017
	DP 2-6	Water	Aminex HPX 42A	Refractive index	Akpinar <i>et al.</i> 2010
	DP 2-5	Water	RSO- Oligosaccharide Ag + 4 % Phenomenex	Refractive index	Wan Azelee <i>et</i> <i>al.</i> 2016
	DP 2-6	5 mM H ₂ SO ₄	Aminex HPX-87H	Refractive index	Lin <i>et al.</i> 2017
	DP 2-6	Water	Aminex HPX-42A	Refractive index	Surek and Buyukkileci, 2017
RP- HPLC	DP 2-4	Sodium phosphate and acetonitrile	C18 column	UV at 245 nm	Morgan <i>et al.</i> 2017
	r		1	1	
GC- MS*	DP 2-4	Helium	HT5 Tagged "polycarborane siloxane"	Mass- spec	Hernández <i>et al.</i> 2009
	1		1	1	1
HPTLC	DP 2-5	Acetonitrile and water (85:15)	TLC plates of silica Gel 60 F254	0.2% Orcinol	Chapla, Pandit, and Shah 2012

 Table 6-1. Analytical methods for the separation of oligosaccharides.

Method	Analytes	Mobile Phase	Stationary phase	Detection	Reference
	-	Butanol:Pyridine:Water (6:4.5:2.5)	TLC plates of silica Gel 60 F254	UV light at 366 nm	Gowdhaman and Ponnusami 2015
TLC	DP 2-5	n-butanol/acetic acid/water (2:1:1, v/v/v)	TLC plates of silica Gel	0∙02% Orcinol	Benamrouche <i>et al.</i> 2002
	DP 2-6	2000mM sodium acetate and 200mM sodium hydroxide	CarboPac PA-1		Gullón <i>et al.</i> 2008
НРАЕ	DP 3-6	400mM sodium acetate and 100mM sodium hydroxide	CarboPac PA-100		Courtin <i>et al.</i> 2009
	DP 2-6	500mM sodium acetate and 100mM sodium hydroxide	CarboPac PA200	Pulsec	Zhang, Xu, and Yu 2017
	DP 2-6	500mM sodium acetate and 100mM sodium hydroxide	CarboPac PA100	d ampero	Álvarez <i>et al</i> . 2017
C-PAD	DP 2-6	450mM sodium acetate and 150mM sodium hydroxide	CarboPac PA100	metric de	Li <i>et al.</i> 2013
	500mM sodium DP 2-6 acetate and 80mM sodium hydroxide		CarboPac PA100	tector	Lafond <i>et al.</i> 2011
	DP 2-6	1000mM sodium acetate and 200mM sodium hydroxide	CarboPac PA100		Arruda, Pereira, and Pastore 2017
	DP 2-6	300mM sodium acetate and 100mM sodium hydroxide	CarboPac PA100		Bian <i>et al.</i> 2013, 2014

*Two-steps derivatisation was performed (oximation and trimethylsilylation)

High performance liquid chromatography (HPLC) was one of the earliest techniques used for the separation of oligomers from enzymatic degradation of plant cell wall polysaccharides after the development of fixed-ion resin columns (Schnaak, 1984). The addition of metal ions to the resin of ion exchange columns enhanced the separation of the oligosaccharides. Calcium resins proved efficient in the separation of monosaccharides, while silver and lead columns were more effective for the separation of oligosaccharides. Voragen *et al.* (1986) used an HPLC system with lead-resin column and a refractive index detector for the separation of XOS. The applied method lacked resolution as shown in Figure 6-1, while the precipitation of proteins and uronic acids

on the column required extensive cleaning and preparation of the sample prior to the analysis, as well as frequent change of the guard column. Moreover, the method was limited to the detection of DP1-DP4



Figure 6-1. Separation of Xylooligosaccharides by HPLC-RI (Voragen et al., 1986).

Applying an ion-exchange column increases the ability to separate the oligosaccharides, and better detection can be achieved with a refractive index (RI) detector (Brienzo *et al.,* 2010; Sabiha-Hanim *et al.,* 2011; Jayapal *et al.,* 2013; Uçkun Kiran *et al.,* 2013; De Figueiredo *et al.,* 2017). The main drawback with this method is the low sensitivity of the RI detector for carbohydrates as well as the high sensitivity of the detector to any factor that affects the refractive index such as temperature, pressure and mobile-phase composition. Furthermore, the incompatibility of the RI detector with gradient elution (as the gradient elution changes the refractive index of the mobile phase) significantly hampers the efficiency of the separation (Swartz, 2010).

A degree of in-depth understanding of the formation and the structure of XOS and AXOS has been obtained via more sophisticated methods such as mass spectroscopy (MS) and
Nuclear magnetic resonance (NMR) (Teleman *et al.*, 1995). Gruppen *et al.* (1992) used NMR to characterise the products of enzymatic hydrolysis of alkali-extracted AX from wheat flour after ion exchange separation, revealing the type and location of the arabinofuranosyl groups. Hoffmann *et al.* (1992), using the same technique, described the sequence of the branching of XOS by studying the structure of hepta- to tetradeca-saccharides obtained by digestion of wheat endosperm arabinoxylan with endoxylanase, concluding that AXOS with two 2, 3-branches are separated by one or two unbranched xylose units. Despite the insights revealed by these techniques (achieved despite the lack of commercial standards at the time), the cost of the equipment used for the analysis limits the availability of these methods for routine analysis.

High performance liquid chromatography (HPLC) has become one of the most reliable techniques used in food and pharmaceutical research. Morgan *et al.* (2017) used a reversed phase HPLC system with C18 column and UV detection at 245 nm with a mixture of sodium phosphate buffer and acetonitrile as a mobile phase for the separation of XOS. The method was limited to measuring XOS of degree of polymerisation up to four monomers (DP4), as shown in the chromatogram in Figure 6-2. Moreover, the lack of chromophore groups in the carbohydrates limits the UV detection ability to detect low concentration.



Figure 6-2. Chromatogram of XOS standards DP 2-4 (1000 ppm) analysed by RP-HPLC (Morgan et al., 2017).

Apparently, the utilisation of RP-HPLC does not give the optimum picture of the xylanase activity or the oligosaccharide profile in the samples, especially as very high concentrations of the analytes are needed in order to be detected.

Gas chromatography (GC) is widely used for carbohydrate determination; however, due to the non-volatile nature of monosaccharides and oligosaccharides, a lengthy derivatisation procedure is required prior to the analysis. Methylation of sugars for GC analysis is used to determine linkage types between component monosaccharide units in the oligo- and polysaccharides. This is usually accomplished by treating the polysaccharide with methylsulphinyl carbanion to create polyalkoxide ions, followed by treating the mixture with methyl iodide to add a methyl group (Harris *et al.*, 1984; Sandford and Conrad, 1966). The methylated polysaccharides then undergo a hydrolysis step to break down the polysaccharide into the constituent monosaccharides, which undergo a reduction step followed by acetylation. The resultant alditol acetates of methylated sugars are separated by GC-MS. Hernández *et al.* (2009) reported a successful quantification method that separates mono-, di-, tri- and tetra-saccharides using gas chromatography–mass spectrometry (GC–MS) with a two-step derivatisation involving oximation and trimethylsilylation.

Despite the importance of methylation analysis in evaluating the qualitative linkage pattern in polysaccharides, the quantitative utilisation of this procedure is hindered by several factors. The alditol acetate derivatives of the methylated sugars might result in under-methylation, and the kinetics of the reaction are not well studied. Also, the recovery of the acidic hydrolysis, the monosaccharides' degradation and demethylation during the hydrolysis, are not well established. Another factor is the unstudied recovery of both reduction and acetylation reactions, and the losses of volatile products during evaporation of solvents while performing the various steps. All these factors hinder the utilisation of the methylation method for quantitative analysis.

Among all the chromatographic techniques, high performance anion exchange chromatography with pulsed amperometric detection (HPAEC-PAD) has been the method of choice for carbohydrates analysis (Rocklin and Pohl, 1983). The utilisation of

pulsed amperometry has enabled sensitive detection, which is indispensable in the case of complex samples. Recently, the commercial availability of XOS standards up to DP6 and a smaller number of AXOS standards has facilitated the separation of these analytes in a simple and direct analysis using an HPAEC-PAD system. Falck *et al.* (2014) determined the XOS and AXOS produced by different xylanases using a CarboPac PA200 column and a fixed 100 mM sodium hydroxide with a linear increase of sodium acetate from 0 to 120 mM within a 30 minutes run, as illustrated in Figure 6-3. The accurate determination of xylanase activity by directly measuring the produced oligosaccharides has thus become a routine analysis.



Figure 6-3. Analysis of XOS with HPAEC-PAD (Falck et al., 2014).

Cürten *et al.* (2018) reported a faster method for the analysis of XOS as an automated method for the assay of xylanases activity using a CarboPac PA100 column with a more concentrated mobile phase composition. They also studied the effect of sodium hydroxide concentration, sodium acetate concentration, column temperature and flow rate, as illustrated in Figure 6-4.



Figure 6-4. Effect of mobile phase strength, column temperature and flow rate on the separation of XOS (Cürten et al., 2018).

Other reports have showed the effective usage of HPAEC-PAD for the quantification of XOS and/or AXOS, variously using CarboPac columns PA1, PA100 and PA200 to measure XOS in the range DP2-6 (Gullón *et al.*, 2008; Courtin *et al.*, 2009; Lafond *et al.*, 2011; Li *et al.*, 2013; Bian *et al.*, 2013, 2014; Arruda *et al.*, 2017; Zhang *et al.*, 2017). All of these used a fixed (isocratic) concentration of the main solvent, sodium hydroxide, with gradient elution of sodium acetate, in varying proportions and molarities.

The fast and reliable analysis has made the HPAEC-PAD technique the favoured choice for oligosaccharides analysis. However, the isocratic elution of strong sodium hydroxide solution is recommended by the manufacturers, as the manual of CarboPac PA 200 column reads (Dionex Corporation, 2004, p. 11):

"To maintain baseline stability, it is important to keep the sodium hydroxide concentration constant during the sodium acetate gradient, because acetate has no buffering capacity at high pH."

The isocratic elution of strong eluents severely reduces the resolution of the separation of the monosaccharides and to some extent the oligosaccharides; these recommendations are probably inherited from the first generations of ion exchange techniques when RI detectors were the only option. Practically, the low retention of the small oligosaccharides (DP2-3) makes the analysis of these XOS almost impossible in complex samples such as food and feed products, which can be rich with proteins and amino acids. Figure 6-5 shows a chromatogram of an isocratic elution method for the analysis of XOS content in a real animal feed sample performed in AB Enzymes labs (unpublished work). Clearly, the amount of interfering materials has completely covered the area of the small XOS elution, and it was impossible to determine the content of the oligosaccharides without developing a higher resolution method.



Figure 6-5. XOS analysis in animal feed sample (Unpublished work, AB Enzymes labs).

Moreover, the analysis of monosaccharides required different conditions from those for oligosaccharides, as the strong solvent elution used for the oligosaccharides elution is not suitable for the monosaccharides; hence, to get the full picture of the enzyme hydrolysis activity, each sample needs to be analysed twice under two different conditions with two different columns, to quantify the monosaccharides and the oligosaccharides separately (Brienzo et al., 2010; Mathew, Karlsson *et al.*, 2017; Sabiha-Hanim *et al.*, 2011; Zhang *et al.*, 2017). Dionex[®], the leading manufacturer of HPAEC-PAD, is marketing two major versions of columns for the carbohydrates analysis: CarboPac PA10 and PA20 for the analysis of monosaccharides and CarboPac PA100 and PA200 for the analysis of oligosaccharides, requiring two sets of conditions.

However, previous attempts to perform simultaneous separation of mono- and oligosaccharides using HPAEC technique following a multistep gradient elution approach, where a low concentration of NaOH is used to elute monosaccharides and

followed by sudden increase of the concentration of the mobile phase to elute the oligosaccharides and uronic acid (Anders *et al.*, 2015; Wang *et al.*, 2012). The multistep gradient elution successfully separated the analytes in the presented reports, but it required a previous knowledge of the peaks and their retention time. Also the sudden change in the mobile phase causes the disruption and increased noise of the base line as illustrated in Figure 6-6 which affects the overall peak integration and the method repeatability (Snyder *et al.*, 2012).



Figure 6-6. Separation of 1. arabinose; 2. galactose; 3. glucose; 4. xylose; 5. xylonic acid; 6. gluconic acid; 7. galacturonic acid; 8. glucuronic acid with step gradient method (Wang et al., 2012).

Due to these various technical limitations, the precise activity of xylanases in terms of their activity on oligosaccharides, particularly those from wheat, has not been well described in the literature.

Therefore, this chapter introduces a novel high-resolution HPAEC-PAD method based on the linear gradient elution of both sodium hydroxide and sodium acetate to give a high-resolution method for the simultaneous determination of cereal monosaccharides, xylooligosaccharides, arabinoxylooligosaccharides and uronic acids (glucuronic and galacturonic acid). The new method allows a comprehensive description, quantitative and qualitative, of the modes of action of xylan-degrading enzymes as well as the substrate specificity of the enzymes. The chapter reports the materials and methods used, followed by the description of the method performance by reporting relevant validation parameters. The new method was demonstrated to evaluate the performance of Econase to release XOS from a relatively clean substrate (wheat bran) and to measure the XOS content in some external samples submitted to the University of Huddersfield from industrial and academic partners.

6.3 Materials and methods

6.3.1 Materials

Standards of monosaccharides fucose (as internal standard), arabinose, galactose, glucose and xylose were obtained from Sigma UK. Xylo-oligosaccharides (xylobiose (X2), xylotriose (X3), xylotetraose (X4), xylopentaose (X5) and xylohexaose (X6)) and arabinoxylan oligosaccharides (2^{3} - α -L-arabinofuranosyl-xylotriose (A2XXX), 3^{3} - α -L-arabinofuranosyl-xylotetraose (XA3XXX), 2^{3} - α -L-arabinofuranosyl-xylotetraose (XA2XXX) and 2^{3} , 3^{3} -di- α -L-arabinofuranosyl-xylotriose (A2A3XXX)) were purchased from Megazyme (Bray, Ireland). Glucuronic acid, galacturonic acid, sodium acetate, sodium hydroxide (52%), citric acid and sodium phosphate mono-basic were obtained from Fisher Scientific UK.

Commercial xylanase Econase XT 25L was supplied by AB Vista (Marlborough, UK). Wheat bran was kindly provided by the Biorenewables Development Centre, York, UK.

The chromatography system used was the Dionex ICS-3000 described earlier in Section 4.5.4.

6.4 Chromatographic conditions

Gradient elution of two mobile phases was performed: Solution C comprising 10 mM sodium hydroxide; and Solution D comprising a mixture of 200 mM sodium hydroxide and 125 mM sodium acetate. Figure 6-7 presents the elution conditions applied;

starting with 100% Solution C at time 0, a linear increase of Solution D was applied to reach 100% after thirty minutes, then held at 100% D for five minutes. Column equilibration was performed by maintaining starting conditions (100% C) for at least 15 min before sample injection. The mobile phase was sparged with helium and kept under helium during the analysis. The elution flow rate was maintained at 0.3 mL min⁻¹ and the injection volume was 20 μ L. Pulsed wave form "carbohydrates (standard quad)", wave form description "carbohydrates (std. quad. Potential)" was used for the detection. Fucose was used as an internal standard. Solutions A and B are water and 200 mM sodium hydroxide respectively, and are used for system washing.



Figure 6-7. Developed HPAEC-PAD method condition for the new method.

6.4.1 Preparation of standard solutions

A primary stock solution of each standard was prepared in MilliQ (18.2 M Ω ·cm) water to obtain a concentration of 2 mg mL⁻¹. Then a secondary stock solution of a mixture of all the standards was prepared at concentrations of 100 ppm each. Serial dilutions with water were performed to prepare the calibration curve for each standard. The concentration range for working standard solutions was 1-20 ppm. The standards solution was found to be stable when kept refrigerated at 4°C.

6.4.2 Sample preparation

6.4.2.1 Aberystwyth University samples

Two samples of defatted steam exploded brewers spent grain were received to measure the oligosaccharide yield of the steam explosion pretreatment. The samples were dry, so were solubilised in ultrapure water to yield a concentration of 10 mg mL⁻¹, centrifuged and filtered over 0.45μ m syringe filters.

6.4.2.2 AB Vista samples

AB Vista is part of AB Agri, the agricultural part of Associated British Foods (ABF), and is one of the leading international companies in the field of animal nutrition technologies. Four samples of XOS condensed syrup were submitted by AB Vista to measure the content of oligosaccharides following dilution with water and treatment with xylanases. Samples were diluted, filtered and analysed via HPAEC-PAD using the new method. The samples were labelled as:

- 108-656 (XOS syrup)
- 109-580 (Water + XOS syrup (2:1))
- 109-581 (Xylanase 1 + XOS syrup (2:1))
- 109-582 (Xylanase 2 + XOS syrup (2:1))

6.4.2.3 Wheat bran—Econase treatment

Wheat bran was washed with water at 70°C with continuous stirring for 60 minutes, then strained over a 250 μ m sieve three times to ensure the removal of any residual starch. The washed bran was oven-dried overnight at 75°C.

A sample of 20.0 g of the washed dried bran was weighed into a flask with 200 mL of citrate/ phosphate buffer at pH 4.5 and placed in a shaking water bath at 50°C. One mL of the enzyme was added (equivalent to 50 mL per kg of dry sample), after which 10 mL

samples of the solution were taken after 0, 30, 60 and 120 minutes and submerged in an ice bath to halt the enzyme activity. The samples were diluted and filtered over 0.45 μ m syringe filters and analysed in the HPAEC system. Samples were injected manually directly after removal from the ice bath (thus samples remained cold such that enzyme activity did not resume prior to analysis).

6.5 Results and discussion

6.5.1 Chromatography and specificity

The general concept of chromatographic separation is based on the variation in the distribution of the analysed substances between two dissimilar phases, the stationary and mobile phases. In liquid chromatography, the mixture of analytes is dissolved in the mobile phase and separated by passing through the column which contains the particles of the stationary phase. The separation occurs as each molecule in the sample act in a different way with the stationary phase. Molecules of strong solubility in the mobile phase and less affinity to the stationary phase move faster throughout the column and have lower retention times. Considering the analysis of monosaccharides and oligosaccharides with HPAEC, it is clear that the strength of the mobile phase is the key driver behind the solubility of the sugars in the mobile phase and thereby the different retention times of monosaccharides of similar Mw. The molecular weight and shape also plays an important role as well. The applied gradient elution allows varying interaction with the mobile phase to take place and yields in a gradual separation of monosaccharides followed by the migration of linear XOS according to their molecular weight, then the elution of the branched AXOS and the uronic acids.

Figure 6-8 shows a chromatogram of the standards mixture, each at a concentration of 20 ppm, obtained using the CarboPac PA200 column. The chromatographic run time of 35 minutes was sufficient for the elution of all the standards. The baseline was smooth and stable in all the analysed samples, and the gradient elution gave a high resolution separation and did not affect the baseline.



Figure 6-8. Chromatogram of mono and oligosaccharides mixture using CarboPac PA200.

In fact, the linear gradient elution of the alkaline solvent not only enhanced the separation of oligosaccharides, it also increased the capability of the columns used for the separation. The CarboPac PA20 is commercialised as the column of choice for mono- and disaccharides, but with the applied conditions the PA20 was also able to separate the oligosaccharides efficiently although over a longer time, as illustrated in Figure 6-9. The PA200 was used for the remainder of the current work as it allowed faster run times.



Figure 6-9. Chromatogram of mono and oligosaccharides mixture using CarboPac PA20.

The validation of analytical methods addresses the objective of obtaining consistent, reliable and accurate measurements. The validation procedure followed in this

research complies with the International Conference of Harmonisation (ICH) guidelines on method validation (ICH Q2(R1), 1994).

6.5.2 Linearity

To assess linearity, calibration curves were plotted using the ratio of the analyte peak area to the internal standard peak area against concentrations of different dilutions (n = 9) covering the range 1-100 ppm for monosaccharides and 1-20 ppm for XOS, AXOS and uronic acids. The correlation coefficient values were acceptable for all the standards and ranged between 0.985 and 0.999. The correlation coefficients (R²), slopes and intercepts are reported in Table 6-2, along with Limits of Detection (LOD) and Quantitation (LOQ).

	R2	Intercept	Slope	LOD (ppm)	LOQ (ppm)
Arabinose	0.999	0.039	0.154	0.03	0.08
Galactose	0.996	-0.176	0.132	0.08	0.1
Glucose	0.996	-0.231	0.149	0.08	0.1
Xylose	0.997	0.087	0.213	0.04	0.1
Xylobiose	0.984	0.226	0.079	0.001	0.008
Xylotriose	0.983	0.221	0.075	0.005	0.01
Xylotetraose	0.981	0.202	0.056	0.2	0.4
Xylopentaose	0.985	0.112	0.041	0.3	0.5
Xylohexaose	0.986	0.12	0.039	0.03	0.1
A3XXX	0.994	0.073	0.049	0.04	0.08
XA2XXX	0.993	0.057	0.045	0.01	0.03
XA3XXX	0.990	0.081	0.043	0.01	0.03
Gal A	0.984	-0.031	0.033	0.06	0.2
Glu A	0.997	-0.021	0.056	0.06	0.1
A2A3XXX	0.995	0.047	0.035	0.08	0.2

Table 6-2. Method linearity, LOD and LOQ for each analyte.

6.5.3 Limits of detection (LOD) and quantitation (LOQ)

LOD and LOQ measure analytical method sensitivity to low concentrations of analyte. LOD is the lowest concentration detected and LOQ is the minimum quantifiable concentration measured by the method. Signal-to-noise (S/N) ratios of 3:1 and 10:1 were used to define LOD and LOQ, respectively, and were calculated by the Chromeleon[®] software. The peaks were clear to identify after the dilutions and had acceptable precision. LOD and LOQ values for all the analytes are listed in Table 6-2. The table shows variation in the detection and quantification limits of the standards, which is a reflection of the variation in the peak heights and areas for the same concentrations, as evident in Figure 6-8. In general, the monosaccharides showed lower detectable concentrations than the oligosaccharides, except for xylobiose which is more readily detected than xylose and gave the lowest values of LOD and LOQ.

6.5.4 Accuracy and precision

The accuracy of the method was studied by analysing triplicates of control samples covering three concentration levels (5, 10, 15 ppm); the results are listed in Table 6-3 as percentage recovery. The calculated recoveries for all the studied standards ranged acceptably between 97-103%.

The precision of the method reflects the agreement among individual test results. Intraday precision (repeatability) was estimated by analysing seven replicates of a control sample in the same HPAEC run. Inter-day precision (intermediate precision) was assessed from nine triplicates over three consecutive days (three runs per day) using the same instrument with three different batches of the mobile phase. Both repeatability and intermediate precision results are listed in Table 6-3 as the percentage coefficient of variation (CV%). Both the precision and the intermediate precision are in acceptable ranges for all the analytes.

		Arabinose	Galactose	Glucose	Xylose	Xylobiose	Xylotriose	Xylotetraose	Xylopentaose	Xylohexaose	АЗХХХ	XA2XXX	ХАЗХХХ	A2A3XXX	Gal A	Glu A
%	15 ppm	101.02 ±0.64	100.46 ±1.46	100.28 ±0.82	100.42 ±1.2	100.81 ±0.78	99.99 ±1.23	100.34 ±0.31	100.4 ±0.86	102.67 ±3.2	102.13 ±3.33	102.07 ±2.42	100.01 ±1.12	100.12 ±0.6	100.4 ±1.02	100.26 ±0.41
ecovery 9	10 ppm	101.21 ±2.52	100.91 ±1.22	102.38 ±3.42	99.89 ±2.73	101.23 ±1.93	103.29 ±4.63	99.68 ±1.08	101.14 ±2.4	100.59 ±0.64	99.41 ±1.1	101.55 ±1.41	99.97 ±0.9	100.26 ±2.25	99.81 ±2.1	101.2 ±1.15
~	5 ppm	100.19 ±3.03	100.5 ±3.07	101.95 ±2.87	100.09 ±0.87	97.70 ±2.83	103.91 ±4.61	100.20 ±1.88	99.60 ±1.76	100.08 ±0.33	99.57 ±0.51	101.33 ±3.17	100.79 ±2.97	100.07 ±1.83	99.55 ±0.92	100.74 ±2.18
Prec	ision CV%	0.006	0.015	0.008	0.012	0.008	0.012	0.003	0.009	0.031	0.033	0.024	0.011	0.006	0.01	0.004
ion CV%	Day 1	0.03	0.017	0.047	0.005	0.014	0.026	0.011	0.002	0.05	0.049	0.018	0.018	0.017	0.019	0.017
Intermediate precis	Day 2	0.015	0.014	0.012	0.025	0.01	0.024	0.017	0.015	0.041	0.022	0.004	0.02	0.014	0.016	0.003
	Day 3	0.006	0.015	0.008	0.012	0.008	0.012	0.003	0.009	0.031	0.033	0.024	0.011	0.006	0.01	0.004

 Table 6-3. Method accuracy (reported as percentage recovery) and precision (reported as CV%).

6.5.5 System suitability tests

The fitness of the chromatography system was estimated by a range of system suitability test parameters: Capacity factor (or retention factor, k'), Resolution factor (R_s), number of theoretical plates (Column efficiency) and Asymmetry (A_s), as calculated by the Chromeleon[®] software. The results of these parameters for each analyte are listed in Table 6-4. The values of the resolution factor reflect high-resolution separation of the analytes except for galactose where the R_s was only 1.71. In general, all the calculated parameters gave acceptable values; the asymmetry factor was between 0.95 and 1.37, the capacity factor was more than 2 and the number of theoretical plates was greater than 15000 for all the analysed standards.

	Resolution	Asymmetry	Canacity	theoretical
	factor R _s	As	factor k'	plates
Arabinose	3.63	1.35	2.676	17620
Galactose	1.71	1.10	3.114	15701
Glucose	2.64	1.08	3.334	18981
Xylose	10.23	1.26	3.649	26455
Xylobiose	26.6	1.10	5.851	1081600
Xylotriose	9.46	1.23	6.132	112710
Xylotetraose	9.15	1.33	7.071	80127
Xylopentaose	8.25	1.09	8.228	69836
Xylohexaose	13.27	1.04	9.500	61511
A3XXX	3.91	1.10	11.737	91312
XA2XX	3.13	0.95	12.430	82712
XA3XX	3.69	0.95	13.026	82668
Gal A	3.06	1.07	13.684	130671
Glu A	7.47	1.37	14.035	793131
A2A3XXX	23.52	1.05	15.298	57549

Table 6-4. Method resolution, asymmetry factor, capacity factor and number of theoreticalplates.

6.5.6 Sample analysis results

This subject area is of active interest to other industrial and academic parties who, knowing of our development of this new analytical method, were keen to send us samples for analysis. This section illustrates the new method and the sorts of samples for which it is relevant, by reporting results for samples sent from Aberystwyth University and AB Vista. The new method was also used to detect the results of Econase treatment of wheat bran.

6.5.6.1 Aberystwyth University samples results

The chromatograms from the two spent grain samples from Aberystwyth University are shown in Figure 6-10. The chromatograms present clearly the complex nature of the samples with the presence of unidentified peaks. However, they also show that the targeted XOS were separated efficiently and could be identified. The samples were also spiked with the standards solution and reanalysed, to confirm that the peaks were identified correctly.



Figure 6-10. Chromatographs of Aberystwyth University spent grain samples.

Table 6-5 reports the percentage content of the analysed mono- and oligosaccharides in the analysed samples. Clearly, the content of XOS in the samples was almost negligible and the pretreatment with steam explosion was not sufficient to produce significant amounts of the desired XOS.

	Concentration of the sample (%)									
	Ara	Glu	Xylose	X2	Х3	X4	X5	X6		
S1	0.04	0.01	0.02	0.05	0.05	0.05	0.05	0.08		
S2	0.03	0.01	0.01	0.01	0.01	0.01	0.05	0.03		

Table 6-5. Concentration of XOS in Aberystwyth University spent grain samples.

6.5.6.2 AB Vista samples

Figure 6-11 shows the chromatogram of the first of the four samples. Clearly, there was a significant amount of XOS in the syrup, particularly X2 and X3.



Figure 6-11. AB Vista (108-656 XOS syrup) sample chromatogram.

Figure 6-12 reports the amounts of each mono- and oligosaccharide in the four samples. In the original XOS syrup (Sample 108-656), xylose (X1) and xylobiose (X2) dominate, followed by X3, with significant amounts of X4-X6 as well as arabinose, galactose and glucose. Dilution reduced the concentration of all components by a factor of three (Sample 109-580). Both enzymes then increased dramatically the concentrations of X1 and X2, apparently by acting on X3, X5, X6 and presumably larger oligos not measured (the increase in X1 and X2 is more than can be accounted for by the decreases in X3, X5 and X6) but, interestingly, appearing to leave X4 largely untouched. The apparent lack of change in X4 is probably an illusion arising from the dynamic balance between the production of X4 from larger oligos (X5, X6 and beyond) and its degradation; probably largely by coincidence, under the conditions used, X4 happened to be produced as fast as it was degraded, while for the other oligos these were not balanced. This result demonstrates how fully understanding the enzymatic production of oligosaccharides requires population balance modelling of the evolution of the oligosaccharide profile, a theme picked up in the next chapter. For now, the results demonstrate that the new method can be applied to characterise commercial materials containing XOS and to determine the effects of enzymes on these materials, in order to maximise XOS production. It is also noteworthy that the enzymes produced significant amounts of the xylose monosaccharide, which is not expected from an endo-xylanase.



Figure 6-12. XOS syrup enzyme treatment samples.

6.5.6.3 Wheat bran—Econase treatment

The oligosaccharides released from wheat bran by xylanase treatment were measured and estimated as a percentage of the original wheat bran on a weight basis. Figure 6-13 shows the release of each analyte over time for up to 180 minutes. (For comparison, the release in buffer solution without enzyme after 180 minutes is also shown; significant extraction of endogenous glucose and xylose into buffer occurred without enzyme, but this was not the case for the oligosaccharides or GluA and GalA, indicating these were not present endogenously in the original wheat bran in detectable amounts.) Clearly both glucose and xylose increased over time, the former suggesting some amylolytic side activity or contamination in the enzyme. Xylobiose (X2) also increased monotonically, but by contrast, X3 and X4, after an initial burst of production, decreased over time as they were further degraded by the xylanase. Small amounts of X5 and X6 also appeared briefly but were degraded even more quickly. Significant amounts of GalA and GluA were also released, to levels that appeared to become stable over time.



Figure 6-13. Mono- and oligosaccharides released from wheat bran by xylanase treatment for 180 minute at 50°C.

Figure 6-14 illustrates more clearly the dynamics by showing the evolution of X1-6 over time. The results indicate a pattern of the creation and degradation of the xylan oligosaccharides as initially fragments are released by enzyme action on larger AX polysaccharides, then the enzymes turn their action onto the released oligos. Thus the majority of the oligos were produced in the first 30-60 minutes, including the brief appearance of X6 and X5 during this time. Thereafter, a balance of producing new small oligos (X2) and degrading the larger oligos (X3, X4, X5 and X6) becomes evident. The steady increase of X2 during the treatment suggests that this endoxylanase cannot act on the disaccharide. On the other hand, the increasing amounts of xylose, which is not expected for endoxylanases (Araki and Kitamikado, 1982; McCleary *et al.*, 1982) (as they attack the middle of xylan chains), could be explained by three hypotheses: endoxylanases could have a side activity as exoxylanases; there could be some contaminating exoxylanase in the enzyme preparation; or (least likely for steric hindrance reasons), the xylose production could be the result of two enzyme molecules attacking the same xylan chain one unit apart at the same time.



Figure 6-14. Evolution of XOS from wheat bran over time following xylanase treatment.

These results confirm the efficacy and usefulness of this new method for analysing samples containing XOS, and for understanding the evolving profile as enzymes create and then destroy oligosaccharides. These preliminary trials were on an arbitrary selection of industrially relevant samples, to demonstrate the method and to begin to gain insights into the patterns of enzyme behaviour. Having done so, it becomes clear that there is a need for more fundamental and controlled studies to understand the behaviour of xylanases on individual XOS molecules.

6.6 Summary

The analysis of xylan and arabinoxylan oligosaccharides has been a major obstacle in the path to develop and study these products. High performance anion exchange chromatography with pulsed amperometric detection (HPAEC-PAD) has proved to be well suited for the separation and analysis of XOS. Previously reported methods showed good performance in characterising the XOS with DP2-6. However, the resolution of the separation has made it difficult to apply these methods to real samples such as food and animal feed formulations. Also, the need to perform a separate analysis for monosaccharides increases the time and effort to identify the full carbohydrate profile of a sample. Moreover, the measurement of uronic acids in the samples, which are a major component of wheat AX, required a third run with different conditions as well.

Therefore, this part of the work has presented a novel and reproducible HPAEC-PAD method for the simultaneous separation of cereal monosaccharides, xylan oligosaccharides, arabinoxylan oligosaccharides and uronic acids. The method is based on gradient elution of the solvents, which was shown to give excellent separation of xylose, arabinose and glucose as well as the xylo-oligosaccharides in the range X2-X6 plus galacturonic and glucuronic acids. The new method is applicable for testing and developing xylan-degrading enzymes and for measuring their production of mono- and oligosaccharides in food and feedstuffs.

The developed method was applied to measure the XOS content of some commercial and research samples from collaborators, and also allowed the precise observation of Econase action on wheat bran. The preliminary investigation raised vital questions about the mode of action of this endoxylanase towards the production and degradation of XOS. As the new method can give a clear picture of the mono- and oligosaccharide profile, the next step of this project aimed to observe the action of Econase on XOS pure standards, to clarify the precise behaviour of this enzyme on these small oligosaccharides.

7 Endoxylanases degradation of XOS standards

7.1 Introduction

The possibility of enzymatic production of xylan and arabinoxylan oligosaccharides has been reported extensively in the literature; this has mostly been addressed by the employment of xylanases. The common understanding of the activity of xylanases divides them into two main categories: exo-xylanases (β -xylanases), which possess the ability to attack the terminal units in the xylan chain and produce xylose monomers; and endo-xylanases, which attack the middle of the xylan polysaccharide to produce smaller polymer molecules such as oligomers.

In the experiments reported in Chapter 6, it was observed that significant amounts of xylose monomers were present after the utilisation of the commercial endoxylanase Econase. Such presence of monosaccharides was not expected to accompany the production of XOS/AXOS. This unexpected result underlines that the role of endoxylanases in degrading oligosaccharides is not well established. Having developed the new method for the simultaneous determination of mono- and oligosaccharides, and with the recent availability of high purity AXOS standards, it became possible to investigate and explore the precise effect of xylanases on the evolution of the oligosaccharide profile. This chapter presents a brief review of the origin, classification and usages of xylanases. It then describes experimental work to treat XOS standards with Econase and follow the evolving production and degradation of the profiles of xylo-oligosaccharides alongside the xylose monomer.

7.2 Xylanases – a review of their features and applications

In the nineteenth century, Schulze (1891) isolated a fraction of plant material using dilute alkaline solution, which was made of sugars but not cellulose, which he named "hemicellulose". Hemicelluloses comprise a complex family of structures that can be formed from a variety of polysaccharides such as glucuronoxylan, arabinoxylan, glucomannans, arabinogalactan, and galactoglucomannans (Kulkarni et al., 1999; Subramaniyan and Prema, 2002). They contribute to the structural integrity of plant cell walls and are thus closely associated with the other structural components, cellulose and lignin. The distribution of these polysaccharides varies according to the plant species; unlike cellulose, hemicellulose is not a defined chemical compound or material, but more like a class of plant polysaccharides with a widely variable set of properties peculiar to the constituent polysaccharides. Courtin and Delcour (2002) described cereal hemicelluloses, *i.e.* arabinoxylans, as a "peculiar group of molecules", reflecting their strange diversity of features and behaviours.) The most common polysaccharide in hemicelluloses is Xylan, which can be degraded by xylanolytic enzymes (Bajpai, 2014b). Hemicellulose degradation was first reported by Hoppe-Seyler (1889, cited in Sharma, 2013) who described the degradation of xylan wood slurry by microbes from river mud. As hemicelluloses are made up of 5-carbon sugars, they have been known as "pentosans", and in 1955, xylan-degrading enzymes were reported as pentosanases (Whistler and Masak, 1955; Collins et al., 2005). Thereafter, they were recognised by the International Union of Biochemistry and Molecular Biology (IUBMB) in 1961 and were given the enzyme code (EC 3.2.1.8.) (Collins *et al.*, 2005).

Naturally, xylanases are as diverse as the composition and structure of xylan polysaccharides; they can be produced by both plants and microorganisms to serve various purposes. In the plant kingdom, xylanases have been found in cereal grains where they help in the remodelling and the expansion of cell wall during seed growth and also for the cell wall degradation during seed germination (Dornez *et al.*, 2009). Xylanases have been found in wheat grain (Preece and MacDougall, 1958; Kulp, 1968), wheat bran (Bergmans *et al.*, 1996), wheat flour (Cleemput *et al.*, 1997), barley (Taiz and Honigman, 1976) and in germinated barley seeds (Slade *et al.*, 1989). Xylanases

expressed by microorganisms (bacteria, fungi, actinomycetes and yeast) serve a wide range of purposes: some organisms excrete a number of hydrolysing enzymes including xylanases in order to provide a survival carbon source (Prade, 1995), whereas other organisms produce xylanases as a part of infecting and colonising plant cells (Hrmová *et al.*, 1984; Sunna and Antranikian, 1997; Bajpai, 2014a). In a broader context, xylanases have also been reported in marine algae, protozoans, crustaceans, insects and snails (Sunna and Antranikian, 1997).

7.3 Classification of xylanases

The number of the identified xylanases had been increasing due to the heterogeneity of xylan polysaccharides in nature and the tremendous number of xylanase producing microorganisms, considering that many organisms can produce several xylanases (Gilbert and Hazlewood, 1993; Collins *et al.*, 2005). Xylanases are thus known for their wide diversity, with overlapping and complex physicochemical properties, structures, specific activities and yields. Several classifications have been attempted for these enzymes. Wong *et al.* (1988) classified xylanases according to their physicochemical properties and categorised them into two groups: small molecular weight (<30 kDa) and basic isoelectric point, and large molecular weight (>30 kDa) and acidic isoelectric point. Nearly 30% of the characterised xylanases do not fit into this classification, particularly the fungal xylanases (Matte and Forsberg, 1992; Sunna and Antranikian, 1997).

A broader and more comprehensive classification was constructed for all the glycosidase enzymes (EC 3.2.1.x.) by the International Union of Biochemistry and Molecular Biology (IUBMB) (Henrissat *et al.*, 1989). This new classification has become the standard classifying system for all the hydrolysing enzymes; its basis, according to the carbohydrate active enzyme (CAZy) website (<u>http://www.cazy.org/Glycoside-Hydrolases.html</u>), is:

"Glycoside hydrolases (EC 3.2.1.x) are a widespread group of enzymes which hydrolyse the glycosidic bond between two or more carbohydrates or between a carbohydrate and a non-carbohydrate moiety. The IUBMB Enzyme nomenclature of glycoside hydrolases is based on their substrate specificity and occasionally on their molecular mechanism."

The IUBMB classification relies mainly on the substrate specificity, indicated in the numbering system (EC 3.2.1.x) in which x defines the substrate; for example, β -glucosidase is referred to as (EC 3.2.1.21), and β -galactosidases are (EC 3.2.1.23).

It was recognised that a varied group of enzymes collectively degrade xylan in nature; they were therefore classified under several Enzyme Commission numbers (EC): endo-xylanase (endo-1, 4- β -xylanase E.C.3.2.1.8), β -xylosidase (1, 4- β -xylosidase E.C.3.2.1.37), α -arabinofuranosidase (α -L-arabinofuranosidase, E.C.3.2.1.55), acetylxylan esterase (E.C.3.1.1.72) and feruloyl esterase (EC 3.1.1.73).

Exo-xylanases or β -xylosidase (1,4- β -xylosidase E.C.3.2.1.37) have the ability to hydrolyse xylan and xylo-oligosaccharides, releasing xylose residues by attacking the terminal units of the chain (Araki and Kitamikado, 1982; McCleary *et al.*, 1982), as Figure 7-1 illustrates. By contrast, endoxylanases or endo- β -(1,4)-D-xylanohydrolase, (E.C. 3.2.1.8) hydrolyse xylan to β -D-xylopyranosyl oligomers by attacking the glycosidic bonds between xylose units in the xylan backbone resulting in a general reduction in the degree of polymerisation (Mandal, 2015).



Figure 7-1. AX hydrolysing enzymes site of attack (Dornez et al., 2009).

Under this classification, endoxylanases from different origins are divided into several glycoside hydrolase families (GH) (5, 7, 8, 9, 10, 11, 12, 16, 26, 30, 43, 44, 51 and 62) (Moreira and Filho, 2016; Uday *et al.*, 2016). Among these families, GH10 and GH11 are well-studied in the literature, and present differences in the substrate susceptibility. It is believed that the GH10 family has the ability to hydrolyse substituted xylan chains and thereby produces substituted or branched oligosaccharides. By contrast, xylanases from the GH11 family, customarily known as "the true xylanases", are strictly able to attack only the xylan backbone and perform only on the unsubstituted regions of the polysaccharide (Collins *et al.*, 2002; Imjongjairak *et al.*, 2015). Studies on the modes of action of β -xylosidases and endoxylanases revealed that the endoxylanases have no effect on small XOS of DP2-3, whereas β -xylosidases can hydrolyse them (Deshpande *et al.*, 1986).

7.4 Industrial Applications of Xylanases

In the last few decades, enzymes have witnessed dramatic growth, particularly in the field of biotechnology. The enzyme global market share reached one billion dollars in 1990 and two billion in 2005, the market share climbed to nearly \$4.6 billion and \$4.9 billion in 2014 and 2015 respectively and is expected to soar upwards to \$6.3 billion in 2021 according to bccResearch (Anon, 2017b). Polizeli *et al.* (2005) reported that xylanases, cellulases and pectinases collectively account for 20% of the world enzymes market.

Xylanases have been exploited in several industrial fields, of which paper and pulp bleaching is the largest market so far. In this context, cellulose is the main component of paper, coming primarily from the wood. In order to remove the lignin component from the woody starting material, several chemical approaches are used incorporating chlorine, hydrogen peroxide and oxygen. However, the environmental effect of this industry was the primary precursor of searching for other cleaning methods; since xylan plays the role of the connector between the undesirable lignin and the desirable cellulose, the enzyme-mediated degradation of this linkage has become of interest as it is low cost and eco-friendly (Amin, 2006; Bajpai, 1999; Paice *et al.*, 1992; Viikari *et al.*, 1994). Xylanase treatment in paper-making is usually called pre-bleaching or bleach boost, as the enzyme does not affect the chromophores of the lignin, but it degrades the xylan network surrounding and protecting the lignin. This effect makes lignin more vulnerable to chemical bleaching and reduces the amounts of chemical bleachers used in the process (Saleem *et al.*, 2009; Ko *et al.*, 2011; Walia *et al.*, 2017).

Xylanases are also widely used in the food industry, particularly bread making, along with other enzymes such as α -amylase, malting amylase, glucose oxidase and proteases (Butt et al., 2008; Harris and Ramalingam, 2010). Bread making is the most established xylanase application in the food world; the hydrolysis of non-starch polysaccharides seems to improve the rheological behaviour of bread dough and yield a general improvement in bread specific volume, crumb firmness, loaf volume and crumb structure (Baillet et al., 2003; Guy and Sarabjit, 2003; Shah et al., 2006). Xylanases also decrease starch retrogradation, reducing the initial crumb firmness during storage and having as anti-staling effect (Haros et al., 2002; Jiang and Tan et al., 2005; Jiang and Li et al., 2005). In biscuit-making, xylanases help to improve wafer palatability, texture and uniformity (Polizeli et al., 2005). Xylanases, with other glycosidic hydrolases, are applied in fruit and vegetable juice making as they increase both the yield and quality of the juice by facilitating the liquefaction process and increasing the recovery of essential oils, vitamins, mineral salts and edible dyes (Bajpai, 2014a; Polizeli et al., 2005). Xylanases in the brewing industry improve the filterability of the mash by reducing the viscosity (Sadosky *et al.,* 2002).

The recent utilisation of xylanases for the production of xylan prebiotics is a promising and exciting field. The increasing amounts of lignocellulosic biomasses produced in several industrial domains, along with the insatiable need of the food market for novel and healthy products, have developed a great opportunity for the production of xylooligosaccharides.

The most convenient method for XOS production is the partial breakdown of hemicellulosic xylan chains into smaller chains. Interestingly, the conversion of hemicellulose into valuable XOS material has received a great deal of attention in the last decade. Several reports suggested hydrothermal (Vegas *et al.*, 2005; Ho *et al.*, 2014; Rose and Inglett, 2010; Buruiana *et al.*, 2017; Surek and Buyukkileci, 2017) and enzymatic treatments for the xylan degradation in different biomasses (Rydlund and Dahlman, 1997; Swennen *et al.*, 2006; Escarnot *et al.*, 2012; Falck *et al.*, 2013; Falck *et al.*, 2018). Table 7-1 illustrates the most relevant papers describing XOS production by enzymatic means, utilising xylanases from different origins to produce XOS from a variety of substrates including cereals, sugarcane bagasse, maize and tobacco stalks. The yields varied according to the substrate and the enzymes applied and the pretreatment method.

The production of XOS and AXOS could be achieved in several ways; the most direct method is utilising xylanases to hydrolyse the xylan chain in biomass, but although this method is cost effective, the produced yields are very low. Alternatively, chemical extraction of AX followed by enzyme treatments can give a higher yield. Utilising harsher treatments such as steam and hydrothermal processes could significantly increase the yield. However, the cost of the process and the simultaneous production of undesired monosaccharides and other chemicals, such as furfural, promotes the use of xylanases as the optimum route (Bajpai, 2014a; Rose and Inglett, 2010).

Biomass	Pre-treatment	Enzymes	Yields of XOS from original material*	Reference
Sugarcane bagasse	Aqueous ammonia	Bacterial arabinofuranosidase + β-xylosidase	*	Reddy and Krishnan, 2016
Sugarcane bagasse	Alkaline extraction	Bacterial cellulase + endoxylanase	6%	Xue <i>et al.,</i> 2016
Wheat bran	Destarching	Fungal endoxylanase + Feruloyl esterase	**	Wu <i>et al.,</i> 2017
Wheat chaff	Ultrasound	Endoxylanase	8% (+2-3% xylose)	Antov and Đorđević, 2017
Wheat straw	Alkaline extraction	Bacterial endoxylanase	11%	Faryar <i>et al.,</i> 2015
Wheat straw	Steam explosion	Commercial endoxylanase + glucosidase	8.9%	Álvarez <i>et</i> <i>al.,</i> 2017
Birchwood	-	Fungal endoxylanase GH11	28.8%	Nieto- Domínguez <i>et al.,</i> 2017
Birchwood, Rye	-	Bacterial xylanase	20% birchwood 3.3% rye	Falck <i>et al.,</i> 2013
Kenaf stems (<i>Hibiscus</i> <i>cannabinus</i>)	Ca(OH) ₂ followed by peracetic acid	Endoxylanase and arabinofuranosidase	35.2%	Wan Azelee <i>et al.,</i> 2016
Corncob	Alkaline extraction	Bacterial endoxylanase	**	Gowdhaman and Ponnusami, 2015
Corncob	Alkaline extraction	Fungal β-xylosidase	10.7%	Chapla <i>et</i> <i>al.,</i> 2012
Corncob	Acidic hydrolysis	Cellulase	14% (+ 32.8% glucose, 14.8% xylose)	Zhang <i>et al.,</i> 2017
Corncob	Hydrothermal + 16% alkali	Commercial endoxylanase	35% AX extract containing 9% XOS	Samanta <i>et</i> al., 2015
Corncob	Alkali/acid pressure cooking	Fungal endoxylanase	**	Aachary and Prapulla, 2009
Tobacco stalk	Alkaline extraction	Commercial endoxylanase	11%	Akpinar <i>et</i> <i>al.,</i> 2010

	Table 7-1.	Methods	reported	for XOS	production.
--	------------	---------	----------	---------	-------------

* Yields are measured and reported in a variety of ways and are indicative rather than directly comparable.

** Insufficient data

Studies of enzymatic XOS production have only occasionally reported the accompanying release of monosaccharides. Uçkun Kiran *et al.* (2013) reported that two commercial endoxylanases released different amounts of undesired xylose monosaccharide in the extracts. Antov and Đorđević (2017) advised that the endoxylanase treatment time is crucial to minimise the production of undesired monosaccharides.

Generally, the activity of endoxylanases on oligosaccharides is not well understood. For example, only a limited number of contradictory reports in the literature discuss whether endoxylanases produce monosaccharides or not (Antov and Đorđević, 2017; Cürten *et al.*, 2018; Uçkun Kiran *et al.*, 2013). Also, the action or preference of endoxylanases for larger or smaller oligosaccharides chains has not previously been reported, as the interest in the hydrolysing enzymes has been primarily their efficacy in breaking down the polysaccharides either all the way to the constituent monosaccharides (as a fermentation feedstock, for example) or just to reduce viscosity. Now, with the increasing interest in oligosaccharides arising from their prebiotic functionality, it has become essential to investigate in detail the mode of action of xylanases on the production and degradation of small xylo- and arabinoxylo-oligosaccharides. Such investigations have faced the difficult technical challenges of measuring these oligosaccharides; however the new HPAEC-PAD method developed in the last chapter now makes such studies feasible.

Therefore, the aim of this part of the project was to study and observe Econase activity on the small oligosaccharides, using XOS standards, to identify the substrate preference and the end products of the enzyme, aiming to increase our understanding of xylanase behaviour which will allow better deployment of this and other endoxylanases to produce XOS.

7.5 Materials and methods

Arabinoxylan oligosaccharide standards were obtained from Megazyme as described in Section 6.3.1. For the analysis of oligosaccharides following treatment with Econase, the newly developed method described in Chapter 6 was used. Solutions containing the XOS standards were incubated with Econase at a dose of 50 mL per kg of dry sample, at pH 5.5 phosphate/citrate buffer in a water bath at 50°C. (Note that this dosage is 500× the recommended commercial dose, as the purpose of current commercial usage is based on viscosity reduction, which can be achieved by low dosages, whereas the current study aimed to accelerate XOS production which was expected to require higher enzyme dosages; earlier studies with the IPOS DPn method had shown undetectable effects at lower dosages when applied to wheat bran.) Samples were taken during the hydrolysis to evaluate the XOS production and degradation over time. The samples were diluted using phosphate-citrate buffer (at pH 7) and boiled for 20 minutes to deactivate the enzyme permanently (to allow queuing in the autosampler without further enzyme activity), then kept at room temperature until analysis.

7.6 Results and discussion

Figure 7-2 shows the breakdown of xylohexaose by Econase and the evolution of smaller oligos over time. Clearly, the X6 is rapidly degraded by the Econase at this high dosage. The predominant product is the X3, as expected as cleavage in the middle of the X6 chain would result in two X3 molecules, while cleavage elsewhere would give X2+X4 (a single molecule of each) or X1+X5 (although terminal cleavage to produce a monomer is not expected from an endoxylanases). Thus less X4 than X3 was produced at the beginning of the treatment, which soon were subject to further breakdown into xylobiose units. The presence of the X5, in minor amounts in the first 10 minutes, suggests that the enzyme could attack the terminal units, while the increasing amounts of the monosaccharide, xylose, indicate the unexpected breakdown of the small oligos (X3 and X2) into monosaccharides.



Figure 7-2. Breakdown of xylohexaose upon Econase treatment.

The breakdown of xylopentaose by Econase shows a similar behaviour of attacking the middle of the chain, resulting in production of X2 and X3, as shown in Figure 7-3. The presence of increasing – but small – amounts of monosaccharides again suggests breakdown of the small oligos into xylose. Terminal unit attack appeared to be small as only small amounts of X4 appear along the treatment. The rate of the breakdown of X5 was slower than the breakdown of X6, as 90% of X5 was degraded in the first hour, whilst the whole amount of X6 was degraded after only 20 minutes of treatment.



Figure 7-3. Breakdown of xylopentaose upon Econase treatment.

The difference in the degradation speeds of X6 and X5 indicates a preference of the enzyme for the longer chain oligos. This preference was shown more clearly when a mixture of X6 and X5, each at 100 ppm, was treated with Econase at double the dosage, as presented in Figure 7-4. Again, degradation of X6 was complete within 20 minutes, whereas the X5 again took 120 minutes with little change in the curve shape, which suggests that the presence of the longer X6 chain did not slow the degradation of the X5 (the overall enzyme:substrate ratio staying constant). It can be concluded that possibly the rate of the reaction is the main driver behind the enzyme preferences.



Figure 7-4. Econase degradation of a mixture of xylohexaose and xylopentaose.

Figure 7-5 reports the degradation of X4 upon treating with Econase. In this case, the main degradation product is X2 with the presence of very small amounts of X3 and corresponding X1 monosaccharides. However, the relative quantities of the monosaccharides and xylotriose does not support that the presence of xylose is a result of breaking X4 into X3 and X1 only, which ought to result in almost three times as much X3 as X1 on a mass basis; the greater mass of X1 implies some degradation of X2. Again, this is unexpected for an endo-xylanase; as noted in the previous chapter, this may indicate some side activity or, more likely, indicates the presence of a contaminating exo-xylanase in the Econase preparation.



Figure 7-5. Breakdown of xylotetraose upon Econase treatment.

Figure 7-6 and Figure 7-7 show the effect of Econase treatment on the smaller oligosaccharides X3 and X2, showing that Econase can perform very slowly on the small oligos, and again confirming the production of the monosaccharide.


Figure 7-6. Breakdown of xylotriose upon Econase treatment.



Figure 7-7. Breakdown of xylobiose upon Econase treatment.

These results are in accordance with the findings of Biely *et al.* (1981) who found the Bond-cleavage frequency of xylotriose, xylotetraose and xylopentaose to be

concentration dependent. However, these results differ from those of Mitsuishi *et al.* (1988) who studied three endoxylanases and reported that the enzymes were able to hydrolyse the oligos of X8-X3, but none of the enzymes was able to hydrolyse xylobiose. However, both papers report the dependency of the endoxylanase hydrolysing activity upon the concentration of the substrate. In other words, it might be more plausible to correlate the activity of the enzyme to the substrate-enzyme ratio, or to the enzyme dosage. Figure 7-8 shows the effect of Econase on xylobiose with 20 times the dose used earlier. Surprisingly, the enzyme was capable of degrading almost 99% of the xylobiose substrate, which agrees with Biely *et al.*'s (1981) observation in relation to dose effects, and confirms that, at high dose and in the absence of a preferable substrate, the commercial Econase preparation is able to cleave xylobiose to the monomer.



Figure 7-8. Breakdown of xylobiose upon severe Econase treatment.

The conflicting findings and reports in the literature are a reflection of the immensity of the xylanases world and the limited knowledge of their behaviour on small oligosaccharides. For the studied enzyme, Econase, it can be concluded that it is able to hydrolyse linear oligosaccharides of degrees of polymerisation in the range DP6-DP2, and that the rate of the hydrolysis is greater for the larger molecules. Figure 7-9 compares the rates of degradation of the studied oligos, showing the much faster degradation of the larger chains.





Figure 7-12 shows the treatment of 3^3 - α -L-arabinofuranosyl-xylotetraose and 2^3 - α -Larabinofuranosyl-xylotetraose with Econase, both are illustrated in Figure 7-10. The experiment showed that Econase activity was different in the presence of the arabinose side unit. Apparently, the 3^3 - α -L-arabinofuranosyl-xylotetraose molecule was not susceptible to Econase activity during the treatment, whereas Econase slightly attacked the 2^3 - α -L-arabinofuranosyl-xylotetraose molecule and chopped off the terminal xylose unit producing 2^3 - α -L-Arabinofuranosyl-xylotriose, shown in Figure 7-11.



Figure 7-10. Structure of 3^3 - α -L-arabinofuranosyl-xylotetraose and 2^3 - α -L-arabinofuranosylxylotetraose.



Figure 7-11. Structure of 2^3 - α -L-Arabinofuranosyl-xylotriose.



Figure 7-12. Treatment of 3^3 - α -L-arabinofuranosyl-xylotetraose and 2^3 - α -L-arabinofuranosylxylotetraose with Econase.

Clearly, Econase affinity towards the substituted oligosaccharide varied according to the O-3 and O-2 α -L-arabinofuranosyl groups. In either case, the middle of the tetra-xylan chain was protected by the arabinose side unit, and the enzyme was unable to break it down into xylobiose. However, in the case of O-2 α -L-arabinofuranosyl group, it seems that the steric hindrance of the side chain could not protect the terminal xylose unit and a "slow" or weak hydrolysis occurred, yielding tiny amounts of xylose and a smaller oligosaccharide.

7.7 Summary

Xylanases are a widely varying group of enzymes that hydrolyse the xylan chains in hemicelluloses. There is an unlimited number of sources of xylanases including plants, insects, fungi and bacteria and thereby a huge variation in their nature, characteristics and mode of action. It can be said that the only thing in common among xylanases is the fact that they hydrolyse xylan polysaccharides. Xylanases are widely used in industry, with paper and pulp bleaching the largest market, and they are also widely used in the food industry, particularly as an ingredient in breadmaking. Xylanases now offer a promising new market for the production of oligosaccharides with prebiotic functionality.

In earlier trials carried out in this project aiming to produce AXOS, it was observed that Econase, the endoxylanase used, was yielding only small amounts of oligosaccharides and monosaccharides with an evolving profile as treatment progressed. Knowledge about the precise effects of xylanases on production and degradation of oligosaccharides is not well established; as this has only become of interest relatively recently, there has not been a good method to measure these oligosaccharides, and there has not been the availability of standards. Therefore, it was essential to observe the Econase mode of action on oligosaccharide production and degradation with time during treatment.

The general understanding of endoxylanases is that they perform the hydrolysis in the middle of the polysaccharide chain to create two smaller polysaccharides, in contrast to

 β -xylanases which hydrolyse the terminal sugar unit, reducing the polysaccharide chain into monosaccharides. The studied commercial enzyme preparation, Econase, broke down larger oligosaccharides more rapidly than smaller ones, while showing unexpected production of xylose monomer. Econase treatment of larger molecules, X6 and X5, gave a complex evolving profile of smaller oligos as these were created and then destroyed. Even X3 and X2 were susceptible to Econase degradation, especially at high dosages of the enzyme, possibly due to side activity or, more likely, contamination with an exo-xylanase in the enzyme preparation. Minimising or eliminating the production of the monomer is crucial for maximising the creation of the larger molecules that have the prebiotic functionality.

The substitution pattern of the substrate is a major key in the xylanase mode of action. It was clear that Econase has limited performance in the presence of arabinose side units, as the side chains protect the xylan backbone by steric hindrance. This hindrance is also subject to the location of the arabinose linkage to the xylose unit in the chain; the O-3 linkage showed complete protection of the tetra-xylose chain in the studied AXOS, whereas the O-2 linkage allowed some hydrolysis of the terminal xylose unit, releasing small amounts of xylose. This chain protection may explain the low XOS yields obtained from the treatment of DWG AX in Section 5.5.2, as the extracted AX had a relatively high A/X ratio which represent a high branching pattern that would tend to protect the xylan chain from enzyme attack.

Thus, producing AX extracts of lower A/X ratio may be key for the conversion of AX into (A)XOS. The water-extractable arabinoxylan with its lower A/X ratio might provide a more suitable substrate for the XOS production. Therefore, the next step in this project aimed to integrate the extraction of WEAX from the Solubles fraction within the bioethanol production process and to evaluate the conversion of the produced AX into (A)XOS, in comparison with the water-unextractable AX from the DWG.

8 Extraction of water-extractable and water-unextractable arabinoxylan

within biorefineries

8.1 Introduction

The need to reduce the dependence on fossil fuel along with the ambition to reduce greenhouse gas emissions and climate change have led to the emergence of biorefineries. However, the challenges facing the development and expansion of biorefineries industry were greater than their economic performance could sustain, as demonstrated by the recent Vivergo closure. The transformation into the integrated biorefinery where multiple products are produced and achieve economic viability through successful integration is key for the biorefinery sector to survive and compete with the oil refinery industry which already benefits from extensive process integration. Arabinoxylans are promising candidates to be co-produced within a biorefinery via integration with ethanol production to give a range of AX products including high viscosity AX (high branching WUAX), medium viscosity AX (medium branching, WEAX) and (A)XOS prebiotics. This chapter reviews possible integration scenarios for the development of biorefineries, and presents trials performed to integrate the production of a range of AX products within the biofuel production process using the GUNT bioethanol plant.

8.2 Integrated biorefineries status

Biorefineries are one of the evolving bio-economies that possess great potential to deliver economically competitive and environmentally favourable models of the chemical and energy industries in the 21st century. However, the emergence of biorefineries has been a fragile undertaking thus far, as illustrated by the closure in late 2018 of the Vivergo plant in the UK. The development of the biorefinery industry, as an immature industry, is still struggling against political, technological and economic barriers. Integrated processing, already a key feature of the oil refineries with which

biorefineries must compete, will be key to the technological and economic success of biorefineries. AX-based materials offer a natural co-product for integration with bioethanol, and therefore represent a strategically important opportunity to enhance the viability of biorefineries, while bringing new food, feed and non-food products to these markets.

On the technological and economic side, the integration of biorefineries within an industrial cluster can significantly increase energy efficiency and reduce production costs (Ng et al., 2015). In such a scenario, the industrial cluster will provide both sources and sinks of heat and energy that can be optimised through integration, thereby minimising the process demand for energy. Also, the co-location of several production processes in the same cluster will reduce transportation costs if the feedstock of one process is produced by a co-located process. Another important advantage of this scenario is the reduction in the cost of the infrastructure as facilities of water treatment, steam, safety, electricity, etc. can be shared among several processes (Kimm, 2008). Figure 8-1 represents an overview of the biorefinery cluster concept and the potential products of fuel, energy and biochemicals. However, this scenario lacks the production of food and nutraceuticals; in general biorefineries and food have been seen as incompatible and have been kept separate in government policy (*e.g.* in the emphases of research initiatives), when in fact there are opportunities for synergies that could benefit both biorefineries and the food industry; AX-based products are a notably promising example (Martinez-Hernandez et al., 2018). Nevertheless, the concept of multiple co-products made economically viable through co-located processes that benefit from integration remains at the heart of scenarios for successful biorefineries.



Figure 8-1. Overview of biomass conversion processes and potential products (Ng et al., 2015).

Misailidis *et al.* (2009) presented a process simulation and economic analysis of the coproduction of arabinoxylan alongside ethanol within biorefineries, showing that in this context production of AX, while not cheap, could be economically viable. The integration of the extraction of relatively pure AX from wheat bran, recovered by pearling technology prior to the bioethanol fermentation, with the utilisation of the biorefinery's facilities to recover the used ethanol, seems to offer a feasible pathway to commercialise the production of AX. Recently, Martinez-Hernandez *et al.* (2018) presented a conceptual design for the integration of AX and AXOS production within a bioethanol production facility using mass pinch analysis to minimise bioethanol usage; a range of AX and AXOS products, precipitated at varying ethanol concentrations, increased the scope for integration and for minimising the costs. The work illustrated that retrofitting existing bioethanol processes for integrated AX and AXOS production might offer significant new revenue streams for biorefineries.

However, the production of AX from bioethanol by-products and evaluation studies of integrated AX production in the literature have focused on the solid intermediate wet by-product (DWG), the final dry by-product (DDGS) or the wheat bran fraction before fermentation as the source of AX. The earlier analysis carried out in the current project of the intermediate by-products of bioethanol process, DWG and Solubles, illustrated that around 25% of the total AX of the starting wheat is located in the Solubles fraction (Figure 3-9); evidently water-extractable AX was released from the cell wall structure and solubilised during the hydrolysis and the distillation thermal treatment such that it ended up in the Solubles fraction.

This fraction of AX has been perhaps neglected in the literature due to the assumption that other components of unfermented materials will also be solubilised, such as the plant pigments, proteins and oligopeptides, cellulose oligosaccharides and fractions from the yeast cell wall. Also, it is perhaps assumed that these soluble AX molecules have a small Mw and the separation and purification of the WEAX fraction will not be effective. However, now that it is clear that a range of AX products, including the small XOS and AXOS molecules, are of interest, it is appropriate to examine the AX material in the Solubles stream of a bioethanol plant. Therefore, this part of the work presents trials to recover the WEAX from the Solubles while also extracting the WUAX from the DWG. Both extracts were characterised and then tested as substrates for enzymatic conversion to oligosaccharides.

8.3 Materials and methods

The materials used in this study (the starting raw material, chemicals and enzymes) are as listed in Sections 3.4.1 and 5.4.1. The CE-640 bioethanol plant, described in Section 3.4.2, was used for both the ethanolic fermentation to produce the DWG and Solubles stream and the AX extraction from both DWG and Solubles.

For the liquid-solid separation, a Carlson[®] filter press was used, shown in Figure 8-2 (Carlson Filtration Limited, UK), with an EE4.6HF depth filter sheet $(20 \times 20 \text{ cm})$.



Figure 8-2. Carlson filter press.

For the ultrafiltration concentration and purification, the hybrid version of the ultrafiltration system (described in Section 5.4.2) was further upgraded by combining the Eco[®] gear pump with a high performance AMI[®] spiral membrane polyethersulfone with 10,000 MWCO cut-off, 10 m² surface area and 4" diameter × 40" length dimensions, presented in Figure 8-3, obtained from Wateranywhere[®], USA.



Figure 8-3. AMI[®] spiral ultrafiltration membrane and housing.

The combination of the powerful gear pump with the large surface area spiral membrane enabled fast and effective concentration and diafiltration of the extract with a filtration rate of 1.5-2 litre per minute. The system is described diagrammatically in Figure 8-4 and the final assembled kit is shown in Figure 8-5.



Figure 8-4. UF system diagram.



Figure 8-5. Assembled large-scale ultrafiltration kit.

8.3.1 Methods

Aiming to extract both WEAX and WUAX from DWG and Solubles, the whole process was optimised. The liquefaction was performed in the distiller of the fermentation plant as it offers the facility boil the material. The saccharification, fermentation and the chemical extraction of WUAX were similar to the process carried out in earlier parts of this project. Figure 8-6 illustrates the complete process of WEAX and WUAX production.



Figure 8-6. The process of WEAX and WUAX production.

A batch of 10 kg of ground wheat grain was mixed with 25 litres of water and boiled at 100°C inside the CE-640 distiller for 3 hours. Thereafter, the mixture's temperature was reduced to 90°C, and the whole mixture was transferred manually to the mash tank. The distiller tank was rinsed with 5 litres of water, and the wash was added to the mixture in the mash tank. The enzymatic hydrolysis of the grain starch was commenced by adding α -amylase with continuous stirring at 90°C for one hour. The hydrolysis was

continued for 3 hours with gluco-amylase after reducing the mash temperature to 55°C and adjusting the pH to 4.5.

By the end of the hydrolysis, the mash was transferred to the fermentation tank where the temperature was adjusted to 28±2°C, and the yeast was added. The fermentation was allowed to take place over 72 hours with intermittent stirring. Thereafter, the fermented mash was transferred to the distillation unit where the produced ethanol was separated over 4 hours. After the completion of the distillation procedure, the ethanol was placed into secure storage and the power was switched off to allow the material inside the distiller to cool to the ambient temperature overnight.

The leftover material from the solid fermentation was separated from the mash by filtration using the Carlson filter press and washed with 20 litres of water to remove the adsorbed Solubles from the Distillers Wet Grain.

The DWG was suspended in 20 litres of water in the mash tank, and the pH was adjusted to 8 to be treated with protease at 60°C for four hours. The protein-reduced DWG was filtered and recovered, and the filtrate was discarded. The filtration cake was suspended again in 20 litres of water in order to extract the Water-unextractable arabinoxylan, the pH was adjusted to 11+, the temperature was elevated to 60°C, antifoaming agent was added, and hydrogen peroxide was added gradually with vigorous stirring to reach the concentration of 2%, with the total extraction time being 4 hours. The mixture was cooled to 25°C, the pH was neutralised with hydrochloric acid, the solubilised AX extract was separated from the solids by filtration, the cake was washed with 10 litres of fresh water and filtered, and the filtrate was added to the original filtrate. The total extract was concentrated by ultrafiltration, then diafiltered by adding 20 litres of water to the solution and concentrated again. WUAX was precipitated with 65% ethanol overnight, and precipitated WUAX was collected by centrifugation, dried in a 40°C oven overnight, and stored for analysis and characterisation. The ethanol was recovered by distillation.

The Solubles portion from the fermentation, about 50 litres, was subject to ultrafiltration over 10 kDa to remove the small molecules and recover the water soluble arabinoxylan. The concentrated Solubles were precipitated with 80% ethanol overnight, and precipitated WEAX was collected by centrifuging, dried and stored for further analysis. The ethanol was again recovered again by distillation.

8.3.2 Characterisation of the extracts

Both extracts of WEAX and WUAX were characterised for AX content, average molecular weight, protein, ash, moisture and fat contents. The constituent sugars analysis was carried out using the HPAEC-PAD, and SEC-MALLS was used for the Mw as described in Section 5.5.1. The protein, ash and moisture contents were evaluated as described in Sections 3.6.3 and 3.6.4.

8.3.3 Enzymatic treatment of WEAX and WUAX

For the enzymatic oligosaccharide production, 1 g from each extract was suspended in 100 mL of phosphate-citrate buffer at pH = 5, the samples were stirred overnight to ensure the complete solubilisation of AX in the solution. Samples were placed in a water bath at 50°C and Econase was added at a dose of 50 mL per kg. Sub-samples were taken after 1 and 2 hours and diluted with phosphate-citrate buffer of pH = 7, then boiled for 20 minutes to deactivate the enzyme permanently.

During this research, Econase has been used at a dose of 50 mL per kg which is 500× the commercially recommended dose. This dose was chosen according to preliminary results where the normal dose did not show any effect (Chapter 4). However, the findings presented in Chapter 7 clarified that Econase has the ability to degrade XOS. Thus it became clear that the high dosages employed to accelerate AX breakdown may have also degraded any XOS produced. The development of the new HPAEC-PAD method for the accurate measurement of XOS, and the extraction of more suitable substrate, the WEAX, offered the opportunity to investigate the effect of the enzyme dosage on the production of XOS. Therefore, samples of WEAX was solubilised in buffer

at pH 5, placed in a water bath at 50°C and Econase was added at a doses of 50 mL per kg, 50 mL per 10 kg, 50 mL per 100 kg and 50 mL per 1000 kg for 2 hours. The samples were again diluted with phosphate-citrate buffer of pH = 7 and boiled for 20 minutes to deactivate the enzyme permanently.

The ability of Econase to produce xylose monosaccharide as shown in Chapter 7, and the unusual behaviour of the enzyme towards the branched AX oligosaccharides discussed in Section 7.6, suggest the possibility of Econase being contaminated with traces of an exo-xylanase which might be responsible for the monosaccharides production. As Econase contains a thermal stable endo-xylanase, thermal deactivation is a strategy for eliminating contaminating enzymes. To test this strategy and clarify the reason for the appearance of xylose monomer, Econase solution was heated to 70°C for 30 minutes then applied to the WEAX extract at dose of 50 mL per kg as described above.

8.4 Results and discussion

The fermentation procedure yielded from 10 kg wheat a total of 3.2 litres of 90% ethanol, 2.1 kg (db) of solids in the form of DWG and around 50 litres of Solubles solution. The chemical extraction of WUAX yielded 85 g of crude extract (a 0.85% yield) while the extraction of WEAX resulted in 92 g of crude extract (a 0.92% yield).

The characterisation of the extracts included the analysis of the AX purity, moisture, protein and ash, as shown in Table 8-1. The determination of AX purity was performed by measuring the arabinose and xylose sugars in the sample using HPAEC-PAD after hydrolysing the sample with 2M trifluoroacetic acid for one hour. The recovery of the hydrolysis was calculated for each extract using a suitable mixture of low viscosity AX (50 kDa) and medium viscosity AX (300 kDa) Megazyme standards. To ensure these standards were sufficiently similar to the extracted material to allow recoveries to be calculated, the molecular weight distribution of the samples was evaluated.

	WEAX	WUAX
AX purity %	74.24 ± 0.13	51.62 ± 0.09
A/X ratio	0.49	1.18
Average Mw (kDa)	70	110
Total crude protein %	5.8 ± 0.21	9.8 ± 0.31
Ash %	0.72 ± 0.1	4.54 ± 0.1
Moisture %	8.90 ± 0.49	14.15 ± 0.47

Table 8-1. Chemical composition of WEAX and WUAX

The original 10 kg wheat contained 590 g AX which was divided between ~350 g in DWG and ~150 g in the Solubles. The recovery of 92 g of WEAX of 74% purity implies that 70 g of the Solubles AX was recovered (around 47% of the available 150 g). The purity of the 85 g of WUAX was 51%; this means only 42 g of DWG AX was recovered (around 12% of the available 350 g) which is less than the AX obtained by the extraction performed in Section 5.5.1 where 43% of the DWG AX was recovered. However, the purity of AX extraction had been a challenge (Bell 2015; Jacquemin *et al.,* 2015b) and the purification applied in this project significantly increased the purity of the extract.

The molecular weight distribution of both WEAX and WUAX was measured using SEC-MALLS, giving average weights of 70 and 110 kDa, respectively. The Mw of the water-extractable AX was lower than the water-unextractable AX as expected, because the WEAX naturally comprises AX units with fewer linkages to the cell wall matrix, hence fewer arabinose side branches, whereas WUAX consists of larger molecules with more linkages to the other cell wall components. However, the extracted WUAX presented a smaller Mw than the previously extracted AX, described in Section 5.5.1, which is a result of changing the extraction conditions, particularly the solids–liquid ratio of the extraction. This change in the Mw as a result of the different extraction conditions explains the low yields obtained, as the WUAX has a lower average molecular weight than the earlier material, more AX was lost during the diafiltration procedure.

The total crude protein content in the Solubles WEAX is lower than the DWG WUAX, (5.8% compared with 9.8), even though the dried Solubles had a total protein content of 37% (Section 3.7.3). It is convenient that the majority of the proteins in the Solubles appear to have been small molecules that were able to pass the 10 kDa ultrafiltration

membrane. However, the extraction of WEAX in this project was performed according to a minimum cost; a purification step of protease treatment might be useful in reducing the 5% protein content to an even lower level.

The water-unextractable AX protein content was about 10%, which is lower than the 14% protein content of the previous extracts (Section 5.5.1). This indicates that protease treatment could be more effective when it precedes the AX extraction, as performed in this current experiment.

The low ash content in the WEAX compared to the WUAX is due to the chemical-free recovery method used in this project. However, the ~5% ash content in the WUAX is significantly less than the ash content in the previous extracts, around 13%; this is a reflection of the performance of the upgraded ultrafiltration system and the use of diafiltration.

The low arabinose substitution of the WEAX suggested this might be a suitable substrate for oligosaccharide production via enzymatic treatment. 1 g of this extract was therefore resolubilised in 100 mL of buffer at pH 5 and treated with 50 μ L Econase at 50°C for 1 and 2 hours. The amounts of xylose plus oligosaccharides up to X6 produced were measured.

Figure 8-7 shows the percentage conversion of the extracted AX after Econase treatment for one and two hours. Clearly, Econase treatment of the WEAX for 1 hour produced significant amounts of X2 and X3 as well as some xylose monomer, but there was no evidence of the larger oligos (X4-X6). Longer treatment degraded the X3 into X2 and X1. In general, the treatment yielded 46% conversion of the material in the form of small XOS (xylobiose 27% and xylotriose 19%). A small portion (11%) of the WEAX was converted into monosaccharide after one-hour treatment. The longer treatment reduced the amount of xylotriose to 40% and doubled the amount of xylose.



Figure 8-7. XOS production from WEAX and WUAX via Econase treatment for 1 hour and 2 hours.

For the WUAX, the treatment was much less effective, yielding mostly xylose and a smaller amount of xylobiose after 1 hour, and a little more after two hours. Evidently the high branching nature of WUAX protects the molecule against the enzymatic hydrolysis apart from the release of some xylose and traces of xylobiose, whereas the low substituted WEAX seems to be a more susceptible substrate for the enzyme activity.

Although the WUAX might not be the optimal substrate for prebiotic production, this does not compromise its importance for utilisation in other application such as food and pharmaceutical industry as a novel biopolymer that can serve several purposes (such as a thickener, gel making, film making) (Chen *et al.*, 2016; Erum *et al.*, 2015; Khathuriya *et al.*, 2015; Lazaridou and Biliaderis, 2007; Phan *et al.*, 2002).

Figure 8-8 presents the Econase treatment of WEAX when used at a doses of 50 mL per kg, 50 mL per 10 kg, 50 mL per 100 kg and 50 mL per 1000 kg (500×, 50×, 5× and 0.5× the commercial dose) for 2 hours. Clearly, the monosaccharide concentration was greater with the high dose (about 23%), whilst with lower doses the xylose concentration did not exceed 3%. The total released XOS was about 53% with the high (500×) dose comprising mainly X2 (32%), X3 (17%) and X4 (3%); whilst the total XOS was smaller with the lower doses (3% for 0.5×, 34% for 5× and 45% for 50×). The large XOS (X5 and X6) appears only with the lower dosages.



Figure 8-8. WEAX treatment with Econase at $500\times$, $50\times$, $5\times$ and $0.5\times$ the commercial dose.

Evidently, the high dose of the Enzyme produced greater amounts of the total XOS with notable elimination of the larger XOS (X5 and X6). However, the high yield of X2 and X3 is desired as these small oligos have the highest prebiotic activity (Gullón *et al.*, 2008a; Van Craeyveld *et al.*, 2008).

The ability of Econase to produce undesired monosaccharides was significantly reduced after thermal treatment. Figure 8-9 shows the difference in Econase effect on WEAX (at a dose of 50 mL per kg) before and after heating the enzyme solution to 70°C for 30 minutes. The amount of released monosaccharide was significantly decreased from 20% to 1.5% after the thermal treatment. Also, the total XOS produced (X2, X3 and X4) was increased from 51% to 59%, also the XOS profile shows a higher X3 after thermal treatment, whilst X2 was predominant in the direct treatment.



Figure 8-9. Econase effect before and after thermal treatment at 70°C for 30 minutes.

Clearly, the results suggest that the commercial Econase preparation does not contain a pure endoxylanase, and a contaminant exo-xylanase is affecting the overall behaviour of the Econase. The increased amount of X3 implies that the contaminant enzyme has an exo-xylanase activity that converts the X3 into X2 and monomeric xylose. It is not evident but possible that this contamination is affecting all the produced XOS. Also, this contamination explains the Econase behaviour towards the branched oligos discussed in Section 7.6 where Econase performed on the terminal xylose in the branched AXOS. Building on the emerging picture from the earlier chapters, these trials have drawn together several threads of evidence to demonstrate that the Solubles stream from a bioethanol plant is a promising source of AX for enzymatic production of (A)XOS; the low A/X ratio of the WEAX extracted from Solubles makes them susceptible to enzyme degradation to produce the oligosaccharides. This work has also confirmed that Econase contains some contaminating exo-xylanase, the activity of which could be removed through a simple thermal treatment. Eliminating the production of xylose monomer increased the yield of the desired xylobiose and xylotriose.

The discovery of this contamination compromises the interpretation of the enzyme dynamics from the study presented in Chapter 7 on Econase treatment of standards. That study also required a greater range of enzyme/substrate ratios to obtain a full picture of Econase kinetics. Nevertheless, the results from the last few chapters have increased our understanding of Econase behaviour and have demonstrated how this endoxylanase, or other xylanases, could be deployed to optimise production of (A)XOS prebiotics from bioethanol by-products.

8.5 Summary

The bioethanol production process yields two intermediates by-products, Distillers Wet Grain and Solubles, to which thermal evaporation and drying procedures are applied to combine the intermediate by-products into the final and major by-product in biorefineries, Distillers Dried Grain with Solubles (DDGS).

The isolation of arabinoxylans from the DWG and the Solubles, prior to the drying procedure, yields different AX materials with different properties, including differing susceptibilities to xylanase treatment to produce oligosaccharides. Production of these two AX products and of XOS/AXOS products from them could significantly enhance the economic performance of cereal-based biorefineries, by creating new products that could be commercialised for the food market.

In this project, two arabinoxylan products, the water-extractable AX from the Solubles and the water-unextractable AX from the DWG, were successfully produced. The waterextractable AX was extracted from the aqueous phase of the fermentation mash, the Solubles. The thermal treatment of the wheat during the process of ethanol production, *i.e.* the starch hydrolysis prior to fermentation and the ethanol distillation after fermentation, was sufficient to release a substantial portion of AX into the Solubles. The isolation of WEAX from the Solubles was performed using an in-house assembled ultrafiltration system with a membrane of 10 kDa cut-off.

The WUAX was chemically extracted from the DWG; the oxidative alkaline extraction was able to break the material open and release the tightly connected AX from other cell wall components. After extraction, the liquid extracts were concentrated, diafiltered and precipitated with 65% ethanol. The purity of the extracted WUAX was higher than the purity of the extract in Chapter 6 due to the utilisation of a more effective ultrafiltration system. However, the yield was lower and the average molecular weight was smaller due to changing the extraction conditions, particularly the liquid-solid ratio.

WEAX was recovered from the Solubles by mean of ultrafiltration followed by ethanol precipitation. The yield was about 1% of the original wheat, and about 47% of the available AX in the Solubles. Despite the relatively low yield, the low cost of the process where no chemical extraction is required, and the high purity of the recovered WEAX (74%) make this route of AX production viable in the context of biorefineries.

The enzymatic conversion of WEAX into XOS prebiotics yielded a 46% conversion of the WEAX into XOS mainly xylobiose and xylotriose.

The enzyme dosage trial showed that higher doses of the enzyme produce greater amounts of XOS (X2, X3 and X4) with higher concentrations of xylose monosaccharide, whilst the lower doses produced a wide range of XOS molecules (X2 – X6) but the total XOS was less. The production of monosaccharides was confirmed to be the result of the

contamination of the enzyme with traces of an exo-xylanase, which could be deactivated via gentle thermal treatment.

The results demonstrate a previously unconsidered source of AX, the Solubles fraction, and show that its low A/X ratio makes it potentially well suited to enzymatic production of XOS. The ability to measure the XOS profile accurately and to understand its dynamic evolution opens the way to optimise production of the desired prebiotic oligosaccharides.

9 Conclusions and Recommendations

9.1 Progress made in the current work

In the context of exploring strategies for enhancing the economic income from biorefineries, the literature advocates the integrated biorefinery approach as an effective basis for exploiting interaction opportunities arising from the co-production of several products. Biorefineries are currently derived principally from ethanol production, and ethanol can serve as a precipitating agent for arabinoxylans, which makes arabinoxylans potentially a natural co-product of bioethanol, and raising the opportunity for bringing a new class of functional food ingredients to the market.

The primary motivation of the current work was identified earlier as the pilot-scale production of arabinoxylan and arabinoxylan oligosaccharides within integrated biorefineries. In order to investigate this opportunity, several objectives were identified.

The first objective was to perform the bioethanol fermentation process on a pilot-scale to produce the wet intermediate by-products. The wet materials produced during the project, DWG and Solubles, had similar characteristics to the materials reported in the literature, and are representative of the intermediate by-products produced in commercial bioethanol plants.

The second objective was to investigate the direct enzymatic production of AX prebiotics from the DWG. The investigations revealed that the enzymes were incapable of releasing the xylooligosaccharides from the cell wall matrix, which emphasised the importance of the third objective.

The third objective was to extract AX chemically by mean of the alkaline oxidative method. The extraction process yielded two extracts with different characteristics according to the purification methods used. Use of enzymes and ultrafiltration increased the purity of the extract from 19% (the first un-purified extract) to 44% after

purification. The first extract was sent to Nottingham Trent University, where collaborating academics introduced the extract to chicken feed formulation trials to assess the effect of the AX polysaccharide on poultry nutrition. The second extract was subjected to several enzymatic treatments aiming to produce significant quantities of XOS. The trials produced only small amounts of XOS, and revealed the limitations of the analytical techniques used for XOS measurement. The trials also stressed the importance of understanding the studied enzyme's mode of action.

The fourth objective addressed in this project was the development of a robust separation and measurement technique for the simultaneous determination of monosaccharides, xylo-oligosaccharides, arabinoxylan oligosaccharides and uronic acids. The developed HPAEC-PAD method is a novel and effective tool for XOS analysis and enzymes activity studies. The simultaneous screening of all the potential products of enzyme activity significantly enhances the basis for enzyme development in relation to XOS production, and gives the potential for precise and conclusive descriptions of enzyme modes of action. The method was validated, and several samples were sent to our lab from academic and industrial partners to be analysed, indicating the timeliness and value of this new method.

The fifth objective was to study the enzymatic production and simultaneous degradation of XOS and to evaluate the ability of Econase, an endoxylanases, to produce monosaccharides. The study showed that Econase was active on all the XOS standards up to DP6, acting more rapidly on the larger oligosaccharides, and with significant and unexpected monosaccharide production. The enzyme had only minimal ability to hydrolyse the branched substrates. Therefore, an AX substrate with less branching (a lower A/X ratio) is likely to be a more suitable feedstock for XOS production.

The final objective of the project was to produce both the Water-Extractable and the Water-Unextractable AX within the biorefinery. The WUAX was extracted chemically following the same methods used earlier, but with more extensive diafiltration purification. The WEAX was recovered from the Solubles fraction by ultrafiltration over a 10 kDa membrane. The processing of 10 kg of wheat yielded only 85 g of 51% WUAX

and 92 g of 74% WEAX. The enzymatic conversion of WEAX into XOS was capable of converting 46% of the material into oligosaccharides and 11% into monomers. Thermal treatment of the Econase largely eliminated the production of xylose, indicating this had arisen from a heat-labile contaminant exo-xylanase.

Overall the work has moved forward our understanding of the practicalities of producing AX materials at scale from the DWG and Solubles streams in a bioethanol plant, has introduced a powerful new method for analysing mono- and oligosaccharides, has clarified the kinetics of Econase activity in relation to production and degradation of oligosaccharides, and has identified the Solubles stream as a promising source of AX with a low A/X ratio, suitable for enzymatic production of XOS.

9.2 Recommendations for future work and industrial uptake

The AX extract produced suffered from low yields and purities, in common with results from other researchers (Bell 2015; Jacquemin *et al.*, 2015b). Steps to improve purity naturally reduced yields, but were able to get up to 74% AX content. Purification of the extracts might be enhanced by the utilisation of large-scale ion exchange columns. Also, a high-speed centrifugation might significantly enhance the extraction. In general, an objective of future work ought to be to continue to explore ways of enhancing AX yields and purities. It is also recommended to satudy the effect of the drying method on the characteristics of the final AX extracts.

The enzymatic production of arabinoxylan oligosaccharides was accompanied with the creation of undesired xylose monomers. In part this was because, having created oligosaccharides from the larger AX, the xylanase then turns its attention to these molecules and further degrades them. Therefore, it may be plausible to perform the enzymatic hydrolysis simultaneously with ultrafiltration, where the small produced oligos would be removed from the solution, and the enzyme activity would be directed towards the large molecule. In general, there is a need for greater understanding of the dynamics of the evolution of the profile of oligosaccharides, in order to implement and

optimise enzyme treatment to maximise production of the desired XOS with prebiotic functionality.

The current work successfully produced two types of soluble AX, WEAX from the Solubles and WUAX from the DWG. The first had a lower A/X ratio indicating less substitution, hence was more readily hydrolysed by xylanase and more suitable for enzymatic conversion into XOS, while the WUAX was resistant to enzyme activity but may have food ingredient uses as an emulsifier, thickener or just a source of fibres for fibre-rich diets, or may find uses in non-food applications.

The techno-economic study of Misailidis *et al.* in 2009 formed the original justification for ongoing work into AX production in the context of integrated biorefineries. Since then it has become clear that a portfolio of products, including prebiotic AXOS, would be more economically attractive than a single AX product. Ongoing techno-economic studies will be vital to determine the cost and the feasibility of the AX and AXOS production procedures presented in this project prior to introducing them to industrial scale production.

The progress achieved in this work started with the pilot-scale production of representative bioethanol co-products to allow the investigation of functional AX and prebiotic AXOS production within biorefineries. The new method for mono- and oligo-saccharide analysis has opened the way to understanding the mode of action of arabinoxylan hydrolysing enzymes and refining the targeted production of desired products. The new method is also applicable for other fields of study regarding production of oligosaccharides, acidic and amino sugars. The identification of the AX from the Solubles fraction as a promising source of prebiotic AXOS is a significant milestone in creating marketable products. More generally this work, by clarifying the scope for multiple AX-based co-products and establishing the basis for their effective production and characterisation, has helped moved the entire sector forwards toward multiple product and economically successful biorefineries that can deliver their unique role in providing food, biomaterials and fuel sustainably for future generations.

References

- (AOAC), A. of O. A. C., and Horwitz, W. (2000). Protein in grains. Method 979.09. *Official Method of Analysis of AOAC International*, 30–34.
- #BPstats. (2016). BP Statistical Review of World Energy June 2016. Retrieved from https://www.bp.com/content/dam/bp/pdf/energy-economics/statistical-review-2016/bp-statistical-review-of-world-energy-2016-full-report.pdf
- Aachary, A. A., and Prapulla, S. G. (2009). Value addition to corncob: Production and characterization of xylooligosaccharides from alkali pretreated lignin-saccharide complex using Aspergillus oryzae MTCC 5154. *Bioresource Technology*, 100(2), 991–995. https://doi.org/10.1016/j.biortech.2008.06.050
- Aachary, A. A., and Prapulla, S. G. (2011). Xylooligosaccharides (XOS) as an Emerging Prebiotic: Microbial Synthesis, Utilization, Structural Characterization, Bioactive Properties, and Applications. *Comprehensive Reviews in Food Science and Food Safety*, 10(1), 2–16. https://doi.org/10.1111/j.1541-4337.2010.00135.x
- Agarwal, A. K., Agarwal, R. A., Gupta, T., and Gurjar, B. R. (2017). *Biofuels: technology, challenges and prospects*.
- Akpinar, O., Erdogan, K., Bakir, U., and Yilmaz, L. (2010). Comparison of acid and enzymatic hydrolysis of tobacco stalk xylan for preparation of xylooligosaccharides. *LWT - Food Science and Technology*, 43(1), 119–125. https://doi.org/10.1016/j.lwt.2009.06.025
- Alberici, S., and Toop, G. (2013). UK biofuel industry overview, 2. Retrieved from https://www.gov.uk/government/uploads/system/uploads/attachment_data/file /266090/ecofys-uk-biofuel-industry-overview-v1.5.pdf
- Alijosius, S., Miezeliene, A., Alencikiene, G., and Bliznikas, S. (2018). Effect of Triticale and Non-Starch Polysaccharides (NSP) Degrading Enzymes on Colour And Sensory Characteristics of Broiler Meat. *Veterinarija Ir Zootechnika* ·, (June). https://doi.org/https://vetzoo.lsmuni.lt/data/vols/2018/076/pdf/alijosius.pdf
- Almeida, J. R. M., Modig, T., Petersson, A., Hähn-Hägerdal, B., Lidén, G., and Gorwa-Grauslund, M. F. (2007). Increased tolerance and conversion of inhibitors in lignocellulosic hydrolysates by Saccharomyces cerevisiae. *Journal of Chemical Technology and Biotechnology*, 82(4), 340–349.
- Álvarez, C., González, A., Negro, M. J., Ballesteros, I., Oliva, J. M., and Sáez, F. (2017). Optimized use of hemicellulose within a biorefinery for processing high valueadded xylooligosaccharides. *Industrial Crops and Products*, 99, 41–48. https://doi.org/10.1016/j.indcrop.2017.01.034
- Amin, H. M. (2006). Extended usage of xylanase enzyme to enhance the bleaching of softwood kraft pulp. *Tappi Journal*, *5*(1), 23–26.
- Anders, N., Humann, H., Langhans, B., and Spieß, A. C. (2015). Simultaneous determination of acid-soluble biomass-derived compounds using high

performance anion exchange chromatography coupled with pulsed amperometric detection. *Analytical Methods*, 7(18), 7866–7873. https://doi.org/10.1039/C5AY01371B

- Andrewartha, K. A., Phillips, D. R., and Stone, B. A. (1979). Solution properties of wheat-flour arabinoxylans and enzymically modified arabinoxylans. *Carbohydrate Research*, 77(1), 191–204. https://doi.org/10.1016/S0008-6215(00)83805-7
- Annison, G., Choct, M., and Cheetham, N. W. (1992). Analysis of wheat arabinoxylans from a large-scale isolation. *Carbohydrate Polymers*, *19*(3), 151–159. https://doi.org/10.1016/0144-8617(92)90155-J
- Anonymous. (2009). *Biorefineries: adding value to the sustainable utilisation of biomass*. Retrieved from http://www.ieabioenergy.com/wp-content/uploads/2013/10/Task-42-Booklet.pdf
- Anonymous. (2017a). Advanced drop in biofuels UK production capacity outlook to 2030 Final Report. UK Department for Transport. Retrieved from https://www.gov.uk/government/publications/advanced-drop-in-biofuels-ukproduction-capacity-outlook-to-2030
- Anonymous. (2017b). Global Markets for Enzymes in Industrial Applications: BIO030J | BCC Research. Retrieved August 11, 2018, from https://www.bccresearch.com/market-research/biotechnology/enzymesindustrial-applications-report-bio030j.html
- anonymous a. (2016). Renewables 2016 Global Status Report. Renewables 2016 Global Status Report. https://doi.org/ISBN 978-3-9818107-0-7
- Antov, M. G., and Đorđević, T. R. (2017). Environmental-friendly technologies for the production of antioxidant xylooligosaccharides from wheat chaff. *Food Chemistry*, 235, 175–180. https://doi.org/10.1016/j.foodchem.2017.05.058
- Araki, T., and Kitamikado, M. (1982). Purification and characterization of a novel exoβ-mannanase from Aeromonas sp. F-25. *Journal of Biochemistry*, *91*(4), 1181– 1186.
- Arruda, H. S., Pereira, G. A., and Pastore, G. M. (2017). Oligosaccharide profile in Brazilian Cerrado fruit araticum (Annona crassiflora Mart.). LWT - Food Science and Technology, 76, 278–283. https://doi.org/10.1016/j.lwt.2016.05.017
- Asche, F., Gjølberg, O., and Völker, T. (2003). Price relationships in the petroleum market: An analysis of crude oil and refined product prices. *Energy Economics*, 25(3), 289–301. https://doi.org/10.1016/S0140-9883(02)00110-X
- Austin, S. C., Wiseman, J., and Chesson, A. (1999). Influence of non-starch polysaccharides structure on the metabolisable energy of U.K. wheat fed to poultry. *Journal of Cereal Science*, 29(1), 77–88. https://doi.org/10.1006/jcrs.1998.0213
- Avellar, B. K., and Glasser, W. G. (1998). Steam-assisted biomass fractionation. I. Process considerations and economic evaluation. *Biomass and Bioenergy*, *14*(3), 205–218. https://doi.org/10.1016/S0961-9534(97)10043-5
- Ayala-Soto, F. E., Serna-Saldívar, S. O., and Welti-Chanes, J. (2016). Effect of processing

time, temperature and alkali concentration on yield extraction, structure and gelling properties of corn fiber arabinoxylans. *Food Hydrocolloids*, *60*, 21–28. https://doi.org/10.1016/j.foodhyd.2016.03.014

- Bacic, A., Harris, P. J., and Stone, B. A. (1988). Structure and function of plant cell walls. *The Biochemistry of Plants*, *14*, 297–371.
- Bacic, A., and Stone, B. A. (1981). Chemistry and Organization of Aleurone Cell Wall Components From Wheat and Barley. *Functional Plant Biology*, 8(5), 475–495. https://doi.org/http://dx.doi.org/10.1071/PP9810475
- Baillet, E., Downey, G., and Tuohy, M. (2003). Improvement of texture and volume in white bread rolls by incorporation of microbial hemicellulase preparations. In *Recent Advances in Enzymes in Grain Processing, Proceedings of the 3rd European Symposium on Enzymes in Grain Processing (ESEGP-3)* (pp. 255–259). Katholieke Universiteit Leuven, Leuven, Belgium.
- Bajpai, P. (1999). Application of enzymes in the pulp and paper industry. *Biotechnology Progress*, *15*(2), 147–157. https://doi.org/10.1021/bp990013k
- Bajpai, P. (2014a). Industrial Applications of Xylanases. In *Xylanolytic Enzymes* (pp. 69–104). Academic Press. https://doi.org/10.1016/B978-0-12-801020-4.00008-1
- Bajpai, P. (2014b). Introduction. In *Xylanolytic Enzymes* (pp. 1–7). Academic Press. https://doi.org/10.1016/B978-0-12-801020-4.00001-9
- Banerjee, P. N. (2014). Comparative Analysis of Sugarcane Baggase Using Different
 Methods. International Journal of Scientific and Engineering Research, 5(9), 134–
 137. Retrieved from http://www.ijser.org
- Bashir, S., Erum, A., Saghir, S., Tulain, U. R., and Rashid, A. (2014). Physicochemical characterization and evaluation of suspending properties of arabinoxylan from Ispaghula (Plantago ovata) husk. *Pakistan Journal of Pharmaceutical Sciences*, 27(6).
- Beaugrand, J., Chambat, G., Wong, V. W. K., Goubet, F., Rémond, C., Paës, G., ... Chabbert, B. (2004). Impact and efficiency of GH10 and GH11 thermostable endoxylanases on wheat bran and alkali-extractable arabinoxylans. *Carbohydrate Research*, 339(15), 2529–2540. https://doi.org/10.1016/j.carres.2004.08.012
- Bedford, M. R., and Morgan, A. J. (1996). The use of enzymes in poultry diets *. World's Poultry Science Journal, 52(March 1995), 61–68. https://doi.org/10.1079/WPS19960007
- Beldman, G., Osuga, D., and Whitaker, J. R. (1996). Some characteristics of D xylopyranosidases, alpha- L - arabinofuranosidases and an arabinoxylan alpha- L arabinofuranohydrolase from wheat bran and germinated wheat. J Cereal Sci, 23(2), 169–180.
- Bell, R. M., Webb, S. C., and Campbell, G. (2015). Extraction of Arabinoxylan from Animal Feed and Investigations into its Functionality as an Ingredient in Bread Dough . [Thesis]. Manchester, UK: The University of Manchester; 2015. The University of Manchester, Manchester, UK. Retrieved from https://www.escholar.manchester.ac.uk/uk-ac-man-scw:266994

- Belyea, R. L., Rausch, K. D., Clevenger, T. E., Singh, V., Johnston, D. B., and Tumbleson, M. E. (2010). Sources of variation in composition of DDGS. *Animal Feed Science and Technology*, *159*(3–4), 122–130. https://doi.org/10.1016/J.ANIFEEDSCI.2010.06.005
- Benamrouche, S., Crônier, D., Debeire, P., and Chabbert, B. (2002). A Chemical and Histological Study on the Effect of (1→4)-β-endo-xylanase Treatment on Wheat Bran. *Journal of Cereal Science*, 36(2), 253–260. https://doi.org/10.1006/jcrs.2001.0427
- Bergmans, M. E. F., Beldman, G., Gruppen, H., and Voragen, A. G. J. (1996).
 Optimisation of the selective extraction of (glucurono)arabinoxylans from wheat bran: Use of barium and calcium hydroxide solution at elevated temperatures.
 Journal of Cereal Science, 23(3), 235–245. https://doi.org/10.1006/jcrs.1996.0024
- Berti, P., and Levidow, L. (2014). Fuelling expectations: A policy-promise lock-in of UK biofuel policy. *Energy Policy*, 66, 135–143. https://doi.org/10.1016/j.enpol.2013.09.044
- Bian, J., Peng, F., Peng, P., Xu, F., and Sun, R. C. (2010). Isolation and fractionation of hemicelluloses by graded ethanol precipitation from Caragana korshinskii. *Carbohydrate Research*, 345(6), 802–809. https://doi.org/10.1016/j.carres.2010.01.014
- Bian, J., Peng, F., Peng, X. P., Peng, P., Xu, F., and Sun, R. C. (2013). Structural features and antioxidant activity of xylooligosaccharides enzymatically produced from sugarcane bagasse. *Bioresource Technology*, 127, 236–241. https://doi.org/10.1016/j.biortech.2012.09.112
- Bian, J., Peng, P., Peng, F., Xiao, X., Xu, F., and Sun, R. C. (2014). Microwave-assisted acid hydrolysis to produce xylooligosaccharides from sugarcane bagasse hemicelluloses. *Food Chemistry*, 156, 7–13. https://doi.org/10.1016/j.foodchem.2014.01.112
- Biely, P., Vrsanska, M., Tenkanan, M., Kluepfel, D. (1997). Endo-P 1,4-xylanase families: differences in catalytic properties. *Journal of Biotechnology*, 57(1), 151– 166. https://doi.org/https://doi.org/10.1016/S0168-1656(97)00096-5
- Biely, P., Vršanaka, M., and Kratky, Z. (1981). Mechanisms of Substrate Digestion by Endo-1,4-β-Xylanase of Cryptococcus albidus Lysozyme-Type Pattern of Action. *European Journal of Biochemistry*, *119*(3), 565–571. https://doi.org/10.1111/j.1432-1033.1981.tb05645.x
- Biliaderis, C. G., Izydorczyk, M. S., Rattan, O., Biliaderis, Izydorczyk, and Rattan. (1995). Effect of arabinoxylans on bread-making quality of wheat flours. *Food Chemistry*, 53(2), 165–171. https://doi.org/doi.org/10.1016/0308-8146(95)90783-4
- Biliaderis, Izydorczyk, and Rattan. (1995). Effect of arabinoxylans on bread-making quality of wheat flours. *Food Chemistry*, *53*(2), 165–171.
- Blanchard, P. H. (1992). Technology of Corn Wet Milling and Associated Processes. Elsevier. Retrieved from http://agris.fao.org/agrissearch/search.do?recordID=US201300719001

- Boll, E. V. J., Ekström, L. M. N. K., Courtin, C. M., Delcour, J. A., Nilsson, A. C., Björck, I. M. E., and Östman, E. M. (2015). Effects of wheat bran extract rich in arabinoxylan oligosaccharides and resistant starch on overnight glucose tolerance and markers of gut fermentation in healthy young adults. *European Journal of Nutrition*, 1–10. https://doi.org/10.1007/s00394-015-0985-z
- Bomb, C., McCormick, K., Deurwaarder, E., and Kåberger, T. (2007). Biofuels for transport in Europe: Lessons from Germany and the UK. *Energy Policy*, 35(4), 2256–2267. https://doi.org/10.1016/j.enpol.2006.07.008
- Bonner, P. L. (2007). Enzymes: Biochemistry, Biotechnology and Clinical Chemistry (2nd Ed.). Cambridge: Woodhead Publishing Limited,. Retrieved from https://biotechku.files.wordpress.com/2017/08/t-palmer-and-p-l-bonner-auth-enzymes-biochemistry-biotechnology-clinical-chemistry-woodhead-publishing-2007.pdf
- Brienzo, M., Carvalho, W., and Milagres, A. M. F. (2010). Xylooligosaccharides
 Production from Alkali-Pretreated Sugarcane Bagasse Using Xylanases from
 Thermoascus aurantiacus. *Applied Biochemistry and Biotechnology*, 162(4), 1195–1205. https://doi.org/10.1007/s12010-009-8892-5
- Brillouet, J.-M., and Joseleau, J.-P. (1987). Investigation of the structure of a heteroxylan from the outer pericarp (beeswing bran) of wheat kernel. *Carbohydrate Research*, 159(1), 109–126. https://doi.org/10.1016/S0008-6215(00)90009-0
- Broekaert, W. F., Courtin, C. M., Verbeke, K., Van de Wiele, T., Verstraete, W., and Delcour, J. A. (2011). Prebiotic and Other Health-Related Effects of Cereal-Derived Arabinoxylans, Arabinoxylan-Oligosaccharides, and Xylooligosaccharides. *Critical Reviews in Food Science and Nutrition*, *51*(2), 178–194. https://doi.org/10.1080/10408390903044768
- Buruiana, C. T., Gómez, B., Vizireanu, C., and Garrote, G. (2017). Manufacture and evaluation of xylooligosaccharides from corn stover as emerging prebiotic candidates for human health. *LWT - Food Science and Technology*, 77, 449–459. https://doi.org/10.1016/j.lwt.2016.11.083
- Butt, M. S., Tahir-Nadeem, M., Ahmad, Z., and Sultan, M. T. (2008). Xylanases and their application in baking industry. *Food Technology and Biotechnology*, *46*(1), 22–31.
- Campbell, G. M., Čukelj, N., Alyassin, M., Gomez, L. D., Simister, R., Flint, J., ... Westwood, N. J. (2019). processing of sugarcane bagasse: Arabinoxylan extraction integrated with ethanol production. *Biochemical Engineering Journal*. https://doi.org/Submitted
- Campbell, G., Martinez-Hernandez, E., Sadhukhan, J., and Hallett, J. P. (2018).
 Biorefining and Integrated Bioresource Engineering. *The Chemical Engineer*, 41–43. Retrieved from https://www.thechemicalengineer.com/features/biorefining-and-integrated-bioresource-engineering/
- Cara, C., Ruiz, E., Ballesteros, I., Negro, M. J., and Castro, E. (2006). Enhanced enzymatic hydrolysis of olive tree wood by steam explosion and alkaline peroxide delignification. *Process Biochemistry*, *41*(2), 423–429.

https://doi.org/10.1016/j.procbio.2005.07.007

- Carpita, N. C., and Gibeaut, D. M. (1993). Structural models of primary cell walls in flowering plants: Consistency of molecular structure with the physical properties of the walls during growth. *Plant Journal*, *3*(1), 1–30. https://doi.org/10.1111/j.1365-313X.1993.tb00007.x
- Chanliaud, E., Saulnier, L., and Thibault, J. F. (1995). Alkaline extraction and characterisation of heteroxylans from maize bran. *Journal of Cereal Science*, *21*(2), 195–203. https://doi.org/10.1016/0733-5210(95)90035-7
- Chapla, D., Pandit, P., and Shah, A. (2012). Production of xylooligosaccharides from corncob xylan by fungal xylanase and their utilization by probiotics. *Bioresource Technology*, *115*, 215–221. https://doi.org/10.1016/j.biortech.2011.10.083
- Chatzifragkou, A., Kosik, O., Prabhakumari, P. C., Lovegrove, A., Frazier, R. A., Shewry, P. R., and Charalampopoulos, D. (2015). Biorefinery strategies for upgrading Distillers' Dried Grains with Solubles (DDGS). *Process Biochemistry*, *50*(12), 2194–2207. https://doi.org/10.1016/j.procbio.2015.09.005
- Chen, G.-G., Qi, X.-M., Guan, Y., Peng, F., Yao, C.-L., and Sun, R.-C. (2016). High Strength Hemicellulose-Based Nanocomposite Film for Food Packaging Applications. *ACS Sustainable Chemistry and Engineering*, acssuschemeng.5b01252. https://doi.org/10.1021/acssuschemeng.5b01252
- Chen, H. H., Chen, Y. K., Chang, H. C., and Lin, S. Y. (2012). Immunomodulatory Effects of Xylooligosaccharides. *Food Science and Technology Research*, *18*(2), 195–199. https://doi.org/10.3136/fstr.18.195
- Choct, M., hughes, R. J., and bedford, M. R. (1999). Effects of a xylanase on individual bird variation, starch digestion throughout the intestine, and ileal and caecal volatile fatty acid production in chickens fed wheat. *British Poultry Science*, 40(3), 419–422. https://doi.org/10.1080/00071669987548
- Christophersen, C. T., Petersen, A., Licht, T. R., and Conlon, M. A. (2013). Xylooligosaccharides and inulin affect genotoxicity and bacterial populations differently in a human colonic simulator challenged with soy protein. *Nutrients*, 5(9), 3740–3756. https://doi.org/10.3390/nu5093740
- Chung, Y.-C., Hsu, C.-K., Ko, C.-Y., and Chan, Y.-C. (2007). Dietary intake of xylooligosaccharides improves the intestinal microbiota, fecal moisture, and pH value in the elderly. *Nutrition Research*, 27(12), 756–761. https://doi.org/10.1016/J.NUTRES.2007.09.014
- Cleemput, G., Hessing, M., van Oort, M., Deconynck, M., and Delcour, J. A. (1997). Purification and characterization of a [beta]-D-xylosidase and an endo-xylanase from wheat flour. *Plant Physiology*, *113*(2), 377–386.
- Coelho, E., Rocha, M. A. M., Moreira, A. S. P., Domingues, M. R. M., and Coimbra, M. A. (2016). Revisiting the structural features of arabinoxylans from brewers' spent grain. *Carbohydrate Polymers*, 139, 167–176. https://doi.org/10.1016/j.carbpol.2015.12.006

Collins, T., Gerday, C., and Feller, G. (2005, January 1). Xylanases, xylanase families and

extremophilic xylanases. *FEMS Microbiology Reviews*. Oxford University Press. https://doi.org/10.1016/j.femsre.2004.06.005

- Collins, T., Meuwis, M.-A., Stals, I., Claeyssens, M., Feller, G., and Gerday, C. (2002). A Novel Family 8 Xylanase, Functional and Physicochemical Characterization. *Journal of Biological Chemistry*, 277(38), 35133–35139. https://doi.org/10.1074/jbc.M204517200
- Courtin, C. M., Broekaert, W. F., Swennen, K., Lescroart, O., Onagbesan, O., Buyse, J., ... Delcour, J. A. (2008). Dietary inclusion of wheat bran arabinoxylooligosaccharides induces beneficial nutritional effects in chickens. *Cereal Chemistry*, 85(5), 607– 613. https://doi.org/10.1094/CCHEM-85-5-0607
- Courtin, C. M., and Delcour, J. A. (1998). Physicochemical and Bread-Making Properties of Low Molecular Weight Wheat-Derived Arabinoxylans. *Journal of Agricultural and Food Chemistry*, *46*(10), 4066–4073. https://doi.org/10.1021/jf980339t
- Courtin, C. M., and Delcour, J. A. (2002). Arabinoxylans and endoxylanases in wheat flour bread-making. *Journal of Cereal Science*, *35*(3), 225–243. https://doi.org/10.1006/jcrs.2001.0433
- Courtin, C. M. M., and Delcour, J. A. A. (2001). Relative activity of endoxylanases towards water-extractable and water-unextractable arabinoxylan. *Journal of Cereal Science*, 33(3), 301–312. https://doi.org/10.1006/jcrs.2000.0354
- Courtin, C., Swennen, K., Verjans, P., and Delcour, J. A. (2009). Heat and pH stability of prebiotic arabinoxylooligosaccharides, xylooligosaccharides and fructooligosaccharides. *Food Chemistry*, 112(4), 831–837. https://doi.org/10.1016/j.foodchem.2008.06.039
- Crittenden, R. G., and Playne, M. J. (1996). Production, properties and applications of food-grade oligosaccharides. *Trends in Food Science and Technology*, 7(11), 353–361. https://doi.org/10.1016/S0924-2244(96)10038-8
- Cromwell, G. L., Herkelman, K. L., and Stahly, T. S. (1993). Physical, chemical, and nutritional characteristics of distillers dried grains with solubles for chicks and pigs. *Journal of Animal Science*, *71*(3), 679–686. https://doi.org/10.2527/1993.713679x
- Cürten, C., Anders, N., Juchem, N., Ihling, N., Volkenborn, K., Knapp, A., ... Spiess, A. C. (2018). Fast automated online xylanase activity assay using HPAEC-PAD.
 Analytical and Bioanalytical Chemistry, 410(1), 57–69.
 https://doi.org/10.1007/s00216-017-0712-0
- Cyran, M., Courtin, C. M., and Delcour, J. A. (2004). Heterogeneity in the Fine Structure of Alkali-Extractable Arabinoxylans Isolated from Two Rye Flours with High and Low Breadmaking Quality and Their Coexistence with Other Cell Wall Components. *Journal of Agricultural and Food Chemistry*, *52*(9), 2671–2680. https://doi.org/10.1021/jf030550r
- Cyran, M. R., and Saulnier, L. (2012). Macromolecular structure of water-extractable arabinoxylans in endosperm and wholemeal rye breads as factor controlling their extract viscosities. *Food Chemistry*, *131*(2), 667–676. https://doi.org/10.1016/j.foodchem.2011.09.054

- De Figueiredo, F. C., Carvalho, A. F. A., Brienzo, M., Campioni, T. S., and de Oliva-Neto, P. (2017). Chemical input reduction in the arabinoxylan and lignocellulose alkaline extraction and xylooligosaccharides production. *Bioresource Technology*, 228, 164–170. https://doi.org/10.1016/j.biortech.2016.12.097
- De Vries, S., Pustjens, A. M., Kabel, M. A., Salazar-Villanea, S., Hendriks, W. H., and Gerrits, W. J. J. (2013). Processing technologies and cell wall degrading enzymes to improve nutritional value of dried distillers grain with solubles for animal feed: An in vitro digestion study. *Journal of Agricultural and Food Chemistry*, 61(37), 8821–8828. https://doi.org/10.1021/jf4019855
- Dekker, R. F. H. (1985). Biodegradation of the hemicelluloses. *Biosynthesis and Biodegradation of Wood Components*, 505–533.
- Dervilly-Pinel, G., Rimsten, L., Saulnier, L., Andersson, R., and Åman, P. (2001). Waterextractable Arabinoxylan from Pearled Flours of Wheat, Barley, Rye and Triticale. Evidence for the Presence of Ferulic Acid Dimers and their Involvement in Gel Formation. *Journal of Cereal Science*, *34*(2), 207–214. https://doi.org/10.1006/jcrs.2001.0392
- Dervilly-Pinel, G., Thibault, J.-F. F., and Saulnier, L. (2001). Experimental evidence for a semi-flexible conformation for arabinoxylans. *Carbohydrate Research*, 330(3), 365–372. https://doi.org/10.1016/S0008-6215(00)00300-1
- Dervilly, G., Saulnier, L., Roger, P., and Thibault, J.-F. (2000). Isolation of Homogeneous Fractions from Wheat Water-Soluble Arabinoxylans. Influence of the Structure on Their Macromolecular Characteristics. *Journal of Agricultural and Food Chemistry*, 48(2), 270–278. https://doi.org/10.1021/jf990222k
- Deshpande, V., Lachke, A., Mishra, C., Keskar, S., and Rao, M. (1986). Mode of action and properties of xylanase and β-Xylosidase from Neurospora crassa.
 Biotechnology and Bioengineering, 28(12), 1832–1837.
 https://doi.org/10.1002/bit.260281210
- Dionex Corporation. (2004). Product Manual (CARBOPAC PA200 Column). Dionex USA, Document N(Revision 01), 1–41. Retrieved from https://tools.thermofisher.com/content/sfs/manuals/38510-31992-01_CP_PA200_V21.pdf
- Diplock, A. T., and Fern, E. B. (1999). Scientific Concepts of Functional Foods in Europe: Consensus Document. *British Journal of Nutrition*, *81*(01), i–ii. https://doi.org/10.1079/BJN19990002
- Döring, C., Jekle, M., and Becker, T. (2015). Technological and Analytical Methods for Arabinoxylan Quantification from Cereals. *Critical Reviews in Food Science and Nutrition*, *56*(6), 999–1011. https://doi.org/10.1080/10408398.2012.749207
- Dornez, E., Gebruers, K., Delcour, J. A., and Courtin, C. M. (2009). Grain-associated xylanases: occurrence, variability, and implications for cereal processing. *Trends in Food Science and Technology*, *20*(11–12), 495–510. https://doi.org/10.1016/j.tifs.2009.05.004
- Du, C., Campbell, G. M. G. M. M., Misailidis, N., Mateos-Salvador, F., Sadhukhan, J., Mustafa, M., and Weightman, R. M. M. (2009). Evaluating the feasibility of
commercial arabinoxylan production in the context of a wheat biorefinery principally producing ethanol. Part 1. Experimental studies of arabinoxylan extraction from wheat bran. *Chemical Engineering Research and Design*, *87*(9), 1232–1238. https://doi.org/10.1016/j.cherd.2008.12.027

- Ebringerová, A., and Heinze, T. (2000). Xylan and xylan derivatives Biopolymers with valuable properties, 1: Naturally occurring xylans structures, isolation procedures and properties. *Macromolecular Rapid Communications*, *21*(9), 542–556. https://doi.org/10.1002/1521-3927(20000601)21:9<542::AID-MARC542>3.0.CO;2-7
- Ebringerova, A., and Hromadkova, Z. (1997). The effect of ultrasound on the structure and properties of the water-soluble corn hull heteroxylan. *Ultrasonics Sonochemistry*, 4(4), 305–309. https://doi.org/10.1016/S1350-4177(97)00037-0
- Ebringerova, A., and Hromadkova, Z. (2002). Effect of ultrasound on the extractibility of corn bran hemicelluloses. *Ultrasonics Sonochemistry*, *9*(4), 225–229. https://doi.org/10.1016/S1350-4177(01)00124-9
- Ebringerová, A., and Hromádková, Z. (2010). An overview on the application of ultrasound in extraction, separation and purification of plant polysaccharides. *Central European Journal of Chemistry*. https://doi.org/10.2478/s11532-010-0006-2
- Elmekawy, A., Diels, L., De Wever, H., and Pant, D. (2013, August 20). Valorization of cereal based biorefinery byproducts: Reality and expectations. *Environmental Science and Technology*. American Chemical Society. https://doi.org/10.1021/es402395g
- Englyst, H. N., Quigley, M. E., and Hudson, G. J. (1994). Determination of Dietary Fibre as Non-starch Polysaccharides With Gas-Liquid Chromatographic, Highperformance Liquid Chromatographic or Spectrophotometric Measurement of Constituent Sugars. *Analyst*, *11*. Retrieved from file:///Users/Admin/Desktop/Second year report/Englyst paper.pdf
- Erum, A., Bashir, S., and Saghir, S. (2015). Modified and unmodified arabinoxylans from Plantago ovata husk: Novel excipients with antimicrobial potential. *Bangladesh Journal of Pharmacology*, 10(4), 765. https://doi.org/10.3329/bjp.v10i4.23788
- Erum, A., Bashir, S., Saghir, S., and Sarfraz, R. M. (2015). Arabinoxylan from Plantago ovate (Husk) a novel binder and superdisintegrant. *Dhaka University Journal of Pharmaceutical Sciences*, 13(2), 133–141. https://doi.org/10.3329/dujps.v13i2.21891
- Escarnot, E., Aguedo, M., and Paquot, M. (2011). Characterization of hemicellulosic fractions from spelt hull extracted by different methods. *Carbohydrate Polymers*, *85*(2), 419–428. https://doi.org/10.1016/j.carbpol.2011.03.005
- Escarnot, E., Aguedo, M., and Paquot, M. (2012). Enzymatic hydrolysis of arabinoxylans from spelt bran and hull. *Journal of Cereal Science*, 55(2), 243–253. https://doi.org/10.1016/j.jcs.2011.12.009
- Falck, P., Aronsson, A., Grey, C., Stålbrand, H., Nordberg Karlsson, E., and Adlercreutz,

P. (2015). Production of arabinoxylan-oligosaccharide mixtures of varying composition from rye bran by a combination of process conditions and type of xylanase. *Bioresource Technology*, *174*, 118–125. https://doi.org/10.1016/j.biortech.2014.09.139

- Falck, P., Linares-Pastén, J. A., Karlsson, E. N., and Adlercreutz, P. (2018). Arabinoxylanase from glycoside hydrolase family 5 is a selective enzyme for production of specific arabinoxylooligosaccharides. *Food Chemistry*, 242, 579– 584. https://doi.org/10.1016/j.foodchem.2017.09.048
- Falck, P., Precha-Atsawanan, S., Grey, C., Immerzeel, P., Staišlbrand, H., Adlercreutz, P., and Nordberg Karlsson, E. (2013). Xylooligosaccharides from hardwood and cereal xylans produced by a thermostable xylanase as carbon sources for lactobacillus brevis and bifidobacterium adolescentis. *Journal of Agricultural and Food Chemistry*, 61(30), 7333–7340. https://doi.org/10.1021/jf401249g
- Farelo, F., Lopes, A. M. C., and Ferra, M. I. (2004). Solubilities of sodium chloride and potassium chloride in water + ethanol mixtures from (298 to 323) K. *Journal of Chemical and Engineering Data*, 49(6), 1782–1788. https://doi.org/10.1021/je049782s
- Faryar, R., Linares-Pastén, J. A., Immerzeel, P., Mamo, G., Andersson, M., Stålbrand, H., ... Karlsson, E. N. (2015). Production of prebiotic xylooligosaccharides from alkaline extracted wheat straw using the K80R-variant of a thermostable alkalitolerant xylanase. *Food and Bioproducts Processing*, *93*, 1–10. https://doi.org/10.1016/j.fbp.2014.11.004
- Faulds, C. B., and Williamson, G. (1995). Release of ferulic acid from wheat bran by a ferulic acid esterase (FAE-III) from Aspergillus niger. *Applied Microbiology and Biotechnology*, 43(6), 1082–1087.
- Faulds, C. B., Zanichelli, D., Crepin, V. F., Connerton, I. F., Juge, N., Bhat, M. K., and Waldron, K. W. (2003). Specificity of feruloyl esterases for water-extractable and water-unextractable feruloylated polysaccharides: influence of xylanase. *Journal* of Cereal Science, 38(3), 281–288. https://doi.org/10.1016/S0733-5210(03)00029-8
- Faulds, C., Sancho, A., and Bartolomé, B. (2002). Mono-and dimeric ferulic acid release from brewer's spent grain by fungal feruloyl esterases. *Applied Microbiology and Biotechnology*, 60(4), 489–494.
- Feighner, S. D., and Dashkevicz, M. P. (1988). Effect of dietary carbohydrates on bacterial cholyltaurine hydrolase in poultry intestinal homogenates. *Applied and Environmental Microbiology*, 54(2), 337–342. Retrieved from http://www.ncbi.nlm.nih.gov/pubmed/3355130

Fengler, A. I., and Marquardt, R. R. (1988). Water soluble pentosans from rye: II. Effects on rate of dialysis and on the retention of nutrients by the chick. *Cereal Chemistry*, 65(4), 298–302. Retrieved from http://www.aaccnet.org/publications/cc/backissues/1988/Documents/65_298.p df

Figueroa-Espinoza, M.-C., Poulsen, C., Borch Søe, J., Zargahi, M. R., and Rouau, X.

(2004). Enzymatic Solubilization of Arabinoxylans from Native, Extruded, and High-Shear-Treated Rye Bran by Different Endo-xylanases and Other Hydrolyzing Enzymes. *Journal of Agricultural and Food Chemistry*, *52*(13), 4240–4249. https://doi.org/10.1021/jf034809h

- Figueroa-Espinoza, M. C., and Rouau, X. (1998). Oxidative Cross-Linking of Pentosans by a Fungal Laccase and Horseradish Peroxidase: Mechanism of Linkage Between Feruloylated Arabinoxylans 1. *Cereal Chemistry*, *75*(2), 259–265.
- Fincher, G. B., and Stone, B. A. (1986). Cell walls and their components in cereal grain technology. *Advances in Cereal Science and Technology (USA)*.
- Fujikawa, S., Okazaki, M., and Matsumoto, N. (1991). Effect of xylooligosaccharide on growth of intestinal bacteria and putrefaction products. In *Nippon Eiyo Shokuryo Gakkaishi* (Vol. 44, pp. 37–40). Japan Nutrition and Food Society. https://doi.org/10.4327/jsnfs.44.37
- Fukuda, S., Toh, H., Hase, K., Oshima, K., Nakanishi, Y., Yoshimura, K., ... Ohno, H. (2011). Bifidobacteria can protect from enteropathogenic infection through production of acetate. *Nature*, 469(7331), 543–549. https://doi.org/10.1038/nature09646
- Fuller, R. (1989). Probiotics in man and animals. *Journal of Applied Bacteriology*, *66*(5), 365–378. https://doi.org/10.1111/j.1365-2672.1989.tb05105.x
- Garcia, A. L., Otto, B., Reich, S.-C., Weickert, M. O., Steiniger, J., Machowetz, A., ... Koebnick, C. (2007). Arabinoxylan consumption decreases postprandial serum glucose, serum insulin and plasma total ghrelin response in subjects with impaired glucose tolerance. *European Journal of Clinical Nutrition*, *61*(3), 334– 341. https://doi.org/10.1038/sj.ejcn.1602525
- Gauch, R., Leuenberger, U., and Baumgartner, E. (1979). Quantitative determination of mono-, di- and trisaccharides by thin-layer chromatography. *Journal of Chromatography A*, 174(1), 195–200. https://doi.org/10.1016/S0021-9673(00)87050-8
- Geraylou, Z., Souffreau, C., Rurangwa, E., De Meester, L., Courtin, C. M., Delcour, J. A.,
 ... Ollevier, F. (2013). Effects of dietary arabinoxylan-oligosaccharides (AXOS) and
 endogenous probiotics on the growth performance, non-specific immunity and
 gut microbiota of juvenile Siberian sturgeon (Acipenser baerii). *Fish and Shellfish Immunology*, 35(3), 766–775. https://doi.org/10.1016/j.fsi.2013.06.014
- Gibeaut, D. M., and Carpita, N. C. (1991). Tracing Cell Wall Biogenesis in Intact Cells and Plants : Selective Turnover and Alteration of Soluble and Cell Wall Polysaccharides in Grasses. *Plant Physiology*, 97(2), 551–561. https://doi.org/10.1104/pp.97.2.551
- Gibson, G. R., and Roberfroid, M. B. (1995). Dietary modulation of the human colonic microbiota: introducing the concept of prebiotics. *The Journal of Nutrition*, *125*(6), 1401–1412. https://doi.org/10.1079/NRR200479
- Gilbert, H. J., and Hazlewood, G. P. (1993). Bacterial cellulases and xylanases. *Journal* of General Microbiology, 139(2), 187–194. https://doi.org/10.1099/00221287-139-2-187

- Girhammar, U., and Nair, B. M. (1992). Certain Physical-Properties of Water-Soluble Nonstarch Polysaccharides From Wheat, Rye, Triticale, Barley and Oats. *Food Hydrocolloids*, 6(4), 329–343.
- Gobinath, D., Madhu, A. N., Prashant, G., Srinivasan, K., and Prapulla, S. G. (2010).
 Beneficial effect of xylo-oligosaccharides and fructo-oligosaccharides in streptozotocin-induced diabetic rats. *British Journal of Nutrition*, 104(1), 40–47.
 https://doi.org/10.1017/S0007114510000243
- Gowdhaman, D., and Ponnusami, V. (2015). Production and optimization of xylooligosaccharides from corncob by Bacillus aerophilus KGJ2 xylanase and its antioxidant potential. *International Journal of Biological Macromolecules*, 79, 595–600. https://doi.org/10.1016/j.ijbiomac.2015.05.046
- Grahn, M., Azar, C., and Lindgren, K. (2009). The role of biofuels for transportation in CO2emission reduction scenarios with global versus regional carbon caps. *Biomass and Bioenergy*, 33(3), 360–371. https://doi.org/10.1016/j.biombioe.2008.08.019
- Gruppen, H., Hamer, R. J., and Voragen, A. G. J. (1991). Barium hydroxide as a tool to extract pure arabinoxylans from water-insoluble cell wall material of wheat flour. *Journal of Cereal Science*, 13(3), 275–290. https://doi.org/10.1016/S0733-5210(09)80006-4
- Gruppen, H., Hamer, R. J., and Voragen, A. G. J. (1992). Water-unextractable cell wall material from wheat flour. 1. Extraction of polymers with alkali. *Journal of Cereal Science*, *16*(1), 41–51. https://doi.org/10.1016/S0733-5210(09)80078-7
- Gruppen, H., Hoffmaann, R. A., Kormelink, F. J. M., Voragen, A. G. J., Kamerlin, J. P., and Vliegenthart, J. F. G. (1992). Characterisation by 1H NMR spectroscopy of enzymically derived oligosaccharides from alkali-extractable wheat-flour arabinoxylan. *Carbohydrate Research*, 233, 45–64.
- Gruppen, H., Kormelink, F. J. M., and Voragen, A. G. J. (1993). Water-unextractable Cell Wall Material from Wheat Flour. 3. A Structural Model for Arabinoxylans. *Journal* of Cereal Science, 18(2), 111–128. https://doi.org/10.1006/jcrs.1993.1040
- Gullón, B., Gullón, P., Tavaria, F., Pintado, M., Gomes, A. M., Alonso, J. L., and Parajó, J. C. (2014). Structural features and assessment of prebiotic activity of refined arabinoxylooligosaccharides from wheat bran. *Journal of Functional Foods*, *6*, 438–449. https://doi.org/10.1016/J.JFF.2013.11.010
- Gullón, P., Moura, P., Esteves, M. P., Girio, F. M., Domínguez, H., and Parajó, J. C. (2008a). Assessment on the Fermentability of Xylooligosaccharides from Rice Husks by Probiotic Bacteria. *Journal of Agricultural and Food Chemistry*, 56(16), 7482–7487. https://doi.org/10.1021/jf800715b
- Gullón, P., Moura, P., Esteves, M. P., Girio, F. M., Domínguez, H., and Parajó, J. C. (2008b). Assessment on the Fermentability of Xylooligosaccharides from Rice Husks by Probiotic Bacteria. *Journal of Agricultural and Food Chemistry*, 56(16), 7482–7487. https://doi.org/10.1021/jf800715b
- Guy, R. C. E., and Sarabjit, S. S. (2003). Comparison of effects of xylanases with fungal amylases in five flour types. *Recent Advances in Enzymes in Grain Processing.*

Laboratory of Food Chemistry, Katholieke Universiteit Leuven, Leuven, 235–239.

- Hamit-Haggar, M. (2012). Greenhouse gas emissions, energy consumption and economic growth: A panel cointegration analysis from Canadian industrial sector perspective. *Energy Economics*, 34(1), 358–364. https://doi.org/10.1016/J.ENECO.2011.06.005
- Hansen, H. B., Rasmussen, C. V, Bach Knudsen, K. E., and Hansen, Å. (2003). Effects of genotype and harvest year on content and composition of dietary fibre in rye (Secale cereale L) grain. *Journal of the Science of Food and Agriculture*, 83(1), 76–85. https://doi.org/10.1002/jsfa.1284
- Haros, M., Rosell, C., and Benedito, C. (2002). Effect of different carbohydrases on fresh bread texture and bread staling. *European Food Research and Technology*, 215(5), 425–430. https://doi.org/10.1007/s00217-002-0580-4
- Harris, A. D., and Ramalingam, C. (2010). Xylanases and its Application in Food Industry: A Review. *Journal of Experimental Sciences*, 1(7), 1–11. Retrieved from www.jexpsciences.com
- Harris, P. J., Henry, R. J., Blakeney, A. B., and Stone, B. A. (1984). An improved procedure for the methylation analysis of oligosaccharides and polysaccharides. *Carbohydrate Research*, *127*(1), 59–73. https://doi.org/10.1016/0008-6215(84)85106-X
- Henrissat, B. (1991). A classification of glycosyl hydrolases based on amino acid sequence similarities. *Biochemical Journal*, *280*(2), 309–316. https://doi.org/10.1042/bj2800309
- Henrissat, B., Claeyssens, M., Tomme, P., Lemesle, L., and Mornon, J.-P. (1989). Cellulase families revealed by hydrophobic cluster analysi. *Gene*, *81*(1), 83–95.
- Hernández, O., Ruiz-Matute, A. I., Olano, A., Moreno, F. J., and Sanz, M. L. (2009). Comparison of fractionation techniques to obtain prebiotic galactooligosaccharides. *International Dairy Journal*, 19(9), 531–536. https://doi.org/https://doi.org/10.1016/j.idairyj.2009.03.002
- Ho, A. L., Carvalheiro, F., Duarte, L. C., Roseiro, L. B., Charalampopoulos, D., and Rastall, R. A. (2014). Production and purification of xylooligosaccharides from oil palm empty fruit bunch fibre by a non-isothermal process. *Bioresource Technology*, 152, 526–529. https://doi.org/10.1016/j.biortech.2013.10.114
- Hoffmann, R. A., Geijtenbeek, T., Kamerling, J. P., and Vliegenthart, J. F. G. (1992). 1H-N.m.r. study of enzymically generated wheat-endosperm arabinoxylan oligosaccharides: structures of hepta- to tetradeca-saccharides containing two or three branched xylose residues. *Carbohydrate Research*, 223(C), 19–44. https://doi.org/10.1016/0008-6215(92)80003-J
- Hoffmann, W. F., and Gortner, R. A. (1927). Wheat Flour with special reference to the globulin, albumin and proteose fractions. *Cereal Chemistry*, *4*, 221–229.
- Hollmann, J., and Lindhauer, M. (2005). Pilot-scale isolation of glucuronoarabinoxylans from wheat bran. *Carbohydrate Polymers*, *59*(2), 225–230. https://doi.org/10.1016/j.carbpol.2004.09.015

- Honců, I., Sluková, M., Vaculová, K., Sedláčková, I., Wiege, B., and Fehling, E. (2016). The effects of extrusion on the content and properties of dietary fibre components in various barley cultivars. *Journal of Cereal Science*, 68, 132–139. https://doi.org/10.1016/j.jcs.2016.01.012
- Hoppe-Seyler, E. (1889). Hemicellulose degrading microorganisms from river mud. *Physiologische Chemie*, *3*, 65–101.
- Hrmová, M., Biely, P., Vršanská, M., and Petráková, E. (1984). Induction of celluloseand xylan-degrading enzyme complex in the yeast Trichosporon cutaneum. Archives of Microbiology, 138(4), 371–376. https://doi.org/10.1007/BF00410906
- Hromádková, Z., Paulsen, B. S., Polovka, M., Košťálová, Z., and Ebringerová, A. (2013). Structural features of two heteroxylan polysaccharide fractions from wheat bran with anti-complementary and antioxidant activities. *Carbohydrate Polymers*, 93(1), 22–30. https://doi.org/10.1016/J.CARBPOL.2012.05.021
- Huber, G. W., Iborra, S., and Corma, A. (2006). Synthesis of transportation fuels from biomass: Chemistry, catalysts, and engineering. *Chemical Reviews*, 106(9), 4044– 4098. https://doi.org/10.1021/cr068360d
- Hughes, S. A., Shewry, P. R., Li, L., Gibson, G. R., Sanz, M. L., and Rastall, R. A. (2007). In Vitro Fermentation by Human Fecal Microflora of Wheat Arabinoxylans. *Journal* of Agricultural and Food Chemistry, 55(11), 4589–4595. https://doi.org/10.1021/jf070293g
- Hutkins, R. W., Krumbeck, J. A., Bindels, L. B., Cani, P. D., Fahey, G., Goh, Y. J., ...
 Sanders, M. E. (2016). Prebiotics: why definitions matter. *Current Opinion in Biotechnology*, *37*, 1–7. https://doi.org/10.1016/j.copbio.2015.09.001
- ICH Q2(R1). (1994). VALIDATION OF ANALYTICAL PROCEDURES: TEXT AND METHODOLOGY Q2(R1). Retrieved from https://www.ich.org/fileadmin/Public_Web_Site/ICH_Products/Guidelines/Qualit y/Q2_R1/Step4/Q2_R1__Guideline.pdf
- Imjongjairak, S., Jommuengbout, P., Karpilanondh, P., Katsuzaki, H., Sakka, M., Kimura, T., ... Sakka, K. (2015). Paenibacillus curdlanolyticus B-6 xylanase Xyn10C capable of producing a doubly arabinose-substituted xylose, α-l-Araf-(1→2)-[α-l-Araf-(1→3)]-d-Xylp, from rye arabinoxylan. *Enzyme and Microbial Technology*, 72, 1–9. https://doi.org/10.1016/j.enzmictec.2015.02.002
- Immerzeel, P., Falck, P., Galbe, M., Adlercreutz, P., Nordberg Karlsson, E., and Stålbrand, H. (2014). Extraction of water-soluble xylan from wheat bran and utilization of enzymatically produced xylooligosaccharides by Lactobacillus, Bifidobacterium and Weissella spp. *LWT - Food Science and Technology*, 56(2), 321–327. https://doi.org/10.1016/J.LWT.2013.12.013
- Iqbal, M. S., Akbar, J., Hussain, M. A., Saghir, S., and Sher, M. (2011). Evaluation of hotwater extracted arabinoxylans from ispaghula seeds as drug carriers. *Carbohydrate Polymers*, 83(3), 1218–1225. https://doi.org/10.1016/j.carbpol.2010.09.024
- IRENA. (2017). Synergies between Renewable Energy and Energy Efficiency: a working paper based on REmap. Retrieved from www.irena.org/remap

- Izydorczyk, M., Biliaderis, C. G., and Bushuk, W. (1991). Comparison of the structure and composition of water-soluble pentosans from different wheat varieties. *Cereal Chemistry (USA), 68*(2), 139–144. Retrieved from http://www.aaccnet.org/publications/cc/backissues/1991/documents/68_139.pd f
- Izydorczyk, M. S., and Biliaderis, C. G. (1992). Effect of molecular size on physical properties of wheat arabinoxylan. *Journal of Agricultural and Food Chemistry*, 40(4), 561–568. https://doi.org/10.1021/jf00016a006
- Izydorczyk, M. S., and Biliaderis, C. G. (1995). Cereal arabinoxylans: advances in structure and physicochemical properties. *Carbohydrate Polymers*, 28(1), 33–48. https://doi.org/10.1016/0144-8617(95)00077-1
- Izydorczyk, M. S., Biliaderis, C. G., and Bushuk, W. (1990). Oxidative gelation studies of water-soluble pentosans from wheat. *Journal of Cereal Science*, *11*(2), 153–169.
- Izydorczyk, M. S., and Dexter, J. E. (2008). Barley β-glucans and arabinoxylans: Molecular structure, physicochemical properties, and uses in food products-a Review. *Food Research International*, *41*(9), 850–868. https://doi.org/10.1016/j.foodres.2008.04.001
- Jacquemin, L., Mogni, A., Zeitoun, R., Guinot, C. C., Sablayrolles, C., Saulnier, L., and Pontalier, P. Y. (2015a). Performance evaluation of a semi-industrial production process of arabinoxylans from wheat bran. *Process Biochemistry*, 50(4), 605–613. https://doi.org/10.1016/j.procbio.2015.01.015
- Jacquemin, L., Mogni, A., Zeitoun, R., Guinot, C., Sablayrolles, C., Saulnier, L., and Pontalier, P.-Y. (2015b). Comparison of different twin-screw extraction conditions for the production of arabinoxylans. *Carbohydrate Polymers*, *116*, 86–94. https://doi.org/10.1016/j.carbpol.2014.06.071
- Jacquemin, L., Zeitoun, R., Sablayrolles, C., Pontalier, P.-Y., and Rigal, L. (2012). Evaluation of the technical and environmental performances of extraction and purification processes of arabinoxylans from wheat straw and bran. *Process Biochemistry*, 47(3), 373–380. https://doi.org/10.1016/j.procbio.2011.10.025
- Jarret, G., Cozannet, P., Martinez, J., and Dourmad, J. Y. (2011). Effect of different quality wheat dried distiller's grain solubles (DDGS) in pig diets on composition of excreta and methane production from faeces and slurry. *Livestock Science*, *140*(1–3), 275–282. https://doi.org/10.1016/J.LIVSCI.2011.04.006
- Jaskari, J., Kontula, P., Siitonen, A., Jousimies-Somer, H., Mattila-Sandholm, T., and Poutanen, K. (1998). Oat β-glucan and xylan hydrolysates as selective substrates for Bifidobacterium and Lactobacillus strains. *Applied Microbiology and Biotechnology*, *49*(2), 175–181. https://doi.org/10.1007/s002530051155
- Jayapal, N., Samanta, A. K., Kolte, A. P., Senani, S., Sridhar, M., Suresh, K. P., and Sampath, K. T. (2013). Value addition to sugarcane bagasse: Xylan extraction and its process optimization for xylooligosaccharides production. *Industrial Crops and Products*, 42, 14–24. https://doi.org/10.1016/J.INDCROP.2012.05.019
- Jensen, H. G., and Björnsson, A. H. (2012). *By-products from ethanol production the forgotten part of the equation Possibilities and challenges*. Retrieved from

http://hveiti.dk/sites/default/files/files/By-products from ethanol production - the forgotten part of the equation(1).pdf

- Jiang, Z., Li, X., Yang, S., Li, L., and Tan, S. (2005). Improvement of the breadmaking quality of wheat flour by the hyperthermophilic xylanase B from Thermotoga maritima. *Food Research International*, *38*(1), 37–43. https://doi.org/10.1016/j.foodres.2004.07.007
- Jiang, Z. Q., Yang, S. Q., Tan, S. S., Li, L. T., and Li, X. T. (2005). Characterization of a xylanase from the newly isolated thermophilic Thermomyces lanuginosus CAU44 and its application in bread making. *Letters in Applied Microbiology*, 41(1), 69–76. https://doi.org/10.1111/j.1472-765X.2005.01725.x
- Josefsson, T., Lennholm, H., and Gellerstedt, G. (2002). Steam explosion of aspen wood. Characterisation of reaction products. *Holzforschung*, *56*(3), 289–297. https://doi.org/10.1515/HF.2002.047
- Jung, H. G., Buxton, D. R., Hatfield, R. D., Ralph, J., Iiyama, K., Lam, T. B. T., ... Stone, B. A. (1993). Cell Wall Biosynthesis and Its Regulation. In *Forage Cell Wall Structure* and Digestibility (Vol. accesspubl, pp. 621–683). American Society of Agronomy, Crop Science Society of America, Soil Science Society of America. https://doi.org/10.2134/1993.foragecellwall.c24
- Kabel, M. A., Kortenoeven, L., Schols, H. A., and Voragen, A. G. J. (2002). In Vitro Fermentability of Differently Substituted Xylo-oligosaccharides. *Journal of Agricultural and Food Chemistry*, 50(21), 6205–6210. https://doi.org/10.1021/jf020220r
- Khathuriya, R., Nayyar, T., Sabharwal, S., Jain, U. K., and Taneja, R. (2015). RECENT APPROACHES AND PHARMACEUTICAL APPLICATIONS OF NATURAL POLYSACCHARIDES: A REVIEW. *International Journal of Pharmaceutical Sciences and Research*, 6(12), 4904–4919. https://doi.org/http://dx.doi.org/10.1016/0958-1669(92)90020-J
- Kim, Y., Mosier, N. S., Hendrickson, R., Ezeji, T., Blaschek, H., Dien, B., ... Ladisch, M. R. (2007). Composition of corn dry-grind ethanol by-products: DDGS, wet cake, and thin stillage. https://doi.org/10.1016/j.biortech.2007.09.028
- Kimm, N. K. (2008). Economic and Environmental Aspects of Integration in Chemical Production Sites. der Universität Karlsruhe. Retrieved from https://dnb.info/989980340/34
- Kingsly, A. R. P., Ileleji, K. E., Clementson, C. L., Garcia, A., Maier, D. E., Stroshine, R. L., and Radcliff, S. (2010). The effect of process variables during drying on the physical and chemical characteristics of corn dried distillers grains with solubles (DDGS) - Plant scale experiments. *Bioresource Technology*, 101(1), 193–199. https://doi.org/10.1016/j.biortech.2009.07.070
- Klopfenstein, T. J., Erickson, G. E., and Bremer, V. R. (2008). BOARD-INVITED REVIEW: Use of distillers by-products in the beef cattle feeding industry1. *Journal of Animal Science*, 86(5), 1223–1231. https://doi.org/10.2527/jas.2007-0550
- Knudsen, K. E. B., and Lærke, H. N. (2010). REVIEW: Rye Arabinoxylans: Molecular Structure, Physicochemical Properties and Physiological Effects in the

Gastrointestinal Tract. *Cereal Chemistry*, 87(4), 353–362. https://doi.org/10.1094/CCHEM-87-4-0353

- Ko, C.-H., Tsai, C.-H., Tu, J., Yang, B.-Y., Hsieh, D.-L., Jane, W.-N., and Shih, T.-L. (2011). Identification of Paenibacillus sp. 2S-6 and application of its xylanase on biobleaching. *International Biodeterioration and Biodegradation*, 65(2), 334–339. https://doi.org/10.1016/J.IBIOD.2010.12.006
- Kosik, O., Powers, S. J., Chatzifragkou, A., Prabhakumari, P. C., Charalampopoulos, D., Hess, L., ... Lovegrove, A. (2017). Changes in the arabinoxylan fraction of wheat grain during alcohol production. *Food Chemistry*, 221, 1754–1762. https://doi.org/10.1016/j.foodchem.2016.10.109
- Krawczyk, H., Persson, T., Andersson, A., Jönsson, A.-S., Avellar, B. K., Glasser, W. G., ...
 Holmbom, B. (2008). Isolation of hemicelluloses from barley husks. *Food and Bioproducts Processing*, 86(1), 31–36. https://doi.org/10.1016/j.fbp.2007.10.018
- Kroon, P. A., Garcia-Conesa, M. T., Fillingham, I. J., Hazlewood, G. P., and Williamson, G. (1999). Release of ferulic acid dehydrodimers from plant cell walls by feruloyl esterases. *Journal of the Science of Food and Agriculture*, *79*(3), 428–434.
- Kulkarni, N., Shendye, A., and Rao, M. (1999). Molecular and biotechnological aspects of xylanases. *FEMS Microbiology Reviews*, 23(4), 411–456. https://doi.org/10.1016/S0168-6445(99)00006-6
- Kulp, K. (1968). Pentosans of wheat endosperm. *Cereal Science Today*, 45(339), 414–458.
- Lafond, M., Tauzin, A., Desseaux, V., Bonnin, E., Ajandouz, E. H., and Giardina, T. (2011). GH10 xylanase D from Penicillium funiculosum: Biochemical studies and xylooligosaccharide production. *Microbial Cell Factories*, 10(1), 20. https://doi.org/10.1186/1475-2859-10-20
- Laine, C., Kemppainen, K., Kuutti, L., Varhimo, A., Asikainen, S., Grönroos, A., ... Harlin, A. (2015). Extraction of xylan from wood pulp and brewer's spent grain. *Industrial Crops and Products*, 70, 231–237. https://doi.org/10.1016/j.indcrop.2015.03.009
- Larsson, S., Palmqvist, E., Hahn-Hägerdal, B., Tengborg, C., Stenberg, K., Zacchi, G., and Nilvebrant, N.-O. (1999). The generation of fermentation inhibitors during dilute acid hydrolysis of softwood. *Enzyme and Microbial Technology*, *24*(3), 151–159.
- Lazaridou, A., and Biliaderis, C. G. (2007). Molecular aspects of cereal β-glucan functionality: Physical properties, technological applications and physiological effects. *Journal of Cereal Science*, *46*(2), 101–118. https://doi.org/10.1016/j.jcs.2007.05.003
- le Roux, C. W., and Bloom, S. R. (2005). Peptide YY, appetite and food intake. The Proceedings of the Nutrition Society, 64(2), 213–216. https://doi.org/10.1079/PNS2005427
- Lei, Z., Shao, Y., Yin, X., Yin, D., Guo, Y., and Yuan, J. (2016). Combination of Xylanase and Debranching Enzymes Specific to Wheat Arabinoxylan Improve the Growth Performance and Gut Health of Broilers. *Journal of Agricultural and Food Chemistry*, 64(24), 4932–4942. https://doi.org/10.1021/acs.jafc.6b01272

- Li, H., Qing, Q., Kumar, R., and Wyman, C. E. (2013). Chromatographic determination of 1, 4-β-xylooligosaccharides of different chain lengths to follow xylan deconstruction in biomass conversion. *Journal of Industrial Microbiology and Biotechnology*, 40(6), 551–559. https://doi.org/10.1007/s10295-013-1254-x
- Li, W., Hu, H., Wang, Q., and Brennan, C. S. (2013). Molecular features of wheat endosperm arabinoxylan inclusion in functional bread. *Foods*, 2(2), 225–237. https://doi.org/10.3390/foods2020225
- Lilly, D. M., and Stillwell, R. H. (1965). Probiotics: Growth-promoting factors produced by microorganisms. *Science*, *147*(3659), 747–748. https://doi.org/10.1126/science.147.3659.747
- Lin, Q., Li, H., Ren, J., Deng, A., Li, W., Liu, C., and Sun, R. (2017). Production of xylooligosaccharides by microwave-induced, organic acid-catalyzed hydrolysis of different xylan-type hemicelluloses: Optimization by response surface methodology. *Carbohydrate Polymers*, 157, 214–225. https://doi.org/10.1016/j.carbpol.2016.09.091
- Linares-Pastén, J., Aronsson, A., and Karlsson, E. (2017). Structural considerations on the use of endo-xylanases for the production of prebiotic xylooligosaccharides from biomass. *Current Protein and Peptide Science*, 17(999), 1–1. https://doi.org/10.2174/1389203717666160923155209
- Liu, K. (2008). Particle size distribution of distillers dried grains with solubles (DDGS) and relationships to compositional and color properties. *Bioresource Technology*, 99(17), 8421–8428. https://doi.org/10.1016/J.BIORTECH.2008.02.060
- Liu, K. (2011). Chemical composition of distillers grains, a review. Journal of Agricultural and Food Chemistry, 59(5), 1508–1526. https://doi.org/10.1021/jf103512z
- Lopez, H. W., Levrat, M. A., Guy, C., Messager, A., Demign??, C., and R??m??sy, C. (1999). Effects of soluble corn bran arabinoxylans on cecal digestion, lipid metabolism, and mineral balance (Ca, Mg) in rats. *Journal of Nutritional Biochemistry*, 10(9), 500–509. https://doi.org/10.1016/S0955-2863(99)00036-4
- Louw, A. (2018). Clean Energy Investment Trends 2017. *Bloomberg*, 17. Retrieved from https://data.bloomberglp.com/bnef/sites/14/2018/01/BNEF-Clean-Energy-Investment-Investment-Trends-2017.pdf?elqTrackId=2e6e6b2aa1f946bca67cd74d9e20babbandelq=b1e02255d9 114ef3808f38b3ee62a765andelqaid=10316andelqat=1andelqCampaignId=
- Lu, Z. X., Walker, K. Z., Muir, J. G., and O'Dea, K. (2004). Arabinoxylan fibre improves metabolic control in people with Type II diabetes. *European Journal of Clinical Nutrition*, 58(4), 621–628. https://doi.org/10.1038/sj.ejcn.1601857
- Lynd, L. R., Larson, E., Greene, N., Laser, M., Sheehan, J., Dale, B. E., ... Wang, M. (2009). The role of biomass in America's energy future: framing the analysis. *Biofuels, Bioproducts and Biorefining*, 3(2), 113–123. https://doi.org/10.1002/bbb.134
- Lynd, L. R., Wyman, C., Laser, M., and College, D. (2005). Strategic Biorefinery Analysis : Review of Existing Biorefinery Examples October 2005 Strategic

Biorefinery Analysis : Review of Existing Biorefinery Examples, (October), 40. https://doi.org/10.1007/s40257-016-0192-6

- Macfarlane, S., Macfarlane, G. T., and Cummings, J. H. (2006, September 1). Review article: Prebiotics in the gastrointestinal tract. *Alimentary Pharmacology and Therapeutics*. Wiley/Blackwell (10.1111). https://doi.org/10.1111/j.1365-2036.2006.03042.x
- Madhukumar, M. S., and Muralikrishna, G. (2012). Fermentation of xylooligosaccharides obtained from wheat bran and Bengal gram husk by lactic acid bacteria and bifidobacteria. *Journal of Food Science and Technology*, 49(6), 745– 752. https://doi.org/10.1007/s13197-010-0226-7
- Maes, C., and Delcour, J. a. (2002). Structural Characterisation of Water-extractable and Water-unextractable Arabinoxylans in Wheat Bran. *Journal of Cereal Science*, *35*(3), 315–326. https://doi.org/10.1006/jcrs.2001.0439
- Maes, C., and Delcour, J. A. (2001). Alkaline Hydrogen Peroxide Extraction of Wheat Bran Non-starch Polysaccharides. *Journal of Cereal Science*, *34*(1), 29–35. https://doi.org/10.1006/jcrs.2001.0377
- Maes, C., Vangeneugden, B., and Delcour, J. A. (2004). Relative activity of two endoxylanases towards water-unextractable arabinoxylans in wheat bran. *Journal* of Cereal Science, 39(2), 181–186.
- Maity, S. K. (2015). Opportunities, recent trends and challenges of integrated biorefinery: Part II. *Renewable and Sustainable Energy Reviews*, 43, 1446–1466. https://doi.org/10.1016/j.rser.2014.08.075
- Mandal, A. (2015). Review on Microbial Xylanases and their Applications. *International Journal of Life Sciences*, 4(3), 178–187. Retrieved from www.crdeep.com
- Manning, T. S., and Gibson, G. R. (2004). Prebiotics. *Best Practice and Research Clinical Gastroenterology*, *18*(2), 287–298. https://doi.org/10.1016/j.bpg.2003.10.008
- Mares, D. J., and Stone, B. A. (1973). Studies on lolium multiflorum endosperm in tissue culture II. Fine structure of cells and cell walls and the development of cell walls. *Australian Journal of Biological Sciences*, 26(1), 135–150. https://doi.org/10.1071/BI9730135
- Martinez-Hernandez, E., Sadhukhan, J., and Campbell, G. M. (2013). Integration of bioethanol as an in-process material in biorefineries using mass pinch analysis. *Applied Energy*, *104*, 517–526. https://doi.org/10.1016/j.apenergy.2012.11.054
- Martinez-Hernandez, E., Tibessart, A., and Campbell, G. M. (2018). Conceptual design of integrated production of arabinoxylan products using bioethanol pinch analysis. *Food and Bioproducts Processing*, *112*, 1–8. https://doi.org/10.1016/j.fbp.2018.08.005
- Martirosyan, D. M., and Singh, J. (2015). A new definition of functional food by FFC: what makes a new definition unique? Functional Foods in Health and Disease (Vol. 5). Retrieved from http://functionalfoodscenter.net/files/105582267.pdf
- Mason, W. H. (1928). US1824221A. USA: US Grant. Retrieved from https://patents.google.com/patent/US1824221A/en

- Mathew, S., Karlsson, E. N., and Adlercreutz, P. (2017). Extraction of soluble arabinoxylan from enzymatically pretreated wheat bran and production of short xylo-oligosaccharides and arabinoxylo-oligosaccharides from arabinoxylan by glycoside hydrolase family 10 and 11 endoxylanases. *Journal of Biotechnology*, 260, 53–61. https://doi.org/10.1016/j.jbiotec.2017.09.006
- Matte, A., and Forsberg, C. W. (1992). Purification, characterization, and mode of action of endoxylanases 1 and 2 from Fibrobacter succinogenes S85. *Applied and Environmental Microbiology*, 58(1), 157–168. Retrieved from http://www.ncbi.nlm.nih.gov/pubmed/1539970
- May, K., O'Sullivan, S. E., Brameld, J. M., Masey O'Neill, H. V., Parr, T., and Wiseman, J. (2015). Xylanase supplementation in feed reduces incretin and PYY levels in piglets. *Proceedings of the Nutrition Society*, 74(OCE5), E294. https://doi.org/10.1017/S0029665115003419
- McCleary, B. V, Travel, F. R., and Cheetham, N. W. H. (1982). Preparative-scale isolation and characterisation of 6 1- α -d-galactosyl-(1 \rightarrow 4)- β -d-mannobiose and 6 2- α -d-galactosyl-(1 \rightarrow 4)- β -d-mannobiose. *Carbohydrate Research*, 104(2), 285–297.
- McGovern, P. E., Glusker, D. L., Exner, L. J., and Voigt, M. M. (1996). Neolithic resinated wine. *Nature 1996 381:6582*, *381*(6582), 480. https://doi.org/10.1038/381480a0
- Mendis, M., Leclerc, E., and Simsek, S. (2016). Arabinoxylans, gut microbiota and immunity. *Carbohydrate Polymers*, *139*, 159–166. https://doi.org/10.1016/j.carbpol.2015.11.068
- Mendis, M., and Simsek, S. (2014). Arabinoxylans and human health. *Food Hydrocolloids*, 42, 239–243. https://doi.org/10.1016/j.foodhyd.2013.07.022
- Merali, Z., Collins, S. R. A., Elliston, A., Wilson, D. R., Käsper, A., Waldron, K. W., ... Ziaul-Hussnain, S. (2015). Characterization of cell wall components of wheat bran following hydrothermal pretreatment and fractionation. *Biotechnology for Biofuels*, 8(1), 23. https://doi.org/10.1186/s13068-015-0207-1
- Mikkonen, K. S., Heikkinen, S., Soovre, A., Peura, M., Serimaa, R., Talja, R. A., ... Tenkanen, M. (2009). Films from oat spelt arabinoxylan plasticized with glycerol and sorbitol. *Journal of Applied Polymer Science*, 114(1), 457–466. https://doi.org/10.1002/app.30513
- Minjares-Fuentes, R., Femenia, A., Garau, M. C., Candelas-Cadillo, M. G., Simal, S., and Rosselló, C. (2016). Ultrasound-assisted extraction of hemicelluloses from grape pomace using response surface methodology. *Carbohydrate Polymers*, 138, 180– 191. https://doi.org/10.1016/j.carbpol.2015.11.045
- Misailidis, N., Campbell, G. M., Du, C., Sadhukhan, J., Mustafa, M., Mateos-Salvador, F., and Weightman, R. M. (2009). Evaluating the feasibility of commercial arabinoxylan production in the context of a wheat biorefinery principally producing ethanol: Part 2. Process simulation and economic analysis. *Chemical Engineering Research and Design*, *87*(9), 1239–1250.
- Mitsuishi, Y., Yamanobe, T., and Yagisawa, M. (1988). The modes of action of three xylanases from mesophilic fungus strain Y-94 on xylooligosaccharides. *Agricultural*

and Biological Chemistry, 52(4), 921–927. https://doi.org/10.1271/bbb1961.52.921

- Moore, A. M., Martinez-Munoz, I., and Hoseney, R. C. (1990). Factors affecting the oxidative gelation of wheat water-solubles. *Cereal Chem*, 67(1), 81–84.
- Moreira, L. R. S., and Filho, E. X. F. (2016, June 25). Insights into the mechanism of enzymatic hydrolysis of xylan. *Applied Microbiology and Biotechnology*. https://doi.org/10.1007/s00253-016-7555-z
- Morey, R. V., Hatfield, D. L., Sears, R., Haak, D., Tiffany, D. G., and Kaliyan, N. (2009).
 FUEL PROPERTIES OF BIOMASS FEED STREAMS AT ETHANOL PLANTS. *Applied Engineering in Agriculture*, 25(1), 57–64. Retrieved from https://s3.amazonaws.com/academia.edu.documents/43175294/Fuel_Properties _of_Biomass_Feed_Streams_20160228-1573-47rkpf.pdf?AWSAccessKeyId=AKIAIWOWYYGZ2Y53UL3AandExpires=1503486928 andSignature=ioVdD%2Bq%2EVSYWn2GyoiCDpAQs%3Dandresponsecontent-disposition=in
- Morgan, N. K., Wallace, A., Bedford, M. R., and Choct, M. (2017). Efficiency of xylanases from families 10 and 11 in production of xylo-oligosaccharides from wheat arabinoxylans. *Carbohydrate Polymers*, 167, 290–296. https://doi.org/10.1016/j.carbpol.2017.03.063
- Murashima, K., Kosugi, A., and Doi, R. H. (2003). Synergistic effects of cellulosomal xylanase and cellulases from Clostridium cellulovorans on plant cell wall degradation. *Journal of Bacteriology*, *185*(5), 1518–1524. https://doi.org/10.1128/JB.185.5.1518-1524.2003
- Naidu, D. S., Hlangothi, S. P., and John, M. J. (2018, January 1). Bio-based products from xylan: A review. *Carbohydrate Polymers*. Elsevier. https://doi.org/10.1016/j.carbpol.2017.09.064
- Neyrinck, A. M., Van Hée, V. F., Piront, N., De Backer, F., Toussaint, O., Cani, P. D., and Delzenne, N. M. (2012). Wheat-derived arabinoxylan oligosaccharides with prebiotic effect increase satietogenic gut peptides and reduce metabolic endotoxemia in diet-induced obese mice. *Nutrition and Diabetes*, 2(JANUARY), e28–e28. https://doi.org/10.1038/nutd.2011.24
- Ng, D. K. S., Tan, R. R., Foo, D. C. Y., and El-Halwagi, M. M. (2015). Process design strategies for biomass conversion systems. John Wiley and Sons, Ltd. Retrieved from https://www.wiley.com/engb/Process+Design+Strategies+for+Biomass+Conversion+Systems-p-9781118699157
- Nieto-Domínguez, M., de Eugenio, L. I., York-Durán, M. J., Rodríguez-Colinas, B., Plou, F. J., Chenoll, E., ... Jesús Martínez, M. (2017). Prebiotic effect of xylooligosaccharides produced from birchwood xylan by a novel fungal GH11 xylanase. *Food Chemistry*, 232, 105–113. https://doi.org/10.1016/J.FOODCHEM.2017.03.149
- Nigam, P. S., and Singh, A. (2011). Production of liquid biofuels from renewable resources. *Progress in Energy and Combustion Science*, *37*(1), 52–68.

https://doi.org/10.1016/j.pecs.2010.01.003

- OECD/FAO. (2015). OECD/FAO (2015), "OECD-FAO Agricultural Outlook", OECD Agriculture statistics (database),. https://doi.org/https://doi.org/10.1787/agrdata-en
- Okazaki, M., fujikawa, S., and matsumoto, N. (1990). Effect of Xylooligosaccharide on the Growth of Bifidobacteria. *Bifidobacteria and Microflora*, *9*(2), 77–86. https://doi.org/10.12938/bifidus1982.9.2_77
- Ou, J., and Sun, Z. (2014). Feruloylated oligosaccharides: Structure, metabolism and function. *Journal of Functional Foods*, *7*, 90–100. https://doi.org/10.1016/J.JFF.2013.09.028
- Oxford Dictionary. (n.d.). oligo- | Definition of oligo- in English by Oxford Dictionaries. Retrieved August 3, 2018, from https://en.oxforddictionaries.com/definition/oligo-
- Paice, M. G., Gurnagul, N., Page, D. H., and Jurasek, L. (1992). Mechanism of hemicellulose-directed prebleaching of kraft pulps. *Enzyme and Microbial Technology*, 14(4), 272–276. https://doi.org/10.1016/0141-0229(92)90150-M
- Paloheimo, M., Piironen, J., and Vehmaanperä, J. (2010). Xylanases and Cellulases as Feed Additives. In M. R. Bedford and G. G. Partridge (Eds.), *Enzymes in Farm Animal Nutrition, 2nd edition* (2ed ed., p. 329). Bodmin, UK.: CAB International CABI. https://doi.org/10.1079/9781845936747.0012
- Parker, R. B. (1974). Probiotics, the other half of the antibiotic story. *Anim Nutr Health*, 29, 4–8.
- Patel, A., Falck, P., Shah, N., Immerzeel, P., Adlercreutz, P., Stålbrand, H., ... Nordberg Karlsson, E. (2013). Evidence for xylooligosaccharide utilization in Weissella strains isolated from Indian fermented foods and vegetables. *FEMS Microbiology Letters*, 346(1), 20–28. https://doi.org/10.1111/1574-6968.12191
- Pedersen, M. B., Dalsgaard, S., Knudsen, K. E. B., Yu, S., and Lærke, H. N. (2014). Compositional profile and variation of Distillers Dried Grains with Solubles from various origins with focus on non-starch polysaccharides. *Animal Feed Science* and Technology, 197, 130–141. https://doi.org/10.1016/j.anifeedsci.2014.07.011
- Pedersen, M., Viksø-Nielsen, A., and Meyer, A. S. (2010). Monosaccharide yields and lignin removal from wheat straw in response to catalyst type and pH during mild thermal pretreatment. *Process Biochemistry*, *45*(7), 1181–1186.
- Peng, F., Peng, P., Xu, F., and Sun, R.-C. (2012). Fractional purification and bioconversion of hemicelluloses. *Biotechnology Advances*, 30(4), 879–903. https://doi.org/10.1016/j.biotechadv.2012.01.018
- Peng, F., Ren, J. L., Xu, F., Bian, J., Peng, P., and Sun, R. C. (2010). Fractional study of alkali-soluble hemicelluloses obtained by graded ethanol precipitation from sugar cane bagasse. *Journal of Agricultural and Food Chemistry*, 58(3), 1768–1776. https://doi.org/10.1021/jf9033255
- Petit-Benvegnen, M. D., Saulnier, L., and Rouau, X. (1998). Solubilization of arabinoxylans from isolated water- unextractable pentosans and wheat flour

References

doughs by cell-wall- degrading enzymes. *Cereal Chemistry*, 75(4), 551–556. Retrieved from file://localhost/Users/saulnier/Documents/BiblioLuc/1998/Petit+98.pdf

- Pettersson, D., and Åman, P. (1989). Enzyme supplementation of a poultry diet containing rye and wheat. *British Journal of Nutrition*, *62*(01), 139. https://doi.org/10.1079/BJN19890014
- Phan The, D., Debeaufort, F., Péroval, C., Despré, D., Courthaudon, J. L., and Voilley, A. (2002). Arabinoxylan–Lipid-Based Edible Films and Coatings. 3. Influence of Drying Temperature on Film Structure and Functional Properties. *Journal of Agricultural and Food Chemistry*, *50*(8), 2423–2428. https://doi.org/10.1021/jf010898r
- Phillips, S. (2017). *EU Biofuels Annual 2017*. Retrieved from https://gain.fas.usda.gov/Recent GAIN Publications/Biofuels Annual_The Hague_EU-28_6-19-2017.pdf
- Poesen, R., Evenepoel, P., de Loor, H., Delcour, J. A., Courtin, C. M., Kuypers, D., ... Meijers, B. (2016). The Influence of Prebiotic Arabinoxylan Oligosaccharides on Microbiota Derived Uremic Retention Solutes in Patients with Chronic Kidney Disease: A Randomized Controlled Trial. *PLOS ONE*, *11*(4), e0153893. https://doi.org/10.1371/journal.pone.0153893
- Polizeli, M. L. T. M., Rizzatti, A. C. S. S., Monti, R., Terenzi, H. F., Jorge, J. A., and Amorim, D. S. (2005). Xylanases from fungi: Properties and industrial applications. *Applied Microbiology and Biotechnology*, 67(5), 577–591. https://doi.org/10.1007/s00253-005-1904-7
- Porchia, A. C., Sørensen, S. O., and Scheller, H. V. (2002). Arabinoxylan biosynthesis in wheat. Characterization of arabinosyltransferase activity in Golgi membranes. *Plant Physiology*, 130(1), 432–441. https://doi.org/10.1104/pp.003400
- Prade, R. A. (1995). Xylanase: from biology to biotechnology. *Biotechnology and Genetic Engineering Reviews*, 13(1), 100–131. https://doi.org/10.1080/02648725.1996.10647925
- Preece, I. A., and Hobkirk, R. (1953). Non-Starchy Polysaccharides of Cereal Grains Iii. Higher Molecular Gums of Common Cereals. *Journal of the Institute of Brewing*, *59*(5), 385–392. https://doi.org/10.1002/j.2050-0416.1953.tb02733.x
- Preece, I. A., and MacDougall, M. (1958). ENZYMIC DEGRADATION OF CEREAL HEMIGELLULOSES II. PATTERN OF PENTOSAN DEGRADATION. *Journal of the Institute of Brewing*, *64*(6), 489–500.
- Puls, J. (1992). Substrate analysis of forest and agricultural wastes. *BIOTECHNOLOGY IN AGRICULTURE*, (1), 13–32.
- Raja, D., Poornima, D., Bhaskar, K., and Saravanan, M. (2014). Ispaghula husk-based extended release tablets of diclofenac sodium: formulation, evaluation and in vitro release studies. *Tropical Journal of Pharmaceutical Research*, 13(2), 205– 210.

Ranganathan, S., Macdonald, D. G., and Bakhshi, N. N. (1985). Kinetic studies of wheat

straw hydrolysis using sulphuric acid. *The Canadian Journal of Chemical Engineering*, 63(5), 840–844. https://doi.org/10.1002/cjce.5450630521

- RASCO, B. A., DONG, F. M., HASHISAKA, A. E., GAZZAZ, S. S., DOWNEY, S. E., and SAN BUENAVENTURA, M. L. (1987). Chemical Composition of Distillers' Dried Grains with Solubles (DDGS) from Soft White Wheat, Hard Red Wheat and Corn. *Journal* of Food Science, 52(1), 236–237. https://doi.org/10.1111/j.1365-2621.1987.tb14019.x
- Rastall, R. A., and Gibson, G. R. (2015). Recent developments in prebiotics to selectively impact beneficial microbes and promote intestinal health. *Current Opinion in Biotechnology*, 32, 42–46. https://doi.org/10.1016/j.copbio.2014.11.002
- Rattan, O., Izydorczyk, M. S., and Biliaderis, C. G. (1994). Structure and Rheological Behaviour of Arabinoxylans from Canadian Bread Wheat Flours. *LWT - Food Science and Technology*, *27*(6), 550–555. https://doi.org/10.1006/fstl.1994.1108
- Reddy, S. S., and Krishnan, C. (2016). Production of high-pure xylooligosaccharides from sugarcane bagasse using crude β-xylosidase-free xylanase of Bacillus subtilis KCX006 and their bifidogenic function. *LWT - Food Science and Technology*, 65, 237–245. https://doi.org/10.1016/j.lwt.2015.08.013
- Reis, S. F., Coelho, E., Coimbra, M. A., and Abu-Ghannam, N. (2015). Improved efficiency of brewer's spent grain arabinoxylans by ultrasound-assisted extraction. *Ultrasonics Sonochemistry*, 24, 155–164. https://doi.org/10.1016/j.ultsonch.2014.10.010
- Rocklin, R. D., and Pohl, C. A. (1983). Determination of Carbohydrates by Anion Exchange Chromatography with Pulsed Amperometric Detection. *Journal of Liquid Chromatography*, 6(9), 1577–1590. https://doi.org/10.1080/01483918308064876
- Rose, D. J., and Inglett, G. E. (2010). Production of feruloylated arabinoxylooligosaccharides from maize (Zea mays) bran by microwave-assisted autohydrolysis. *Food Chemistry*, 119(4), 1613–1618. https://doi.org/10.1016/j.foodchem.2009.09.053
- Rose, D. J., Patterson, J. A., and Hamaker, B. R. (2010). Structural differences among alkali-soluble arabinoxylans from Maize (Zea mays), Rice (Oryza sativa), and Wheat (Triticum aestivum) brans influence human fecal fermentation profiles. *Journal of Agricultural and Food Chemistry*, *58*(1), 493–499. https://doi.org/10.1021/jf9020416
- Rumpagaporn, P., Reuhs, B. L., Kaur, A., Patterson, J. A., Keshavarzian, A., and Hamaker, B. R. (2015). Structural features of soluble cereal arabinoxylan fibers associated with a slow rate of in vitro fermentation by human fecal microbiota. *Carbohydrate Polymers*, 130, 191–197. https://doi.org/10.1016/j.carbpol.2015.04.041
- Rydlund, A., and Dahlman, O. (1997). Oligosaccharides obtained by enzymatic hydrolysis of birch kraft pulp xylan: Analysis by capillary zone electrophoresis and mass spectrometry. *Carbohydrate Research*, *300*(2), 95–102.

https://doi.org/10.1016/S0008-6215(97)00038-4

- Sabiha-Hanim, S., Noor, M. A. M., and Rosma, A. (2011). Effect of autohydrolysis and enzymatic treatment on oil palm (Elaeis guineensis Jacq.) frond fibres for xylose and xylooligosaccharides production. *Bioresource Technology*, 102(2), 1234–1239. https://doi.org/10.1016/J.BIORTECH.2010.08.017
- Sadosky, P., Schwarz, P. B., and Horsley, R. D. (2002). Effect of Arabinoxylans, β-Glucans, and Dextrins on the Viscosity and Membrane Filterability of a Beer Model Solution. *Journal of the American Society of Brewing Chemists*, *60*(4), 153– 162. https://doi.org/10.1094/ASBCJ-60-0153
- Saeed, F., Pasha, I., Anjum, F. M., and Sultan, M. T. (2011). Arabinoxylans and arabinogalactans: a comprehensive treatise. *Critical Reviews in Food Science and Nutrition*, *51*(5), 467–476. https://doi.org/10.1080/10408391003681418
- Saini, S., sharma, M., and kumar, A. (2018). Global Warming, Climate Change and Greenhouse Gas Mitigation. *Biofuels: Greenhouse Gas Mitigation and Global Warming*, (April), 1–16. https://doi.org/10.1007/978-81-322-3763-1_1
- Saleem, M., Tabassum, M. R., Yasmin, R., and Imran, M. (2009). Potential of xylanase from thermophilic Bacillus sp. XTR-10 in biobleaching of wood kraft pulp. *International Biodeterioration and Biodegradation*, 63(8), 1119–1124. https://doi.org/10.1016/J.IBIOD.2009.099.009
- Salih, M. E., Classen, H. L., and Campbell, G. L. (1991). Response of chickens fed on hull-less barley to dietary β-glucanase at different ages. *Animal Feed Science and Technology*, 33(1–2), 139–149. https://doi.org/10.1016/0377-8401(91)90052-T
- Samanta, A. K. K., Jayapal, N., Jayaram, C., Roy, S., Kolte, A. P. P., Senani, S., and Sridhar, M. (2015). Xylooligosaccharides as prebiotics from agricultural byproducts: Production and applications. *Bioactive Carbohydrates and Dietary Fibre*, 5(1), 62–71. https://doi.org/10.1016/j.bcdf.2014.12.003
- Samuel, D. (1996). Investigation of ancient Egyptian baking and brewing methods by correlative microscopy. *Science*, *273*(5274), 488–490. https://doi.org/10.1126/science.273.5274.488
- Sandford, P. A., and Conrad, H. E. (1966). The Structure of the Aerobacter aerogenes A3(S1) Polysaccharide. I. A Reexamination Using Improved Procedures for Methylation Analysis *. *Biochemistry*, 5(5), 1508–1517. https://doi.org/10.1021/bi00869a009
- Santiago, R., Butrón, A., Reid, L. M., Arnason, J. T., Sandoya, G., Souto, X. C., and Malvar, R. A. (2006). Diferulate content of maize sheaths is associated with resistance to the Mediterranean corn borer Sesamia nonagrioides (Lepidoptera: Noctuidae). *Journal of Agricultural and Food Chemistry*, 54(24), 9140–9144.
- Santiago, R., and Malvar, R. A. (2010). Role of Dehydrodiferulates in Maize Resistance to Pests and Diseases. *International Journal of Molecular Sciences*, *11*(2), 691– 703. https://doi.org/10.3390/ijms11020691
- Sarker, D. K., Wilde, P. J., and Clark, D. C. (1998). Enhancement of protein foam stability by formation of wheat arabinoxylan-protein crosslinks. *Cereal Chemistry*,

75(4), 493-499. https://doi.org/10.1094/CCHEM.1998.75.4.493

- Saulnier, L., Sado, P.-E., Branlard, G., Charmet, G., and Guillon, F. (2007). Wheat arabinoxylans: Exploiting variation in amount and composition to develop enhanced varieties. *Journal of Cereal Science*, *46*(3), 261–281. https://doi.org/10.1016/j.jcs.2007.06.014
- Scheller, H. V., and Ulvskov, P. (2010). Hemicelluloses. *Annual Review of Plant Biology*, 61(1), 263–289. https://doi.org/10.1146/annurev-arplant-042809-112315
- Schley, P. D., and Field, C. J. (2002). The immune-enhancing effects of dietary fibres and prebiotics. *British Journal of Nutrition*, *87*(S2), S221. https://doi.org/10.1079/BJN/2002541
- Schnaak, W. (1984). Food Constituents and Food Residues. Their Chromatographic Determination. (J. F Lawrence, Ed.), Food / Nahrung (Vol. 30). New York und Basel: Marcel Dekker. Inc. https://doi.org/10.1002/food.19860300305
- Schooneveld-Bergmans, M. E. F., Beldman, G., and Voragen, A. G. J. (1999). Structural Features of (Glucurono)Arabinoxylans Extracted from Wheat Bran by Barium Hydroxide. *Journal of Cereal Science*, 29(1), 63–75. https://doi.org/http://dx.doi.org/10.1006/jcrs.1998.0222
- Schroeder, J. W. (2012). Distillers Grains as a Protein and Energy Supplement for Dairy Cattle. *NDUS Extension Service*, 1–8. Retrieved from https://www.ag.ndsu.edu/pubs/ansci/dairy/as1241.pdf
- Schulze, E. (1891). Information regarding chemical composition of plant cell membrane. *Ber Dtsch Chem Ges*, *24*, 2277–2287.
- Selle, P. H., Ravindran, V., and Partridge, G. G. (2009). Beneficial effects of xylanase and/or phytase inclusions on ileal amino acid digestibility, energy utilisation, mineral retention and growth performance in wheat-based broiler diets. *Animal Feed Science and Technology*, 153(3–4), 303–313. https://doi.org/10.1016/j.anifeedsci.2009.06.011
- Shah, A. R., Shah, R. K., and Madamwar, D. (2006). Improvement of the quality of whole wheat bread by supplementation of xylanase from Aspergillus foetidus. *Bioresource Technology*, 97(16), 2047–2053. https://doi.org/10.1016/J.BIORTECH.2005.10.006
- Sharma, M. (2013). Xylanases: An Overview. *British Biotechnology Journal*, 3(1), 1–28. https://doi.org/10.9734/BBJ/2013/1784
- Shiiba, K., Yamada, H., Hara, H., Okada, K., and Nagao, S. (1993). Purification and characterization of two arabinoxylans from wheat bran. *Cereal Chemistry*, 70(2), 209–214.
- Sicard, D., and Legras, J.-L. (2011). Bread, beer and wine: Yeast domestication in the Saccharomyces sensu stricto complex. *Comptes Rendus Biologies*, *334*(3), 229–236. https://doi.org/10.1016/j.crvi.2010.12.016
- Singh, R. D., Banerjee, J., and Arora, A. (2015). Prebiotic potential of oligosaccharides: A focus on xylan derived oligosaccharides. *Bioactive Carbohydrates and Dietary Fibre*, *5*(1), 19–30. https://doi.org/10.1016/J.BCDF.2014.11.003

- Siró, I., Kápolna, E., Kápolna, B., and Lugasi, A. (2008, November). Functional food. Product development, marketing and consumer acceptance-A review. *Appetite*. https://doi.org/10.1016/j.appet.2008.05.060
- Sitanaka, N. Y., Lemos Budiño, F. E., De-Oliveira, S. R., Vilas Boas, A. D. C., and De-Moraes, J. E. (2018). ENZYME COMPLEX SUPPLEMENTATION ON THE PERFORMANCE OF SWINE IN GROWTH AND FINISHING PHASES. *Revista Caatinga*, 31(3), 748–758. https://doi.org/10.1590/1983-21252018v31n325rc
- Slade, A. M., Hoj, P. B., Morrice, N. A., and Fincher, G. B. (1989). Purification and characterization of three $(1 \rightarrow 4)$ - β -d-xylan endohydrolases from germinated barley. *European Journal of Biochemistry*, *185*(3), 533–539.
- Slavin, J. (2006). Functional Food Carbohydrates. Functional Food Carbohydrates. CRC Press. https://doi.org/10.1016/j.tifs.2007.11.004
- Sluiter, a, Hames, B., Ruiz, R., Scarlata, C., Sluiter, J., and Templeton, D. (2008). Determination of Ash in Biomass Laboratory Analytical Procedure (LAP) Issue Date: 7 / 17 / 2005 Determination of Ash in Biomass Laboratory Analytical Procedure (LAP). https://doi.org/NREL/TP-510-42622
- Sluiter, A., Hames, B., Ruiz, R., Scarlata, C., Sluiter, J., Templeton, D., and Crocker, D. (2008). Determination of structural carbohydrates and lignin in biomass. Laboratory analytical procedure (Vol. 1617). Colorado: National Renewable Energy Laboratory Golden, CO.
- Smil, V. (1999). Crop Residues: Agriculture's Largest Harvest. *BioScience*, 49(4), 299– 308. https://doi.org/10.2307/1313613
- Smith, M. M., and Hartley, R. D. (1983). Occurrence and nature of ferulic acid substitution of cell-wall polysaccharides in graminaceous plants. *Carbohydrate Research*, 118, 65–80. Retrieved from http://www.sciencedirect.com/science/article/pii/0008621583880367
- Spiehs, M. J., Whitney, M. H., and Shurson, G. C. (2002). Nutrient database for distiller's dried grains with solubles produced from new etahnol plants in Minnesota and South Dakota. *Journal of Animal Science*, *80*, 2639–2645. Retrieved from https://www.biofuelscoproducts.umn.edu/sites/biodieselfeeds.cfans.umn.edu/fil

es/cfans_asset_415699.pdf

- Staneva, G., Petkova, D., Hazarosova, R., Georgieva, R., Pankov, R., Skrobanska, R., and Momchilova, A. (2014). Intake of Xylooligosaccharides Alters the Structural Organization of Liver Plasma Membrane Bilayer. *Food Biophysics*, 9(2), 138–144. https://doi.org/10.1007/s11483-013-9326-z
- Stanton, C., Ross, R. P., Fitzgerald, G. F., and Sinderen, D. Van. (2005). Fermented functional foods based on probiotics and their biogenic metabolites. *Current Opinion in Biotechnology*, 16(2), 198–203. https://doi.org/10.1016/j.copbio.2005.02.008
- Stein, K. (2007). *Food vs Biofuel. Journal of the American Dietetic Association* (Vol. 107). https://doi.org/10.1016/j.jada.2007.09.014

- Steinkraus, K. H. (1997). Classification of fermented foods: worldwide review of household fermentation techniques. *Food Control*, 8(5–6), 311–317. https://doi.org/10.1016/S0956-7135(97)00050-9
- Stepan, A. M., Ansari, F., Berglund, L., and Gatenholm, P. (2014). Nanofibrillated cellulose reinforced acetylated arabinoxylan films. *Composites Science and Technology*, 98, 72–78. https://doi.org/10.1016/j.compscitech.2014.04.010
- Stevanic, J. S., Joly, C., Mikkonen, K. S., Pirkkalainen, K., Serimaa, R., Rémond, C., ... Salmén, L. (2011). Bacterial nanocellulose-reinforced arabinoxylan films. *Journal* of Applied Polymer Science, 122(2), 1030–1039. https://doi.org/10.1002/app.34217
- Stock, R. a, Lewis, J. M., Klopfenstein, T. J., and Milton, C. T. (2000). Review of new information on the use of wet and dry milling feed by-products in feedlot diets. *Jornal of Animal Science*, 77, 1–12. https://doi.org/10.2134/jas2000.77E-Suppl1w
- Subramaniyan, S., and Prema, P. (2002). Biotechnology of Microbial Xylanases: Enzymology, Molecular Biology, and Application. *Critical Reviews in Biotechnology*, 22(1), 33–64. https://doi.org/10.1080/07388550290789450
- Sun, R. C., Sun, X. F., and Ma, X. H. (2002). Effect of ultrasound on the structural and physiochemical properties of organosolv soluble hemicelluloses from wheat straw. *Ultrasonics Sonochemistry*, 9(2), 95–101. https://doi.org/10.1016/S1350-4177(01)00102-X
- Sun, X. F., Xu, F., Sun, R. C., Geng, Z. C., Fowler, P., and Baird, M. S. (2005). Characteristics of degraded hemicellulosic polymers obtained from steam exploded wheat straw. *Carbohydrate Polymers*, 60(1), 15–26. https://doi.org/10.1016/j.carbpol.2004.11.012
- Sunna, A., and Antranikian, G. (1997). Xylanolytic enzymes from fungi and bacteria. *Critical Reviews in Biotechnology*, *17*(1), 39–67. https://doi.org/10.3109/07388559709146606
- Surek, E., and Buyukkileci, A. O. (2017). Production of xylooligosaccharides by autohydrolysis of hazelnut (Corylus avellana L.) shell. *Carbohydrate Polymers*, 174, 565–571. https://doi.org/10.1016/j.carbpol.2017.06.109
- Suzuki, K., Kitamura, S., Kato, Y., and Itoh, T. (2000). Highly Substituted Glucuronoarabinoxylans (hsGAXs) and Low-Branched Xylans Show a Distinct Localization Pattern in the Tissues of Zea mays L. *Plant and Cell Physiology*, 41(8), 948–959. https://doi.org/10.1093/pcp/pcd019
- Swartz, M. (2010). HPLC detectors: a brief review. Journal of Liquid Chromatography and Related Technologies, 33, 1130–1150. https://doi.org/10.1080/10826076.2010.484356
- Swennen, K., Courtin, C. M., Lindemans, G. C., and Delcour, J. A. (2006). Large-scale production and characterisation of wheat bran arabinoxylooligosaccharides. In *Journal of the Science of Food and Agriculture* (Vol. 86, pp. 1722–1731). Wiley-Blackwell. https://doi.org/10.1002/jsfa.2470

Świątkiewicz, S., koreleski, J., Świątkiewicz, S., and koreleski, J. The use of distillers

dried grains with solubles (DDGS) in poultry nutrition, 64 World's Poultry Science Journal § (2008). Cambridge University Press on behalf of Worlds Poultry Science Association. https://doi.org/10.1017/S0043933908000044

- Taiz, L., and Honigman, W. A. (1976). Production of cell wall hydrolyzing enzymes by barley aleurone layers in response to gibberellic acid. *Plant Physiology*, 58(3), 380–386. https://doi.org/10.1104/pp.58.3.380
- Tateyama, I., Hashii, K., Johno, I., Iino, T., Hirai, K., Suwa, Y., and Kiso, Y. (2005). Effect of xylooligosaccharide intake on severe constipation in pregnant women. *Journal* of Nutritional Science and Vitaminology, 51(6), 445–448. https://doi.org/10.3177/jnsv.51.445
- Teleman, A., Hausalo, T., Tenkanen, M., and Vuorinen, T. (1995). Identification of the acidic degradation products of hexenuronic acid by NMR spectroscopy. 8th International Symposium on Wood and Pulping Chemistry, III, 109–114.
- Ten Bruggencate, S. J. M., Bovee-Oudenhoven, I. M. J., Lettink-Wissink, M. L. G., Katan, M. B., and van der Meer, R. (2006). Dietary fructooligosaccharides affect intestinal barrier function in healthy men. *The Journal of Nutrition*, 136(1), 70–74. https://doi.org/10.1093/jn/136.1.70
- Thacker, P. A., and Widyaratne, G. P. (2007). Nutritional value of diets containing graded levels of wheat distillers grains with solubles fed to broiler chicks. *Journal of the Science of Food and Agriculture*, *87*(7), 1386–1390. https://doi.org/10.1002/jsfa.2871
- Tong, L.-T., Zhong, K., Liu, L., Qiu, J., Guo, L., Zhou, X., ... Zhou, S. (2014). Effects of dietary wheat bran arabinoxylans on cholesterol metabolism of hypercholesterolemic hamsters. *Carbohydrate Polymers*, 112, 1–5. https://doi.org/10.1016/j.carbpol.2014.05.061
- Uçkun Kiran, E., Akpinar, O., and Bakir, U. (2013). Improvement of enzymatic xylooligosaccharides production by the co-utilization of xylans from different origins. *Food and Bioproducts Processing*, *91*(4), 565–574. https://doi.org/https://doi.org/10.1016/j.fbp.2012.12.002
- Uday, U. S. P., Choudhury, P., Bandyopadhyay, T. K., and Bhunia, B. (2016).
 Classification, mode of action and production strategy of xylanase and its application for biofuel production from water hyacinth. *International Journal of Biological Macromolecules*, *82*, 1041–1054.
 https://doi.org/10.1016/J.IJBIOMAC.2015.10.086
- UN DESA. (2017). World Population Prospects The 2017 Revision. New York. Retrieved from https://esa.un.org/unpd/wpp/publications/files/wpp2017_keyfindings.pdf
- Van Craeyveld, V., Swennen, K., Dornez, E., Van de Wiele, T., Marzorati, M., Verstraete, W., ... Courtin, C. M. (2008). Structurally Different Wheat-Derived Arabinoxylooligosaccharides Have Different Prebiotic and Fermentation Properties in Rats. *The Journal of Nutrition*, *138*(12), 2348–2355. https://doi.org/10.3945/jn.108.094367
- Van Craeyveld, V., Swennen, K., Dornez, E., Van de Wiele, T., Marzorati, M., Verstraete, W., ... Courtin, C. M. (2008). Structurally Different Wheat-Derived

Arabinoxylooligosaccharides Have Different Prebiotic and Fermentation Properties in Rats. *Journal of Nutrition*, *138*(12), 2348–2355. https://doi.org/10.3945/jn.108.094367

- Vázquez, M. . J., Alonso, J. . L., Domínguez, H., Parajó, J. . C., Domínguez, H., and Parajó, J. . C. (2000). Xylooligosaccharides: manufacture and applications. *Trends in Food Science and Technology*, 11(11), 387–393. https://doi.org/10.1016/S0924-2244(01)00031-0
- Vegas, R., Alonso, J. L., Domínguez, H., and Parajó, J. C. (2005). Manufacture and refining of oligosaccharides from industrial solid wastes. *Industrial and Engineering Chemistry Research*, 44(3), 614–620. https://doi.org/10.1021/ie049289+
- Viëtor, R. J., Angelino, S., and Voragen, A. G. J. (1992). Structural features of arabinoxylans from barley and malt cell wall material. *Journal of Cereal Science*, 15(3), 213–222. https://doi.org/Doi: 10.1016/s0733-5210(09)80120-3
- Viikari, L., Kantelinen, A., Sundquist, J., and Linko, M. (1994). Xylanases in bleaching: From an idea to the industry. *FEMS Microbiology Reviews*, *13*(2–3), 335–350. https://doi.org/10.1111/j.1574-6976.1994.tb00053.x
- Vinkx, C. J. A., and Delcour, J. A. (1996). Rye (Secale cerealeL.) Arabinoxylans: A Critical Review. *Journal of Cereal Science*, *24*(1), 1–14. https://doi.org/10.1006/jcrs.1996.0032
- Vinkx, C. J. A., Van Nieuwenhove, C. G., and Delcour, J. (1991). Physicochemical and Functional-Properties of Rye Nonstarch Polysaccharides. 3. Oxidative Gelation of a Fraction Containing Water-Soluble Pentosans and Proteins. *Cereal Chemistry*, 68(6), 617–622.
- von Blottnitz, H., and Curran, M. A. (2007). A review of assessments conducted on bioethanol as a transportation fuel from a net energy, greenhouse gas, and environmental life cycle perspective. *Journal of Cleaner Production*, *15*(7), 607– 619. https://doi.org/10.1016/J.JCLEPRO.2006.03.002
- Voragen, A. (1998, August 1). Technological aspects of functional food-related carbohydrates. *Trends in Food Science and Technology*. Elsevier. https://doi.org/10.1016/S0924-2244(98)00059-4
- Voragen, A. G. J., Schols, H. A., Searle-Van Leeuwen, M. F., Beldman, G., and Rombouts, F. M. (1986). Analysis of oligomeric and monomeric saccharides from enzymatically degraded polyssacharides by high-performance liquid chromatography. *Journal of Chromatography A*, 370(C), 113–120. https://doi.org/10.1016/S0021-9673(00)94679-X
- Walia, A., Guleria, S., Mehta, P., Chauhan, A., and Parkash, J. (2017). Microbial xylanases and their industrial application in pulp and paper biobleaching: a review. *3 Biotech*, 7(1), 11. https://doi.org/10.1007/s13205-016-0584-6
- Wan Azelee, N. I., Jahim, J. M., Ismail, A. F., Fuzi, S. F. Z. M., Rahman, R. A., and Md Illias, R. (2016). High xylooligosaccharides (XOS) production from pretreated kenaf stem by enzyme mixture hydrolysis. *Industrial Crops and Products*, 81, 11– 19. https://doi.org/10.1016/j.indcrop.2015.11.038

- Wang, J., Cao, Y., Wang, C., and Sun, B. (2011). Wheat bran xylooligosaccharides improve blood lipid metabolism and antioxidant status in rats fed a high-fat diet. *Carbohydrate Polymers*, 86(3), 1192–1197. https://doi.org/10.1016/j.carbpol.2011.06.014
- Wang, J., Sun, B., Cao, Y., Tian, Y., and Wang, C. (2009). Enzymatic preparation of wheat bran xylooligosaccharides and their stability during pasteurization and autoclave sterilization at low pH. *Carbohydrate Polymers*, 77(4), 816–821. https://doi.org/10.1016/J.CARBPOL.2009.03.005
- Wang, J., Sun, B., Cao, Y., and Wang, C. (2010). In vitro fermentation of xylooligosaccharides from wheat bran insoluble dietary fiber by Bifidobacteria. *Carbohydrate Polymers*, 82(2), 419–423. https://doi.org/10.1016/j.carbpol.2010.04.082
- Wang, J., Sun, B., Liu, Y., and Zhang, H. (2014). Optimisation of ultrasound-assisted enzymatic extraction of arabinoxylan from wheat bran. *Food Chemistry*, 150, 482–488. https://doi.org/10.1016/j.foodchem.2013.10.121

 Wang, X., Xu, Y., Fan, L., Yong, Q., and Yu, S. (2012). Simultaneous separation and quantitative determination of monosaccharides, uronic acids, and aldonic acids by high performance anion-exchange chromatography coupled with pulsed amperometric detection in corn stover prehydrolysates. *BioResources*, 7(4), 4614–4625. Retrieved from

http://ojs.cnr.ncsu.edu/index.php/BioRes/article/viewFile/BioRes_07_4_4614_W ang_Simultaneous_Separation_Monosaccharide_Uronic_Aldonic_Acid_Corn_Sto ver/1709

- Whistler, R. L., and Masak, E. (1955). Enzymatic Hydrolysis of Xylan1. J. Am. Chem. Soc., 77(5), 1241–1243. https://doi.org/10.1021/ja01610a042
- Willför, S., Pranovich, A., Tamminen, T., Puls, J., Laine, C., Suurnäkki, A., ... Holmbom, B. (2009). Carbohydrate analysis of plant materials with uronic acid-containing polysaccharides-A comparison between different hydrolysis and subsequent chromatographic analytical techniques. *Industrial Crops and Products*, 29(2–3), 571–580. https://doi.org/10.1016/j.indcrop.2008.11.003
- Wong, K. K., Tan, L. U., and Saddler, J. N. (1988). Multiplicity of beta-1,4-xylanase in microorganisms: functions and applications. *Microbiological Reviews*, 52(3), 305–317. Retrieved from http://www.ncbi.nlm.nih.gov/pubmed/3141761
- Wood, C., Aubert, P., Rosentrater, K. A., and Muthukumarappan, K. (2012). Techno-Economic Modeling of a Corn Based Ethanol Plant in 2011. Agricultural and Biosystems Engineering. Retrieved from http://lib.dr.iastate.edu/abe_eng_conf/97
- Wu, H., Li, H., Xue, Y., Luo, G., Gan, L., Liu, J., ... Long, M. (2017). High efficiency coproduction of ferulic acid and xylooligosaccharides from wheat bran by recombinant xylanase and feruloyl esterase. *Biochemical Engineering Journal*, 120, 41–48. https://doi.org/10.1016/j.bej.2017.01.001
- Wu, Y. B., Ravindran, V., Thomas, D. G., Birtles, M. J., and Hendriks, W. H. (2004). Influence of phytase and xylanase, individually or in combination, on

performance, apparent metabolisable energy, digestive tract measurements and gut morphology in broilers fed wheat-based diets containing adequate level of phosphorus. *British Poultry Science*, *45*(1), 76–84. https://doi.org/10.1080/00071660410001668897

- Xiang, Z., and Runge, T. (2016). Emulsifying properties of succinylated arabinoxylanprotein gum produced from corn ethanol residuals. *Food Hydrocolloids*, *52*, 423– 430. https://doi.org/10.1016/j.foodhyd.2015.07.018
- Xu, F., Liu, C. F., Geng, Z. C., Sun, J. X., Sun, R. C., Hei, B. H., ... Je, J. (2006). Characterisation of degraded organosolv hemicelluloses from wheat straw. *Polymer Degradation and Stability*, *91*(8), 1880–1886. https://doi.org/10.1016/j.polymdegradstab.2005.11.002
- Xue, J. L., Zhao, S., Liang, R. M., Yin, X., Jiang, S. X., Su, L. H., ... Feng, J. X. (2016). A biotechnological process efficiently co-produces two high value-added products, glucose and xylooligosaccharides, from sugarcane bagasse. *Bioresource Technology*, 204, 130–138. https://doi.org/10.1016/j.biortech.2015.12.082
- Yadav, M. P., Johnston, D. B., and Hicks, K. B. (2007). Structural Characterization of Corn Fiber Gums from Coarse and Fine Fiber and a Study of Their Emulsifying Properties. *Journal of Agricultural and Food Chemistry*, 55(15), 6366–6371. https://doi.org/10.1021/jf070024q
- Yoo, C. G., Kim, H., Lu, F., Azarpira, A., Pan, X., Oh, K. K., ... Kim, T. H. (2015). Understanding the Physicochemical Characteristics and the Improved Enzymatic Saccharification of Corn Stover Pretreated with Aqueous and Gaseous Ammonia. *BioEnergy Research*, 9(1), 67–76. https://doi.org/10.1007/s12155-015-9662-6
- Yoshihide Suwa; Kunimasa Koga; Shigeaki Fujikawa; Masako Okazaki, Toshio Irie, A. T. N. (1999). BIFIDOBACTERIUM BIFIDUM PROLIFERATION PROMOTING
 COMPOSITION CONTAINING XYLOOLIGOSACCHARIDE. *Journal of the Canadian Dental Association*. https://doi.org/10.1074/JBC.274.42.30033.(51)
- Zeitoun, R., Pontalier, P. Y., Marechal, P., and Rigal, L. (2010). Twin-screw extrusion for hemicellulose recovery: influence on extract purity and purification performance. *Bioresource Technology*, 101(23), 9348–9354. https://doi.org/10.1016/j.biortech.2010.07.022
- Zhang, D., Williams, B. A., Mikkelsen, D., Li, X., Keates, H. L., Lisle, A. T., ... Bryden, W. L. (2015). Soluble arabinoxylan alters digesta flow and protein digestion of red meat-containing diets in pigs. *Nutrition*, 31(9), 1141–1147. https://doi.org/10.1016/j.nut.2015.03.006
- Zhang, H., Xu, Y., and Yu, S. (2017). Co-production of functional xylooligosaccharides and fermentable sugars from corncob with effective acetic acid prehydrolysis. *Bioresource Technology*, 234, 343–349. https://doi.org/10.1016/j.biortech.2017.02.094
- Zhang, Z., Smith, C., and Li, W. (2014). Extraction and modification technology of arabinoxylans from cereal by-products: A critical review. *Food Research International*, *65*, 423–436. https://doi.org/10.1016/j.foodres.2014.05.068
- Zhou, S., Liu, X., Guo, Y., Wang, Q., Peng, D., and Cao, L. (2010). Comparison of the

immunological activities of arabinoxylans from wheat bran with alkali and xylanase-aided extraction. *Carbohydrate Polymers*, *81*(4), 784–789. https://doi.org/10.1016/j.carbpol.2010.03.040

Zyłe, K., Gogol, D., Koreleski, J., Światkiewicz, S., and Ledoux, D. R. (1999). Simultaneous application of phytase and xylanase to broiler feeds based on wheat: In vitro measurements of phosphorus and pentose release from wheats and wheat-based feeds. *Journal of the Science of Food and Agriculture, 79*(13), 1832–1840. https://doi.org/10.1002/(SICI)1097-0010(199910)79:13<1832::AID-JSFA441>3.0.CO;2-Q

Appendix

DWG production batches

Batch No.	Starting wheat kg	DWG kg	Moisture %
1	6.0	5.1	75
2	6.0	5.2	77
3	6.0	5.2	75
4	6.0	5.5	77
5	6.0	6.5	80
6	6.0	4.5	71
7	6.0	4.9	75
8	6.0	5.2	75
9	6.0	4.7	72
10	6.0	4.9	74

Arabinoxylan production batches

Batch	Starting wheat kg	DWG kg db	Crude extract g	AX purity %
АХа	12	2.55	900	19.5
AXb	12	2.55	420	44.4
WEAX	10	2.1	92	74.2
WUAX			85	51.6

Conferences

Oral presentations

- American Association of Cereals Chemists International (AACCI) Conference 2018, London.
- P2P Plants to Products Final Showcase 2018, Birmingham.

Poster presentation

LBnet International conference 2018, Manchester.